# UNITED STATES DISTRICT COURT DISTRICT OF MASSACHUSETTS

IN RE: NEW ENGLAND COMPOUNDING PHARMACY, INC. PRODUCTS LIABILITY LITIGATION	) ) ) MDL No. 1:13-md-2419-FDS
This Document Relates to: All Cases	) ) )
IN RE: NEW ENGLAND COMPOUNDING PHARMACY CASES	) ) ) Master Docket No. 12-12052-FDS
This Document Relates to All Cases	) ) )

# MEMORANDUM OF LAW IN SUPPORT OF CHAPTER 11 TRUSTEE'S MOTION TO TRANSFER PERSONAL INJURY TORT AND WRONGFUL DEATH CASES TO THIS COURT PURSUANT TO 28 U.S.C. §§ 157(b)(5) AND 1334

Paul D. Moore, the Chapter 11 Trustee for the Estate of New England Compounding Pharmacy, Inc. ("NECC" or the "Debtor"), hereby files this memorandum of law in support of his motion for entry of an order transferring to this Court the personal injury and wrongful death cases identified on Exhibit A hereto (the "Pending Actions"), pursuant to 28 U.S.C. §§ 157(b)(5) and 1334 (the "Motion"). In support of the Motion, the Trustee states as follows:

<sup>&</sup>lt;sup>1</sup> The Pending Actions include lawsuits filed against NECC, its shareholders, directors, officers, employees, and other third parties (some of which are related to or affiliated with NECC), and include the following categories of lawsuits: (1) lawsuits pending in federal courts which have not been transferred to the multidistrict forum (listed on Exhibit A1); (2) lawsuits pending in state courts which are in the process of being removed to federal courts pursuant to 28 U.S.C. §§ 1441 and /or 1452(a) (listed on Exhibit A2); and (3) lawsuits pending in state courts for which a notice of removal has not yet been filed (listed on Exhibit A3). The Pending Actions also include lawsuits which do not name NECC as a defendant, but which name as defendants: (a) parties related to NECC; (b) parties which may have indemnification and/or contribution claims against NECC; and/or (c) parties against which NECC may have claims to recover damages for the benefit of creditors of NECC's estate (lawsuits within categories (a) – (c) are listed on Exhibit A4).

#### **INTRODUCTION**

Pursuant to § 157(b)(5), the Trustee seeks to transfer to this Court the Pending Actions, which are at the very heart of the Debtor's bankruptcy case (the "Bankruptcy Case"). Each of the Pending Actions involves claims based upon allegations that: (1) contaminated pharmaceuticals (e.g. injectable steroid methyl-prednisolone acetate) (the "Compounds") were compounded, tested, marketed and/or distributed by NECC and other third parties; and (2) the Compounds caused personal injury or death. These personal injury and wrongful death claims represent the vast majority of all the claims against the Debtor's estate. The orderly and efficient resolution of these claims is essential to the Trustee's ability to formulate and effectuate a chapter 11 plan that treats all of the claimants of the Debtor equitably, irrespective of whether a particular claimant filed a lawsuit against the Debtor. In transferring a number of personal injury and wrongful death actions to the multidistrict litigation pending in this Court relating to the Debtor, In re New England Compounding Pharmacy, Inc. Products Liability Litigation, 13-md-2419-FDS (D. Mass.) (the "NECC MDL"), the Judicial Panel on Multidistrict Litigation (the "JPML") has already expressly recognized that it is in the best interests of litigants to have all of such personal injury actions consolidated in one forum.

The Trustee submits that the appropriate forum for all such matters is this Court. Transfer to this Court would further § 157(b)(5)'s explicit congressional purpose of: (1) centralizing the adjudication of claims against the estate; (2) furthering the efficient administration of the Debtor's estate; (3) ensuring that the Pending Actions are resolved in an expeditious manner that minimizes the burden on the Debtor's estate and the related costs of administering the estate, thereby conserving the Debtor's limited assets and ensuring that similarly situated creditors are treated equitably. Further, centralization of the Pending Actions

in this Court will facilitate opportunities for an efficient, expeditious and economical resolution of the Pending Actions for the benefit of the Debtor's tort claimants and other creditors.<sup>2</sup> Transfer of the Pending Actions to this Court will not affect any substantive rights that the plaintiffs or other non-Debtor parties have or assert in their Pending Actions. All plaintiffs and other parties will have their day in court.

As will be discussed further below, the mandatory language embodied in § 157(b)(5) creates a strong presumption in favor of transferring the Pending Actions. While some courts have held that principles of permissive abstention may have some limited application to the § 157(b) analysis, permissive abstention is inappropriate as to any of the Pending Actions. Indeed, even the Pending Actions that do not identify NECC as a defendant should be transferred because this Court has jurisdiction over such actions pursuant to § 1334 and transfer is otherwise appropriate.<sup>3</sup>

For example, in connection with In re: Twinlabs Personal Injury Cases, No. 03 Civ. 9169 (S.D.N.Y) ("Twinlabs"), and the Twinlabs related Chapter 11 Case, In re TL Administration Corporation, No. 03-15564 (Bankr. S.D.N.Y.), centralization of the personal injury and wrongful death cases concerning or relating to the debtor in the district court pursuant to § 157(b)(5) allowed the district court to enter orders and case procedures which facilitated and resulted in a Chapter 11 plan that satisfactorily and globally resolved all under- or non-insured personal injury and wrongful death cases against the debtor and the non-debtor co-defendants to those actions. See In re: Twinlabs Personal Injury Cases, No. 03 Civ. 9169, 2004 WL 435083 (S.D.N.Y. Mar. 8, 2004) ("Twinlabs II") (transferring "related to" cases - cases where the debtor was not a named party but which had a conceivable effect on the bankruptcy estate – to the Southern District of New York per § 157(b)(5)); Alcantar v. Twin Laboratories, Inc. (In re Twin Laboratories, Inc., et al.), 300 B.R. 836, 840-41 (S.D.N.Y. 2003) ("Twinlabs I") (transferring cases against debtor to the Southern District of New York per § 157(b)(5)); Case Management Order No. 1 ¶¶ VI, VII, VIII. In re: Ephedra Products Liability Litigation, 04 M.D. 1598, Doc. No. 17 (S.D.N.Y. Apr. 27, 2004) (providing for, inter alia, partial withdrawal of the reference, claims centralization and information procedures, and temporary stay of proceedings and order of mediation in aid of global mediation of subject cases against debtor and nondebtors), attached hereto at Exhibit B; In re TL Administration Corporation, No. 03-15564, Doc. Nos. 787, 846 (Bankr. S.D.N.Y. June 3, 2005 & July 27, 2005) (First Amended Plan of Reorganization and Order Confirming same), attached hereto as Exhibit C.

<sup>&</sup>lt;sup>3</sup> By filing this Motion, the Trustee does not seek to waive or in any way modify the automatic stay that applies to the Pending Actions under 11 U.S.C. 362. Further, the filing of this Motion is without prejudice to any position that the Trustee may take regarding whether 11 U.S.C. §§ 105 and 362 operate to stay any Pending Action or to render null and void the post-petition filing of any Pending Action.

#### **BACKGROUND**

On December 21, 2012 (the "<u>Petition Date</u>"), the Debtor commenced the Bankruptcy Case by filing with the Bankruptcy Court a voluntary petition seeking relief under chapter 11 of title 11 of the United States Bankruptcy Code (the "<u>Bankruptcy Code</u>"). Thereafter, on January 24, 2013, the Bankruptcy Court entered an order authorizing the appointment of a Chapter 11 Trustee. On January 25, 2013, Paul D. Moore was appointed as the Chapter 11 Trustee of the Debtor's estate.

Prior to the Petition Date, NECC claims it operated as a compounding pharmacy which combined and mixed ingredients to create specific formulations of pharmaceutical products. Since then, numerous individuals across the country were stricken with fungal meningitis (the "Outbreak"). The Centers For Disease Control and Prevention ("CDC") advises that, as of March 4, 2013, 48 people had died and over 720 individuals had fallen ill.<sup>4</sup> The CDC's investigation is ongoing, and further deaths and illnesses continue to be reported. Indeed, the tally of illnesses and deaths increased since last week. As of February 25, 2013, 714 illnesses and 48 deaths had been reported. As of February 18, 2013, 707 illnesses and 47 deaths had been reported.<sup>5</sup>

NECC avers that the CDC attributes the Outbreak of fungal meningitis to contaminated Compounds distributed by NECC. NECC represents that, in early October, it initiated a nationwide recall of thousands of vials of the Compounds and ceased operation.

Upon information and belief, as of late January 2013, at least 170 lawsuits against NECC

<sup>&</sup>lt;sup>4</sup> Reported at <a href="http://www.cdc.gov/HAI/outbreaks/meningitis-map-large.html#casecount\_table">http://www.cdc.gov/HAI/outbreaks/meningitis-map-large.html#casecount\_table</a> (last visited on March 6, 2013).

<sup>&</sup>lt;sup>5</sup> The CDC updates the count of cases of reported illnesses and deaths on the Monday of each week.

had been filed in state and federal courts across the country in connection with the Outbreak. Since then, the number of lawsuits has increased to at least 180. In addition, NECC has received demands from over 300 additional claimants asserting injury from the contaminated pharmaceuticals, most, if not all, of which likely will lead to further suits. A number of the personal injury and wrongful death claims have already been transferred to this Court for pretrial purposes pursuant to 28 U.S.C. § 1407. For pretrial purposes, such cases have been consolidated in the NECC MDL.

NECC claims it filed the Bankruptcy Case in response to the volume and wide geographic distribution of cases it claims it confronts in order to forge, as was successfully done in *Twinlabs* and in other bankruptcy cases predominated by personal injury and wrongful death cases, a consensual, comprehensive and global resolution of these claims pursuant to a plan establishing a compensation fund for affected claimants based on agreements to be reached among them, the Debtor, its insurers and other non-Debtor parties with potential liability for claims asserted in the cases.

This Court has jurisdiction over this Motion pursuant to 28 U.S.C. §§ 157 and 1334. The statutory bases for the relief requested are set forth in 28 U.S.C. § 157(b)(5).

### **ARGUMENT**

## A. Transfer of the Pending Actions to This Court Is Appropriate.

Section 157(b)(5) provides that, when a bankruptcy case is pending, the district court in which such case is pending shall order that related personal injury tort and wrongful death claims shall be tried in such district court or in the district court in the district in which the claim arose.

<sup>&</sup>lt;sup>6</sup> See N.V.E., Inc. (Bankr. D.N.J.) Case No. 05-35692; Metabolife Int'l Inc. (Bankr. S.D. Cal.) Case No. 05-06040; Muscletech Research and Development, Inc. (Ont. S.C.J., Court File No 06-CL-6241), recognized as foreign main proceeding under Chapter 15, Case No 06-10092 (Bankr. S.D.N.Y).

11 U.S.C. § 157(b)(5).<sup>7</sup> A debtor may therefore move to transfer such personal injury tort and wrongful death claims to one of two venues: (1) the district where the bankruptcy is proceeding; or (2) the district where the claim arose. *In re Pan Am Corp.*, 16 F.3d 513, 516 (2d Cir. 1994) ("*Pan Am II*") (citations omitted); *accord In re Waterman S.S. Corp.*, 63 B.R. 435, 437 (Bankr. S.D.N.Y. 1986) (the district court in which the bankruptcy case is pending is in the best position to gauge whether it can handle the underlying litigations in a way that can be structured to try common issues).

The manifest purpose of § 157(b)(5) is to centralize the administration of cases arising under or related to proceedings under title 11 and to eliminate the multiplicity of forums for the adjudication of parts of a bankruptcy case. *In re Pan Am Corp.*, 950 F.2d 839, 845 (2d Cir. 1991) ("*Pan Am I*") (citations omitted); *A.H. Robbins Co. v. Piccinin*, 788 F.2d 994, 1011 (4th Cir. 1986) (centralized adjudication under § 157 allows "all interests to be heard" and "the interest of all claimants [to] be harmonized" and ensures a "prompt, fair and complete resolution of all claims related to bankruptcy proceedings"); *In re Metabolife Int'l, Inc.*, S.D. Cal. Case No. 3:05-01802, Doc. No. 29 (S.D. Cal. Nov. 3, 2005) (centralizing over 70 personal injury tort and wrongful death cases against a debtor's estate in a single district court under § 157(b)(5) as being in the best interest of all involved parties). In addition, transfers of cases create a "single focal point" for developing a reasonable chapter 11 plan. *See A.H. Robins*, 788 F.2d at 1011, 1014 (transferring personal injury cases naming debtors and non-debtors to district court in which bankruptcy case is pending under § 157(b)(5) is conducive to interests of all concerned).

The foregoing principles and intent of § 157(b)(5) establish that transfer of the Pending

<sup>&</sup>lt;sup>7</sup> Section 157(b)(5) states: "The district court shall order that personal injury tort and wrongful death claims shall be tried in the district court in which the bankruptcy case is pending, or in the district court in the district in which the claim arose, as determined by the district court in which the bankruptcy case is pending." 11 U.S.C. § 157(b)(5).

Actions to this Court is appropriate. Each of the Pending Actions is a personal injury or wrongful death case relating to or associated with the Bankruptcy Case and each involves claims against property of NECC and/or impacts claims NECC may have against other potentially responsible parties. The Pending Actions identify NECC as a defendant and include codefendants whom NECC has a duty to defend and indemnify or who may assert contribution claims against NECC.

Furthermore, each of the Pending Actions involves claims regarding the alleged use of the Compounds and claims regarding the Outbreak. The Pending Actions present a number of similar issues including, *inter alia*, issues relating to the compounding, preparation, manufacture, distribution and marketing of the Compounds. Thus, not only is each of the Pending Actions associated with and/or related to the Bankruptcy Case, each also shares certain issues of fact with the proceedings pending before this Court in the NECC MDL. Indeed, a number of personal injury and wrongful death cases have already been transferred to the NECC MDL for pretrial purposes pursuant to 28 U.S.C. § 1407 [NECC MDL Doc. Nos. 2, 8]. On February 15, 2013,

<sup>&</sup>lt;sup>8</sup> Fifteen of the Pending Actions do not include NECC as a defendant. Two of these fifteen Pending Actions name as defendants, among others, NECC's shareholders, directors and entities whom NECC is obligated to defend and indemnify. The other fifteen Pending Actions are filed against entities (i) which may assert a contingent contribution and/or indemnification claim against NECC on the ground that their liability is directly attributable to NECC or (ii) against which the NECC estate may have claims it may assert for the benefit of NECC's estate.

<sup>&</sup>lt;sup>9</sup> The presence of certain similar issues in the Pending Actions arising from or related to compounding, preparation, manufacture, distribution and marketing of the Compounds should not be interpreted as the Trustee's agreement or admission that the "class" criteria of Fed. R. Civ. P 23 are satisfied here or that a "class" vehicle is appropriate in these circumstances or superior to other available methods (such as a Chapter 11 plan) for the fair and efficient adjudication of the claims. Indeed, the outcome of each individual case will turn on highly particularized and individualized facts including, but not limited to, the victims' diagnosis, the severity of their injuries, the impact of their injuries on their lives and work, all of which would have to be resolved in determining the amount of any allowed claim. *See In re Ephedra Prods. Liab. Litig.* 231 F.R.D. 167 (S.D.N.Y. 2005) (denying class certification as to claims falling within the purview of 28 U.S.C. § 157(b)(5)).

<sup>&</sup>lt;sup>10</sup> In addition to the two transfer orders filed in the NECC MDL, the JPML, in proceeding number 1249, has issued two additional Conditional Transfer Orders, CTO-2 and CTO-3, on February 22, 2013, and March 4, 2013, respectively transferring additional cases to the NECC MDL [JPML MDL Doc. Nos. 130 and 140].

this Court issued an order which provides that "actions transferred to this Court by the JPML pursuant to its order of February 12, 2013, entitled New England Compounding Pharmacy, Inc. Products Liability Litigation, and any tag-along actions transferred by the Panel after that date, and any related actions previously assigned to this Court" are "coordinated for pretrial purposes" [NECC MDL Doc. No. 4].

#### B. The Court Should Not Abstain from Transferring the Pending Actions.

Given the "seemingly mandatory language of § 157(b)(5) and its obvious purpose of giving particular priority to the centralization of bankruptcy-related personal injury claims in a single forum," there is a strong presumption in favor of transfer of the Pending Actions. Indeed, abstention is the exception and should be applied sparingly. *Twinlabs I*, 300 B.R. at 840-41. Consequently, a permissive abstention analysis should be applied with caution in this context, if at all, because of the strong legislative presumption favoring transfer under § 157(b)(5) and because many of the traditional abstention factors employed do not fit into the unique context of § 157(b)(5). *Id.* at 840-41 (relying on *Pan Am I*, 950 F.2d at 845 ("courts should not be too quick to abstain from exercising their transfer powers under 28 U.S.C. § 157(b)(5)")); *see also 1 Norton Bankruptcy Law and Practice* § 8.8 at n.91 (2d ed. 2001) ("to abstain pursuant to the discretionary abstention provision might contravene the legislative history and intent of § 157(b)(5)"). In fact, "in this context, perhaps even more than elsewhere, federal courts have a 'virtually unflagging obligation . . . to exercise the jurisdiction given them." *Twinlabs I*, 300 B.R. at 841 (quoting *Colorado River Water Conserv. Dist. v. United States*, 424 U.S. 800, 817

<sup>&</sup>lt;sup>11</sup> Personal injury tort and wrongful death cases are not subject to the mandatory abstention provisions of 28 U.S.C.  $\S$  1334(c)(2). *See* 28 U.S.C.  $\S$  157(b)(4); *Pan Am I*, 950 F.2d at 845.

(1976)).

The twelve factors courts consider in determining whether discretionary abstention is appropriate include:

(1) the effect or lack thereof on the efficient administration of the estate if a court recommends abstention, (2) the extent to which state law issues predominate over bankruptcy issues, (3) the difficulty or unsettled nature of the applicable state law, (4) the presence of a related proceeding commenced in state court or other nonbankruptcy court, (5) the jurisdictional basis, if any, other than 28 U.S.C. § 1334, (6) the degree of relatedness or remoteness of the proceeding to the main bankruptcy case, (7) the substance rather than form of an asserted "core" proceeding, (8) the feasibility of severing state law claims from core bankruptcy matters to allow judgments to be entered in state court with enforcement left to the bankruptcy court, (9) the burden [on] the court's docket, (10) the likelihood that the commencement of the proceeding in bankruptcy court involves forum shopping by one of the parties, (11) the existence of a right to a jury trial, and (12) the presence in the proceeding of nondebtor parties.

Twinlabs I, 300 B.R. at 841; Balcor/Morristown Limited Partnership v. Vector Whippany Assocs., 181 B.R. 781, 793 (D.N.J. 1995).

Congressional intent and the strong presumption in favor of transfer weigh strongly in favor of transfer of the Pending Actions to this District and this Court's exercise of its jurisdiction, rather than abstaining.

#### 1. Transfer Will Facilitate Orderly Administration of the Debtor's Estate.

With respect to the first, fourth and sixth factors, the Pending Actions involve substantial claims (in both amount and number) against the Debtor's estate. In light of the centrality of these claims to the Bankruptcy Case, transfer and coordination of these Pending Actions in this Court will substantially further the Bankruptcy Court's effective and efficient administration of the estate. In particular, transfer will facilitate the claims estimation process and increase the odds that the plan will "assure fair and non-preferential resolution of outstanding claims."

Lindsey v. O'Brien, Tanski, Tanzer & Young Health Care Providers (In re Dow Corning Corp.), 86 F.3d 482, 496 (6th Cir. 1996) ("Dow I") (quoting A.H. Robins, 788 F.2d at 1011-12).

The JPML, in transferring cases to the NECC MDL pending in this Court, has already acknowledged that concentrating all of the personal injury and wrongful death claims in one forum in which they can be coordinated would further the effective administration of the Debtor's estate. See NECC MDL, Transfer Order [Doc. No. 2]. Certain of the facts and circumstances alleged by the various plaintiffs involve the same alleged conduct by the Debtor, its directors, related entities and third parties that contracted with the Debtor. In addition, plaintiffs allege that they incurred injuries following their alleged injection of Compounds produced by the Debtor. Coordination of these claims in this Court will permit this Court to manage the adjudication of overlapping factual and legal issues. Expert discovery, general medical causation and other scientific matters, as well as evidentiary issues and trial issues, can be consolidated, and threshold issues can be determined in a coordinated and consistent manner, thereby saving costs and maximizing the return to all creditors.

Additionally, centralization will eliminate the possibility of inconsistent rulings on scientific or factual issues and the negative impact that rulings in one case can have on subsequent cases against the Debtor in terms of both case valuation and possible preclusive effect on the litigation. By deciding such issues on a centralized basis in this Court, the limited estate assets will be conserved and protected. There is limited insurance coverage to satisfy all

 $<sup>^{\</sup>rm 12}$  The JPML's Transfer Order issued on February 12, 2013, provides:

Centralization ... will serve the convenience of the parties and witnesses and promote the just and efficient conduct of this [MDL] litigation. ... Centralization will eliminate duplicative discovery; prevent inconsistent pretrial rulings, especially with respect to class certification and discovery issues; and conserve the resources of the parties, their counsel and the judiciary.

of the claims against the estate. Centralization of the Pending Actions will therefore assure that similarly situated creditors are treated fairly and equitably under the Bankruptcy Code.

### 2. The Presence of State Law Issues Does Not Require Abstention.

Although state law issues predominate most of the claims in the Pending Actions (the second and third factors), and the plaintiffs in the Pending Actions have a right to a jury trial, these factors should not be given much, if any, weight in the context of a motion to transfer pursuant to § 157(b).

In *Twinlabs I*, the court ultimately dismissed arguments that the predominance of state law issues favored abstention. 300 B.R. 841. Specifically, the court stated that:

the Court is satisfied that none of the state law issues here involved are difficult, unsettled, or unfamiliar to a federal court, thanks to diversity jurisdiction, has its share of state-law personal injury cases. Nor, in any event, can the predominance of state law issues be given decisive effect in analyzing transfer under § 157(b)(5), for the very subject matter of that section - personal injury claims - is almost always governed by state law, and yet Congress, in enacting § 157(b)(5), singled out such claims as the very ones it wanted transferred.

*Id.* Because the Pending Actions do not involve "a particularly unusual question" or "unsettled questions" of state law, the Pending Actions should be transferred. *See Pan Am I*, 950 F.2d at 846.

#### 3. The Pending Actions are Integrally Related to the Bankruptcy Case.

In relation to the sixth factor, where, as here, the personal injury and wrongful death claims present the majority of claims against the estate, the resolution of those claims is crucial to the estate. *See*, *e.g.*, *A.H. Robins*, 788 F.2d 994; *Dow I*, 86 F.3d 482. This Court is the only court that has an overall view of the Debtor's case and all aspects of that case should be before this Court. Given the centrality of the personal injury and wrongful death claims to the Debtor's bankruptcy proceeding, the claims set forth in the Pending Actions are integrally related to the

Bankruptcy Case, and should be consolidated before this Court.

#### 4. There Will be No Additional Burden on the Court's Docket.

The ninth factor also points in favor of transferring the Pending Actions. The NECC MDL is already before this Court, and comprises over 120 cases. The universe of personal injury and wrongful death cases against the Debtor and related parties will be manageable if they are centralized before this Court and governed by an appropriate case management/coordination plan. Accordingly, the Trustee respectfully suggests that any impact caused by the transfers does not warrant abstention where, as here, transfer is otherwise appropriate and would significantly aid in the administration of the Debtor's estate.

#### 5. There Is No Forum Shopping in Connection With This Matter.

This § 157(b)(5) Motion cannot be viewed as forum shopping, the tenth factor, because a change of venue was expressly contemplated and mandated by Congress in this section. Additionally, the Trustee is not seeking to transfer only select cases pending in certain forums. Instead, he is moving for the transfer of all of the Pending Actions. Finally, the Debtor had an urgent need to file the Bankruptcy Case when it was confronted with approximately 170 personal injury and wrongful death claims alleging injury resulting from the Compounds. Having no choice but to file for protection under chapter 11, the Debtor filed its petition in the District of Massachusetts, which is the only district in which the Bankruptcy Case could have been filed. See 28 U.S.C. § 1408(1). Under these circumstances, there is simply no evidence that the filing of the Bankruptcy Case in this forum or the filing of this Motion is an attempt at forum shopping.

#### 6. The Presence of Non-Debtor Defendants Does Not Favor Abstention.

The presence of non-debtors as co-defendants, the twelfth factor, in some of the Pending Actions does not favor abstention. These non-debtors are, for the most part, related to or

affiliated with the Debtor or connected to its pharmaceuticals. These non-debtors include, among others: directors, officers, shareholders, and employees of the Debtor; entities related to the Debtor, or affiliated with it, that prepared, distributed, marketed and sold the Compounds and leased the premises out of which the Debtor operated; and persons and entities involved in the research, testing, purchasing and distribution of the Compounds. The vast majority of the claims asserted against these non-debtors are directly related to the development, testing, manufacture and sale of the Compounds. Many of these non-debtor defendants are parties to which the Debtor has contractual indemnification obligations. Indeed, the Debtor has express, contractual indemnification obligations to the majority of the non-debtor defendants named in these actions, including, but not limited to, Barry Cadden, Greg Conigliaro, Lisa Conigliaro, Carla Conigliaro, Glenn Chin (collectively, the "Individuals"), GDC Properties Management, LLC, Medical Sales Management, SW, Inc., together with certain pain management clinics and health care providers that purchased Compounds from NECC.

The Individuals are additional insureds under at least one of the Debtor's insurance policies (which is an asset of the estate) and the Debtor's by-laws provide for indemnification of certain of these Individuals in their capacity as the Debtor's officers and directors.

In addition, other non-debtor defendants, including, but not limited to, Ameridose, LLC, Alaunus Pharmaceutical, LLC, health care providers and pain management clinics, may assert contingent contribution claims and indemnification claims against NECC (as noted in the Debtor's Schedule F of Creditors Holding Unsecured Non Priority Claims with respect to certain of such non-debtor defendants) on the ground that their liability is directly attributable to

NECC,<sup>13</sup> and NECC's estate likewise may assert claims against them for contribution or indemnity. In each of these instances, these indemnification and contribution claims will have a significant impact upon the Debtor's estate.

Additionally, having all of these non-debtors - against which claims have been asserted based upon the compounding, testing, manufacture and/or sale of the Compounds - before this Court will enhance the administration of the Debtor's estate by bringing all parties with alleged liability based upon the Debtor's product before this Court. Such centralization will undoubtedly aid in the negotiation of a chapter 11 plan because all potentially liable parties (and their insurers) (i.e, potential contributors to any trust created as part of a plan) will be before this Court. Courts have uniformly observed that the transfer and centralization of related claims against non-debtors aid in the negotiation of a plan. See, e.g., Dow I, 86 F.3d at 496 (centralization increases the odds of developing a reasonable plan); In re Petroleum Tank Lines, 10 B.R. 286, 288 (Bankr. W.D.N.Y. 1981) (Negotiation of a plan will be greatly facilitated if the cases are before a single court.). By contrast, requiring certain non-debtors to proceed with litigation in numerous forums will diminish those non-debtors' ability to contribute to a plan. See, e.g., Nelson v. Gen. Elec. Capital Corp. (In re Steven P. Nelson, D.C., P.A.), 140 B.R. 814, 817 (Bankr. M.D. Fla. 1992). Likewise, it will result in the non-debtor defendants making claims and seeking and obtaining payment from NECC's insurers before NECC is able to make any claims of its own. Dow I, 86 F.3d at 495.

<sup>&</sup>lt;sup>13</sup> For example, some of these entities may seek to pursue claims against NECC based upon theories of negligent misrepresentation, intentional misrepresentation and/or breach of warranties (express or implied).

## C. Transfer of Claims Against Non-Debtors Is Appropriate.

Transfer pursuant to § 157(b)(5) is not limited to claims against a debtor. *See, e.g., Dow I*, 86 F.3d at 497; *Twinlabs II*, 2004 WL 435083, at \*2-4. Section 157(b)(5) allows the transfer of claims against non-debtor defendants that are related to the debtor's bankruptcy case where the court has jurisdiction over such claims. *Id.*; *A.H. Robins*, 788 F.2d 994.

Section 1334(b) of title 28 of the United States Code vests jurisdiction in district courts over, *inter alia*, proceedings that are "related to" a bankruptcy case. <sup>14</sup> "Related to" jurisdiction is construed broadly and encompasses proceedings which potentially have some effect on the bankruptcy estate, such as altering the debtor's rights, liabilities, options, or freedom of action or otherwise having an impact on the handling and administration of the bankruptcy estate. *In re Middlesex Power Equip. & Marine, Inc.*, 292 F.3d 61, 68 (1st Cir. 2002); *In re G.S.F. Corp.*, 938 F.2d 1467, 1475 (1st Cir. 1991) (a proceeding is related to bankruptcy where the outcome of the proceeding could conceivably have any effect on the estate being administered in bankruptcy (*quoting Pacor v. Higgins*, 743 F.2d 984 (3d Cir. 1984))); *accord Boston Reg'l Med. Ctr.*, *Inc. v. Reynolds (In re Boston Reg'l Med. Ctr.*), 410 F.2d 100 (1st Cir. 2005); *see Celotex Corp. v. Edwards*, 514 U.S. 300, 308 (1995) (Congress intended to grant comprehensive jurisdiction to the bankruptcy courts so that they might deal efficiently and expeditiously with all matters connected with the bankruptcy estate).

Where, as set forth at pages 12-13 above, the claims asserted against the non-debtors arise out of those non-debtors' association with the debtor or its products, courts have recognized

<sup>&</sup>lt;sup>14</sup> "[D]istrict courts shall have original but not exclusive jurisdiction of all civil proceedings arising under title 11, or arising in or related to cases under title 11." 28 U.S.C. § 1334(b).

that transfer of those claims is appropriate. As the Court of Appeals for the Sixth Circuit noted in *Lindsey v. The Dow Chemical Co. (In re Dow Corning Corp.)* ("<u>Dow II</u>"):

it makes little practical sense to transfer the claims against [the debtor] while refusing to transfer those against the shareholders. The claims against the shareholders arise from an identical set of facts and are merely duplicates of those against [the debtor]. The shareholders have never manufactured a silicone-implant product and are only named in the suits because of their association with [the debtor]. Thus, the actions against [the debtor's] shareholders are essentially the same exact cases that have already been transferred to the [district court].

113 F.3d 565, 570 (6th Cir. 1997). Consequently, courts have found "related to" jurisdiction where liability for contribution or for contractual indemnity on claims of sizable magnitude will impact a debtor's estate. Cambridge Place Inv. Mgmt. v. Morgan Stanley & Co., Inc., No. 10-11376, 2010 U.S. Dist. LEXIS 142954, at \*20 (D. Mass. Dec. 28, 2010), adopted by 813 F. Supp. 2d 242 (2011); Diversified Mortgage Co. v. Gold (In re Gold), 247 B.R. 574, 578 (Bankr. D. Mass. 2000) ("related to" jurisdiction exists over proceedings where neither the debtor nor the estate is a named defendant if the defendant has indemnification rights against the debtor or the estate); A.H. Robins, 788 F.2d 994, cert denied, 479 U.S. 876 (same); Dow I, 86 F.3d at 494 ("related to" jurisdiction exists over non-debtors asserting contribution and indemnification claims); Philippe v. Shape, Inc., 103 B.R. 355 (D. Me. 1989) (same); see TD Bank, N.A. v. Sewall, et al., 419 B.R. 103, 108 (D. Me. 2009) ("related to" jurisdiction exists, even in the absence of explicit indemnification agreements, over an action by a claimant against an entity that is liable for debtor's obligations because such action will affect that claimant's status vis-àvis other claimants thereby affecting administration of the estate); Twinlabs II, 2004 WL 435083, at \*2-4 (transferring "related to" case against retailer defendant, which was likely to seek indemnity from the debtor after judgment was entered against the defendant). Cf. Pacor, 743 F.2d at 995 ("related to" jurisdiction did not exist where the non-debtor's claim for indemnification was a contingent common law claim).

As in *Dow II*, the claims filed against the non-debtors here are based upon those non-debtors' relationship to the Debtor and/or the Compounds. These claims are related to the Bankruptcy Case and can impact the estate. As set forth above, the claims asserted against non-debtor defendants in the Pending Actions all arise out of the same facts as those underlying the claims against the Debtor. Moreover, because many of these non-debtors have express contractual indemnity claims and contribution claims against the Debtor, these claims would have an undeniable impact upon the estate and are, therefore, related to this Bankruptcy Case.

Further, those Pending Actions that do not name NECC as a defendant were not pending as of the Petition Date. Actions filed after the Petition Date do not name NECC as a defendant due to the automatic stay under 11 U.S.C. § 362. Nevertheless, these Pending Actions are related to this Bankruptcy Case for the reasons set forth above. These actions also include common questions of fact and transfer of these Pending Actions will serve the same purpose set forth above.

Therefore, the Trustee requests transfer of the Pending Actions. Transfer will thwart attempts to "get the jump" on other claimants and will assist in the orderly administration of the Debtor's estate, the conservation of its assets, and the fair and equitable treatment of all creditors.

## **CONCLUSION**

For all the foregoing reasons, the Trustee respectfully requests that this Court enter the attached order transferring the Pending Actions to this Court for further proceedings pursuant to 28 U.S.C. § 157(b)(5).

Dated: March 10, 2013 Respectfully submitted,

DUANE MORRIS LLP

/s/Michael R. Gottfried Michael R. Gottfried (BBO #542156)

100 High Street Suite 2400 Boston, MA 02110-1724

Phone: (857) 488-4200

Email: mrgottfried@duanemorris.com

# **EXHIBIT A**

The following is a list of the Pending Actions that the Trustee has moved to transfer to the United States District Court for the District of Massachusetts, pursuant to 28 U.S.C. § 157(b)(5):

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 20 of 31 Exhibit A1

Plaintiff(s)	Venue	Co-Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Linda S. Boggs	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:2012-cv-50132	Russell Updike
	Virginia, Roanoke	Management, P.C., d/b/a Insight Imaging,		Nolan Nicely
		Roanoke		Jennifer K.M. Crawford
				Wilson, Updike & Nicely
				228 North Maple Avenue
				PO Drawer 590
				Covington, VA 24426
Kimberly Brown	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-584	Russell Updike
	Virginia	Management, P.C., d/b/a Insight Imaging,		Nolan Nicely
		Roanoke		Jennifer K.M. Crawford
				Wilson, Updike & Nicely
				228 North Maple Avenue
				PO Drawer 590
				Covington, VA 24426
George Cary; Lilian Cary	USDC- District of	NECP d/b/a NECC; Barry Cadden; Arl Bio	Adversary No: 13-	Frederic L. Ellis
(Deceased)	Massachusetts	Pharma INC d/b/a Analytical Research	01025	Edward D. Rapaki
		Laboratories; Ameridose, LLC	Civil Docket: 1:13-	Joseph M. Makalusky
			cv-10228	Ellis & Rapaki, LLP
				85 Merrimac Street
				Suite 500
				Boston, MA 02114
				J. Douglas Peters
				Charfoos & Christensen, PC
				Hecker-Smiley Mansion
				5510 Woodward Avenue
				Detroit, MI 40202
Marvin Clark	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-00566	Patrick T. Fennell, Esq.
	Virginia, Roanoke	Management, P.C., d/b/a Insight Imaging,		Law Offices of Crandall & Katt
		Roanoke		366 Elm Avenue, S.W.
				Roanoke, VA 24016

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 21 of 31 Exhibit A1

Plaintiff(s)	Venue	Co-Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Darrel Cummings	USDC- Southern District of Florida	NECP d/b/a NECC; GEO Group, Inc; Tom Levins, Warden; Ms. N. Finnisse, MSM, HAS; Dr. J. Dauphin, D.O.; Dr. Jules Heller, M.D.; Dr. Robert Lins, M.D.; Officer	9:12-cv-81413	PRO SE
		McIntire, CO1		
Roseanna Fusco	USDC- Middle District of	NECP d/b/a NECC	5:13-cv-00095	John H. Piccin, Esq.
	Florida, Ocala Division			Piccin Law Firm
				320 N.W. 3rd Avenue
				PO Box 159
				Ocala, FL 34478
				Colson Hicks Eidson Colson Matthews
				Martinez Gonzalez Kalbec & Kane
Coleta Ray Gipson, as	USDC- Eastern District of	NECP d/b/a NECC	4:13-cv-10818	Alexander & Angelas, P.C.
Personal Respresentative to	Michigan, Southern Division			30200 Telegraph Road
the Estate of Gayle Lynn				Suite 400
Gipson				Bingham Farms, MI 48025
Forrest Linthicum	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-00585	Russell Updike
	Virginia	Management, P.C., d/b/a Insight Imaging,		Nolan Nicely
		Roanoke		Jennifer K.M. Crawford
				Wilson, Updike & Nicely
				228 North Maple Avenue
				PO Drawer 590
				Covington, VA 24426
Michelle Powell	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:2012-cv-00614	Patrick T. Fennell, Esq.
	Virginia	Management, P.C., d/b/a Insight Imaging,		Law Offices of Crandall & Katt
		Roanoke		366 Elm Avenue, S.W.
				Roanoke, VA 24016

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 22 of 31 Exhibit A1

Plaintiff(s)	Venue	Co-Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Basil E. Proffitt	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-00615	Willard J. Moody, Jr.
	Virginia, Roanoke	Management, P.C., d/b/a Insight Imaging,		The Moody Law Firm, Inc.
		Roanoke		500 Crawford Street
				Suite 300
				Portsmouth, VA 23705
				The Miller Firm, LLC
				The Sherman Building
				108 Railroad Avenue
				Orange, VA 22960
Mary E. Radford	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-00562	LichtensteinFishwick PLC
	Virginia, Roanoke	Management, P.C., d/b/a Insight Imaging,		Liberty Trust Building
		Roanoke		Suite 400, 101 South Jefferson Street
				Roanoke, VA 24004-0601
Robert Schroder; Margaret	USDC- District of	NECP d/b/a NECC; Barry Cadden; Gregory	Adversary No: 13-	Ellis & Rapaki, LLP
Schroder	Massachusetts	Conigliaro; Lisa Conigliaro; Glenn Chin;	01024	85 Merrimac Street
		Medical Sales Management, Inc.;		Suite 500
		Ameridose, LLC; ARL Bio Pharma, Inc.	Civil Docket: 1:13-	Boston, MA 02114
		d/b/a Analytical Research Laboratories;	cv-10227	
		GDC Properties Management, LLC		Charfoos & Christensen, PC
				Hecker-Smiley Mansion
				5510 Woodward Avenue
				Detroit, MI 40202
Christopher Sherrill	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-00565	Patrick T. Fennell, Esq.
	Virginia, Roanoke	Management, P.C., d/b/a Insight Imaging,		Law Offices of Crandall & Katt
		Roanoke		366 Elm Avenue, S.W.
				Roanoke, VA 24016

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 23 of 31 Exhibit A1

Plaintiff(s)	Venue	Co-Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Denia Taliaferro	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-00586	Russell Updike
	Virginia	Management, P.C., d/b/a Insight Imaging,		Nolan Nicely
		Roanoke		Jennifer K.M. Crawford
				Wilson, Updike & Nicely
				228 North Maple Avenue
				PO Drawer 590
				Covington, VA 24426
Mary Sharon Walker	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12cv 00564	Willard J. Moody, Jr.
	Virginia	Management, P.C., d/b/a Insight Imaging,		The Moody Law Firm, Inc.
		Roanoke		500 Crawford Street
				Suite 300
				Portsmouth, VA 23705
				The Miller Firm, LLC
				The Sherman Building
				108 Railroad Avenue
				Orange, VA 22960
Barbara Whittaker	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12cv 00563	Patrick T. Fennell, Esq.
	Virginia	Management, P.C., d/b/a Insight Imaging,		Law Offices of Crandall & Katt
		Roanoke		366 Elm Avenue, S.W.
				Roanoke, VA 24016

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 24 of 31 Exhibit A2

# **State Court Cases Pending Removal**

Plaintiff(s)	Venue	Defendant(s)	Docket	Plaintiff(s)' Counsel
			No.	
Virginia Lois Neely	Davidson County, TN	NECP d/b/a NECC	13C66	Douglas E. Jones
				Schulman, LeRoy & Bennett, P.C.
				501 Union Street, 7th Floor
				P.O. Box 190676
				Nashville, TN 37219-0676

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 25 of 31 Exhibit A3

Plaintiff(s)	Venue	Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Sara Culp Andrews	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2495	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
David A. Barley, Jr.	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2437	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Virginia Law Berger	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2439	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Viola Mills Bowling	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2453	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Brenda Brewster	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2438	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Wilma Brumfield	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2399	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Kathleen Cooley and Robert	Middlesex County, MA	NECP d/b/a NECC; Barry Cadden	12 4980-B	Harry Herzog
Cooley				Herzog & Carp
				427 Mason Park Boulevard
				Katy, TX 77450
				Law Offices of Thomas R. Murphy, LLC
				133 Washington Street, 2nd Floor
				Salem, MA 01970

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 26 of 31 Exhibit A3

Plaintiff(s)	Venue	Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Harry Michael English	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2435	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Carolyn Fidler, Curtis Lynn	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2509	Cranwell, Moore & Emick, P.L.C.
Shell, Brian Glenn Caddell,		Management, P.C., d/b/a Insight Imaging		111 West Virginia Avenue
Richard Wayne Henegar, Lynda				Vinton, VA 24179
Bradfield, Elizabeth P.				
Kirkpatrick, Kathy N. Cole, Dana				
M. King				
Christy Fralin	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2434	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Robin Annette Hannabass	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2436	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Magaret Hanson	Middlesex County, MA	NECP d/b/a NECC	MICV2012-	Frank Prokos, Esq.
			04645-F	Law Office of Frank Prokos, LLC
				1600 Providence Highway
				Walpole, MA 02081
William E. Johnson and Sandra	Davidson County, TN	NECP d/b/a NECC	12C4920	Randall L. Kinnard
J. Johnson				Daniel L. Clayton
				Kinnard, Clayton & Beveridge
				127 Woodmont Boulevard
				Nashville, TN 37205
Donna Montee	Middlesex County, MA	NECP d/b/a NECC	MICV2012-	Kimberly Dougherty, Esq.
			04861	Janet, Jenner & Suggs, LLC
				75 Arlington Street
				Suite 500
				Boston, MA 02116

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 27 of 31 Exhibit A3

Plaintiff(s)	Venue	Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Jocelyn Kae Norris	Cumberland County, TN	NECP d/b/a NECC	CV0005650	Scott & Cain
				550 West Main Avenue
				Suite 601
				Knoxville, TN 37902
James Palmer and Michelle	Cumberland County, TN	NECP d/b/a NECC	CV0005649	Scott & Cain
Palmer				550 West Main Avenue
				Suite 601
				Knoxville, TN 37902
Nancy Phillips	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2398	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Dana Scott Pruitt and Lesa	Davidson County, TN	NECP d/b/a NECC	12C4921	Randall L. Kinnard
Shadowhawk, surviving adult				Daniel L. Clayton
children of Elizabeth A. Pruitt				Kinnard, Clayton & Beveridge
(deceased)				127 Woodmont Boulevard
				Nashville, TN 37205
Craig Simas and Joan Simas	Middlesex County, MA	NECP d/b/a NECC; Ocean State Pain	12 4591	Wickstrom Morse, LLP
		Management, Inc		60 Church Street
				Whitinsville, MA 01588
Kathy Sinclair	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2397	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Reba M. Skelton and Roy H.	Davidson County, TN	NECP d/b/a NECC	12C4918	Randall L. Kinnard
Skelton				Daniel L. Clayton
				Kinnard, Clayton & Beveridge
				127 Woodmont Boulevard
				Nashville, TN 37205
Lloyd Smith	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2454	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 28 of 31 Exhibit A3

Plaintiff(s)	Venue	Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Margaret Snopkowski	Suffolk County, MA	NECP d/b/a NECC	SUCV2012-	Elizabeth A. Kaveny
			03872	Burke Wise Morrissey & Kaveny
				161 North Clark Street
				Suite 3250
				Chicago, IL 60601-3330
Donna West	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2396	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 29 of 31 Exhibit A4

# **Cases Naming Non-Debtor Defendants Only**

Case Name	Venue	Defendants	Plaintiff(s)' Counsel
Edward Adams	USDC- District of Massachusetts	Barry Cadden; Lisa Cadden; Gregory	Frederic L. Ellis
	Adversary No: 13-01026	Conigliaro; Glenn Chin; ARL Bio	Edward D. Rapaki
	Civil Docket: 1:13-cv-10229	Pharma, Inc. d/b/a Analytical	Joseph M. Makalusky
		Research Laboratories; Ameridose,	Ellis & Rapaki, LLP
		LLC; Medical Sales Management, Inc.;	85 Merrimac Street
		GDC Properties Management, Inc.	Suite 500
			Boston, MA 02114
			J. Douglas Peters
			Charfoos & Christensen, PC
			Hecker-Smiley Mansion
			5510 Woodward Avenue
			Detroit, MI 40202
Bradley	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Epperly	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Filson	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Foutz	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 30 of 31 Exhibit A4

## **Cases Naming Non-Debtor Defendants Only**

Case Name	Venue	Defendants	Plaintiff(s)' Counsel
Harris	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Kalinoski	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
May	Davidson County, TN	Saint Thomas Outpatient Neurological	William D. Leader
		Center, LLC; Howell Allen Clinic A	George H. Nolan
		Professional Corporation	Leader, Bulso & Nolan, PLC
			414 Union Street, Suite 1740
			Nashville, TN 37219
McFarlane	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Reed	Davidson County, TN	Saint Thomas Outpatient Neurological	William D. Leader
		Center, LLC; Howell Allen Clinic A	George H. Nolan
		Professional Corporation	Leader, Bulso & Nolan, PLC
			414 Union Street, Suite 1740
			Nashville, TN 37219
			William H. Lassiter, Jr.
			John Overton Belcher
			Lassiter Tidwell & Davis, PLLC
			One Nashville Place
			150 4th Avenue North, Suite 1850
			Nashville, TN 37219

# Case 1:13-md-02419-FDS Document 38 Filed 03/10/13 Page 31 of 31 Exhibit A4

# **Cases Naming Non-Debtor Defendants Only**

Case Name	Venue	Defendants	Plaintiff(s)' Counsel	
Leonard Shaffer and	USDC- District of Massachusetts	Barry Cadden; Lisa Cadden; Gregory	Frederic L. Ellis	
Karen Shaffer	Adversary No: 13-01023	Conigliaro; Glenn Chin; ARL Bio	Edward D. Rapaki	
	Civil Docket: 1:13-cv-10226	Pharma, Inc. d/b/a Analytical	Joseph M. Makalusky	
		Research Laboratories; Ameridose,	Ellis & Rapaki, LLP	
		LLC; Medical Sales Management, Inc.;	85 Merrimac Street	
		GDC Properties Management, Inc.	Suite 500	
			Boston, MA 02114	
			J. Douglas Peters	
			Charfoos & Christensen, PC	
			Hecker-Smiley Mansion	
			5510 Woodward Avenue	
			Detroit, MI 40202	
Smith, James Wirt	Roanoke, VA	Image Guided Pain Management,	H. David Gibson	
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore	
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast	
		Corporation, P.C.	P.O. Box 40013	
			Roanoke, VA 24022-0013	
Smith, Randolph	Roanoke, VA	Image Guided Pain Management,	H. David Gibson	
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore	
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast	
		Corporation, P.C.	P.O. Box 40013	
			Roanoke, VA 24022-0013	
Whitlow	Roanoke, VA	Image Guided Pain Management,	H. David Gibson	
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore	
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast	
		Corporation, P.C.	P.O. Box 40013	
			Roanoke, VA 24022-0013	
Wingate	Roanoke, VA	Image Guided Pain Management,	H. David Gibson	
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore	
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast	
		Corporation, P.C.	P.O. Box 40013	
			Roanoke, VA 24022-0013	

# EXHIBIT B

In re: Ephedra Products Liability Litigation, Case Management Order No. 1

UNITED STATES DISTRICT COURT SOUTHERN DISTRICT OF NEW YORK		
	-X	
IN RE: EPHEDRA PRODUCTS LIABILITY LITIGATION	:	04 M.D. 1598 (JSR)
	: X	
PERTAINS TO ALL CASES	11	

# CASE MANAGEMENT ORDER NO. 1

JED S. RAKOFF, U.S.D.J.

## I. <u>INTRODUCTION</u>

Prior the creation of the above-captioned multi-district proceeding, there had already been transferred to this Court, pursuant to 28 U.S.C. § 157(b)(5), approximately 65 civil actions making ephedra-related personal injury and wrongful death claims against bankruptcy debtors TL Administration Corp. (f/k/a Twinlab Corporation) and TL Administration Inc. (f/k/a Twin Laboratories Inc.) (collectively, "TL"). These cases, which are before the Court for all purposes, including trial, are hereinafter referred to as the "Twinlabs cases." At an initial case management conference in these cases held on March 1, 2004, the Court directed the parties to submit an agreed upon Case Management Order that would have these cases trial-ready by April, 2005. Thereafter, on April 13, 2004, the Judicial Panel on Multidistrict Litigation transferred 14 ephedra products liability cases (the "MDL cases") to this Court, noting that some 200 more ephedra cases were likely to be transferred here under 28 U.S.C. § 1407 (the "Tag-along cases"). By Order dated April 19, 2004, the Court consolidated the Twinlabs and MDL cases for all pre-trial purposes under the above caption.

USDC SDNY
DOCUMENT
ELECTRONICALLY FILED
DOC #:
DATE FILED:

On April 21, 2004, the parties to the Twinlabs cases submitted a 31-page draft Case Management Order No.1, which reflected their agreement on most points and disagreements chiefly about pretrial procedure for experts and for claiming privileges. The draft also sought to take account of the MDL transfer and consolidation, subject, however, to modification after counsel in the MDL cases are heard from in a conference now scheduled for May 13, 2004. After review and consideration of the parties' draft, the Court hereby substantially (although not entirely) adopts the points agreed to by the parties, resolves the disagreements among them, and hereby propounds the following case management order:

## II. GENERAL PROVISIONS

### A. Scope

This Order applies to all pretrial proceedings in the Twinlabs cases (including any future cases against TL transferred pursuant to 28 U.S.C. § 157(b)(5), hereinafter the "Subsequent Cases")), the MDL cases, and the Tag-along cases.

#### **B.** Special Master

By Order dated March 4, 2004, the Court appointed James Niss, Esq., as Special Master to help coordinate and oversee pretrial proceedings in the Twinlabs Cases (including any Subsequent Cases). In this capacity, Mr. Niss functions in effect as an arm of the Court, and accordingly the Court will engage in such ex parte conversations with the Special Master as the Court finds useful. See transcript, 3/1/04 (all parties consenting to this procedure). By Order dated April 19, 2004, the Court extended the Special Master's duties to the MDL Cases (including any Tag-along Cases). While all his fees prior to that date will be paid, at least initially, as an administrative expense of the

TL bankruptcy, the Court expects to allocate a portion of his fees subsequent to that date to the parties in the MDL cases, such reallocation to be resolved at the hearing on May 13, 2004.

The Court hereby refers all applications relating to discovery or extensions of time to the Special Master for initial rulings thereon. Such applications shall be submitted by e-mail to the Special Master with copies to all interested parties. Any affected party may ask the Court to review any such ruling of the Special Master by arranging a conference call with the Court and all interested parties as provided in Judge Rakoff's Individual Practices, provided that such conference call is initiated (or attempted to be initiated, see Judge Rakoff's Individual Practices, Rule 2) within two business days following the date of the Special Master's email ruling from which review is sought. Failure to convene (or attempt to convene) such a conference within said period of two business days constitutes a waiver of any and all objections to the applicable ruling of the Special Master. The Court's review of the Special Master's rulings will be de novo, but the Court expects that counsel will exercise professional restraint in deciding whether or not to seek review of the Special Master's rulings.

For non-substantive administrative matters, a party may contact the Special Master by e-mail without copying other parties. If the Special Master believes other parties to be interested in the communication, he shall send the parties potentially interested a copy of his answer together with a copy of the original communication.

# C. Plaintiffs' Coordinating Counsel

By Order dated March 4, 2004, the Court directed counsel for any plaintiff objecting to the designation of certain attorneys as coordinating counsel for plaintiffs, or seeking to be included in such group, to submit written position papers to the Court by no later than March 8, 2004. Having considered the written submissions (including an exparte submission made, with the permission of the Court, in which a prior objection to the designation of these attorneys was withdrawn), the Court designates the following counsel as Plaintiffs' Coordinating Counsel:

Anne Andrews, Esq. Andrews & Thornton 820 N. Parton Street, Second Floor Santa Ana, CA 92701 714-565-7555 aa@andrewsthornton.com

Edward F. Blizzard, Esq. Blizzard, McCarthy & Nabers LLP 440 Louisiana Street, Suite 1710 Houston, TX 77002 713-844-3750 eblizzard@blizzardlaw.com

Paul Rheingold, Esq. Rheingold, Valet, Rheingold Shkolnik McCartney LLP 113 East 37<sup>th</sup> Street New York, New York 10016 212-684-1880 prheingold@rheingoldlaw.com

Christopher A. Seeger, Esq. Seeger Weiss LLP One William Street New York, New York 10004 212-584-0700 cseeger@seegerweiss.com Upon suggestion of the Plaintiffs' Coordinating Counsel, the Court designates Mr. Seeger as Plaintiffs' Liaison Counsel. Among the duties of the Plaintiffs' Liaison Counsel and the Defendants' Liaison counsel (see infra) will be to make themselves available for any telephone conferences convened by the Court and to communicate the substance of any such telephone conference to all other affected counsel on their respective sides of the case.

Any plaintiffs' counsel in the MDL cases desiring to become part of the group serving as Plaintiffs' Coordinating Counsel shall submit to the Court and serve upon all parties a written application for his/her appointment within seven days after entry of this Order or, for plaintiffs' counsel in any Tag-along cases, within seven days after receiving notice of the order of transfer of the Tag-along action to this Court. Any objections to the application shall be filed and served upon all parties within four days after service of the application.

Plaintiffs' Coordinating Counsel shall be generally responsible for coordinating the activities of plaintiffs during pretrial proceedings and shall

- (1) Determine (after consultation with other co-counsel as may be appropriate) and present (in briefs, oral argument, or such other fashion as may be appropriate, personally or by a designee) to the Court and opposing parties the position of the plaintiffs on all matters arising during pretrial proceedings;
- (2) Coordinate the initiation and conduct of discovery on behalf of plaintiffs consistent with the requirements of Fed. R. Civ. P. 26(b)(1) and (2), and (g), including the preparation of joint interrogatories and requests for production of documents and the examination of witnesses in depositions;
- (3) Conduct settlement negotiations on behalf of plaintiffs without entering binding agreements except to the extent expressly authorized, provided that nothing in this Order shall prohibit individual counsel from

- conducting separate settlement negotiations for those cases in which such counsel represents a particular plaintiff;
- (4) Delegate specific tasks to other counsel in a manner to ensure that pretrial preparation for the plaintiffs is conducted effectively, efficiently, and economically;
- (5) Enter into stipulations with opposing counsel necessary for the conduct of the litigation, except that any stipulation regarding an individual action must be agreed to by the plaintiff's counsel in that action;
- (6) Perform such other duties as may be incidental to proper coordination of plaintiffs' pretrial activities or authorized by further order of the Court;
- (7) Keep the Clerk's office informed by e-mail of any change in the name or address (including e-mail address) of the counsel of record for any plaintiff, and inform new plaintiffs' attorneys of their obligation to register for ECF as provided in ¶ 3 of the 4/19/04 Order;
- (8) Receive and, as appropriate, distribute to co-counsel communications from the Court or Special Master and documents from opposing and other counsel that are not filed or to be filed with the Court; and
- (9) Maintain and make available to co-counsel at reasonable hours a complete file of all documents served or produced by, upon or to each party except those documents that are available to all parties in the Court docket.

Counsel for plaintiffs who disagree with a position taken by Plaintiffs' Coordinating Counsel (or by those acting on behalf of Plaintiffs' Coordinating Counsel), or who have individual or divergent positions, may present written or oral arguments, conduct examinations of deponents, and otherwise act separately on behalf of their client(s) as appropriate, provided that in doing so their actions comport with the Federal Rules of Civil Procedure and orders of this Court, and that they do not repeat arguments, questions, or actions of Plaintiffs' Coordinating Counsel.

#### D. <u>Defendants' Coordinating Counsel</u>

The Court designates the following counsel as Defendants' Coordinating Counsel:

Robert M. Hirsh, Esq. Duane Morris, LLP 380 Lexington Avenue New York, New York 10168 PH: (212) 692-1035

FAX: (212) 692-1020

Rex A. Littrell, Esq.
Ulmer & Berne LLP
88 East Broad Street, Suite 1600
Columbus, Ohio 43215
PH: (614) 229-0012
FAX: (614) 229-0013

Peter C. Neger, Esq. Bingham McCutchen 399 Park Avenue New York, New York 10022-4689 212-705-7226 peter.neger@bingham.com

Joseph P. Thomas, Esq. Ulmer & Berne LLP 600 Vine Street, Suite 2800 Cincinnati, Ohio 45202 PH: (513) 698-5000 FAX: (513) 698-5001

The Court also designates Mr. Littrell as Defendants' Liaison Counsel, with the same duties as Plaintiffs' Liaison Counsel (see supra).

Any defendants' counsel in the MDL cases desiring to act as Defendants' Coordinating Counsel shall submit to the Court and serve upon all parties a written application for his/her appointment within seven days after entry of this Order or, for counsel in any Tag-along cases, within seven days after receiving notice of the order of transfer of the Tag-along action to this Court. Any objections to the application shall be filed and served upon all parties within four days after service of the application.

Defendants' Coordinating Counsel shall be generally responsible for coordinating the activities of the Defendants during pretrial proceedings and shall:

- (1) Determine (after consultation with other co-counsel as may be appropriate) and present (in briefs, oral argument, or such other fashion as may be appropriate, personally or by a designee) to the Court and opposing parties the position of the defendants on all matters arising during pretrial proceedings;
- (2) Coordinate the initiation and conduct of discovery on behalf of the defendants consistent with the requirements of Fed. R. Civ. P. 26(b)(1) and (2), and (g), including the preparation of joint interrogatories and requests for production of documents and the examination of witnesses in depositions;
- (3) Conduct settlement negotiations on behalf of the defendants, without entering binding agreements except to the extent expressly authorized, provided that nothing in this Order shall prohibit individual counsel from conducting separate settlement negotiations for any case in which such counsel represents a particular defendant;
- (4) Delegate specific tasks to other counsel in a manner to ensure that pretrial preparation for the defendants is conducted effectively, efficiently, and economically;
- (5) Enter into stipulations with opposing counsel necessary for the conduct of the litigation, except that any stipulation regarding an individual action must be agreed to by each defendant's counsel in that action;
- (6) Perform such other duties as may be incidental to proper coordination of the defendants' pretrial activities or authorized by further order of the Court;
- (7) Keep the Clerk's office informed by e-mail of any change in the name or address (including e-mail address) of the counsel of record for any defendant, and inform new defendants' attorneys of their obligation to register for ECF as provided in ¶ 3 of the 4/19/04 Order;
- (8) Receive and, as appropriate, distribute to co-counsel communications from the Court or Special Master and documents from opposing and other counsel that are not to be filed with the Court; and
- (9) Maintain and make available to co-counsel at reasonable hours a complete file of all documents served or produced by, upon or to each party except those documents that are available to all parties in the Court docket.

Counsel for defendants who disagree with a position taken by Defendants' Coordinating Counsel (or those acting on behalf of Defendants' Coordinating Counsel), or who have individual or divergent positions, may present written or oral arguments, conduct examinations of deponents, and otherwise act separately on behalf of their client(s) as appropriate, provided that in doing so their actions comport with the Federal Rules of Civil Procedure and orders of this Court, and that they do not repeat arguments, questions, or actions of Defendants' Coordinating Counsel.

#### E. <u>Privileges Preserved</u>

No communication made solely among plaintiffs' counsel or made solely among defendants' counsel shall be taken as a waiver of any privilege or protection to which a given plaintiff or a given defendant would otherwise be entitled.

#### III. STATUS CONFERENCES, ECF, COURT NOTICES

#### A. Status Conferences

The next status conference shall be held on May 13, 2004, at 3 p.m. in Courtroom 14-B, 500 Pearl Street, New York, NY. Thereafter, status conferences shall be held in the same courtroom at 4:30 p.m. on the first Thursday of every month except August. If any counsel of record does not attend the status conference, the Court will assume that attending counsel of the applicable Coordinating Committee is authorized to speak on such non-attending counsel's behalf for all purposes.

#### B. <u>Electronic Case Filing (ECF)</u>

As provided in the Order of April 19, 2004, all motions and other papers required to be filed with the Court shall be filed electronically on the Court's ECF system, and all counsel of record must register for ECF by following the instructions at

http://www.nysd.uscourts.gov/cmecf/cmecfattyreg.htm. Counsel in Tag-along cases are directed to register immediately upon notice of the transfer.

All papers filed shall be filed under both the master docket number 04 M.D. 1598 and the docket number for each individual case to which they pertain.

#### C. Notice to Parties by the Court

Notice by the Court to Plaintiffs' Coordinating Counsel and Defendants' Coordinating Counsel of any matter, ruling, order, schedule, or court hearing relating to all actions shall operate as notice to all parties in these coordinated proceedings.

# IV. MOTIONS, CASE CAPTIONS, SERVICE, COMMUNICATIONS WITH COURT

#### A. Motions

Notwithstanding the notice-of-motion procedure set forth in Rule 2(d) of Judge Rakoff's Individual Practices (which otherwise govern), motions -- other than any motion pursuant to Fed. R. Civ. P. 26-37. and any request for extension of time, both of which are governed by the Special Master provisions supra -- may be filed on the ECF system without seeking prior permission. Counsel shall arrange among themselves to deliver to the Court a complete set of courtesy copies on the day when the last paper is filed. Except as otherwise provided infra for "Daubert" and summary judgment motions, the motion briefing schedule set forth in Local Civil Rule 6.1(b) shall apply and oral argument on any motion shall be heard at the next monthly status conference that occurs at least seven days following the completion of briefing.

#### B. Case Captions

All papers filed with the Court shall have the following caption:
UNITED STATES DISTRICT COURT

SOUTHERN DISTRICT OF NEW YORK	-X	
IN RE: EPHEDRA PRODUCTS LIABILITY LITIGATION	:	04 M.D. 1598 (JSR)
	: -X	

If the papers pertain to the cases generally, the papers shall bear the notation "PERTAINS TO ALL CASES" directly below the caption box. If the papers pertain to some but not all of the cases, or have special applicability to particular cases, all papers must be filed electronically under both 04 M.D. 1598 (JSR) and the individual case number(s) to which they specially pertain. In such a case, in place of or in addition to the words "PERTAINS TO ALL CASES," the caption shall read: "[ALSO] PERTAINS TO \_\_\_\_\_\_," giving the docket numbers of the individual case(s).

# VI. PARTIAL WITHDRAWAL OF THE REFERENCE OF EPHEDRA PERSONAL INJURY CLAIMS IN THE BANKRUPTCY AND THE FILING OF NEW LAWSUITS AGAINST TWINLABS

A. The "TL Non-Plaintiff Claims" refers to ephedra-related personal-injury and wrongful-death claims timely filed in the Chapter 11 cases entitled *In re TL Administration Corporation, et al. (f/k/a Twinlab Corporation, et al.)*, Case No. 03-15564 (Bankr. S.D.N.Y.) (CB) (Jointly Administered) for which the claimant is <u>not</u> presently a plaintiff before this Court. The Court has previously withdrawn the underlying reference of the Chapter 11 proceedings to the extent necessary to avoid interference with the expeditious conduct and trial of the TL consolidated cases presently before this Court (the "TL Plaintiff Claims"). <u>See</u> Order, 3/4/04. Additionally, the Court now withdraws the underlying reference of the Chapter 11 proceedings with respect to the TL Non-Plaintiff Claims to the extent necessary to avoid any interference with the implementation

of this Case Management Order or any proceedings arising therefrom. To this end, the Court hereby orders the transfer of the TL Non-Plaintiff Claims to this Court for the limited purpose of implementing the claim-determination procedures set forth <u>infra</u>.

- B. Debtors' counsel shall serve, by first class mail, upon each TL Non-Plaintiff Claimant a written notice of said claimant's obligations and time deadlines thereunder (as described infra) within 7 days after approval by the Court of the Fact Sheet described in ¶ IX.A. below. The service of such notice shall be without prejudice to the rights of Debtors to object to any claim for any reason, including, without limitation, the untimely filing of a claim.
- Shall convert it into a civil action by filing a complaint to which is appended a verified Fact Sheet (as provided in Section IX infra) within 60 days from the mailing of the notice provided in ¶ B supra. Any TL Non-Plaintiff Claim which is not so converted within the prescribed or extended time shall be deemed waived and treated as expunged, invalid and/or disallowed. Claimants filing a complaint shall pay the filing fee for commencing a civil action. TL and the defendants bound by actions of the Defendants' Coordinating Counsel, by having joined in the request for mandatory conversion as provided in this paragraph, are deemed to have waived service of the summons that would otherwise be required by Rule 4 Fed. R. Civ. P.
- D. Until otherwise directed by this Court, TL and other Defendants shall take no action to respond to a complaint for a converted case or otherwise defend the action, except as necessary to review the information provided to provide insurance carriers notice of and information regarding the action. A plaintiff in a converted case may ask

the Special Master to classify it as a "pre-2002 case." Such application must be made by e-mail, with supporting attachments, sent on or before the date for filing the complaint. Upon such classification by the Special Master, the first sentence of this paragraph shall not apply to that case, and ¶ VII.B. below shall apply.

E. Plaintiffs' Coordinating Counsel shall present a tally and qualitative summary of the converted cases at the status conference to be held in July, 2004.

#### VII. TEMPORARY STAY OF CERTAIN PENDING ACTIONS

#### A. 2002-2004 Cases

A list of the Twinlabs cases which raise claims made in the TL insurance policy years 2002, 2003, and 2004 for occurrences in 2001, 2002, 2003, and 2004 and claims for which coverage has been denied (the "2002-2004 Cases") is attached hereto as Exhibit A. In accordance with the parties' request, the Court hereby temporarily stays discovery with respect to, and the obligation of TL and any co-defendants to defend, the 2002-2004 cases. Instead, they shall proceed as follows:

- (1) Each plaintiff in the 2002-2004 Cases shall serve a Fact Sheet in accordance with Section IX of this Order within 45 days from the date on which the Fact Sheet Form is approved by the Court. However, this subparagraph does not apply to the converted cases described in Section VI, <u>supra</u> (the "Converted Cases"), wherein Fact Sheets must be served with the complaint pursuant to paragraph VI.C. <u>supra</u>.)
- (2) TL and all other relevant defendants shall take no further action to defend the 2002-2004 Cases until after the July 2004 status conference, at which the parties shall advise the Court of the status of the 2002-2004 Cases and the parties' respective positions regarding a possible extension of this stay of defense.

Claimants and the General Creditors' Committee in the Chapter 11 proceedings, shall confer and negotiate in good faith, with the assistance of the Honorable Cornelius Blackshear, United States Bankruptcy Judge, Southern District of New York, to devise alternative claims disposition procedures for (i) the 2002-2004 Cases and (ii) such Converted Cases as the parties reasonably believe should be included in the 2002-2004 Cases. The parties shall report to Judge Blackshear regarding their efforts by no later than ten days after entry of this Order.

#### B. Pre-2002 Cases

All Pre-2002 Cases shall conduct discovery and other pretrial proceedings on the schedule set forth in this Case Management Order No. 1 and subsequent orders of the Court.

#### C. Subsequent Cases

Within 14 days of the filing date in this Court (the "Filing Date") of any Subsequent Case (as defined <u>supra</u>), transferred after entry of this Order, Plaintiffs' Coordinating Counsel and Defendants' Coordinating Counsel shall advise the Special Master by e-mail whether such action shall be included in the 2002-2004 Cases or the Pre-2002 Cases. If the Special Master determines that the transferred action should be included in the 2002-2004 Cases, the action shall be handled in accordance with Paragraph VII.A.

# VIII. <u>JOINDER OF PARTIES, AMENDMENT OF PLEADINGS, RULE 12 MOTIONS, ANSWERS, PENDING MOTIONS, AND MOTIONS REGARDING GENERAL EVIDENTIARY ISSUES</u>

#### A. **Joinder of Parties**

Joinder of parties in all actions must be accomplished within 30 days after the date of this Order (for pending cases), or after the Filing Date (for Subsequent Cases), or after notice of transfer (for Tag-along cases).

#### B. **Amendment of Pleadings**

Amended pleadings may be filed without leave of Court within 30 days after (1) the date of this Order (for pending cases), (2) the Filing Date (for Subsequent Cases), or (3) notice of transfer (for Tag-along cases). If the pleading adds or removes any party, there shall be appended thereto a list of the parties added and removed for each action affected. If the pleading adds a party that is not named in any of the consolidated cases, there shall be appended to the pleading a separate list of such newly named parties.

#### C. Rule 12 Motions

Motions under Fed. R. Civ. P. 12 may be filed within 30 days after (1) the date of this Order (for pending cases), (2) the Filing Date (for Subsequent Cases), or (3) notice of transfer (for Tag-along cases).

#### D. Answers

Each defendant who has been served a complaint but has not yet answered shall answer or move under Rule 12 within 30 days after (1) the date of this Order (for pending cases), (2) the Filing Date (for Subsequent Cases), or (3) notice of transfer (for Tagalong cases).

#### E. <u>Pending Motions</u>

No later than May 13, 2004, i.e., the date of the next status conference, the parties in each individual case must advise the Court in writing of any pending motions filed in the transferor court that require decision by this Court. In the written submission, the parties must advise the Court if the pending motions require revised briefing under the Federal Rules of Civil Procedure, the Federal Rules of Evidence, or other applicable federal law.

#### IX. NON-EXPERT DISCOVERY PROVIDED BY PLAINTIFFS

- A. On or before April 30, 2004, Plaintiffs' Coordinating Counsel and Defendants Coordinating Counsel shall present to the Court for approval a form Fact Sheet and Authorization for Release of Medical Records ("Fact Sheet") suitable for use in all ephedra cases. The Fact Sheet shall include a form medical authorization in compliance with the Health Insurance Portability and Accountability Act. The final form of the Fact Sheet shall be determined by the Court.
- B. Except as set forth in subsection (4) below, each plaintiff shall provide verified answers to the Fact Sheet and an executed medical authorization within 45 days after approval of the Fact Sheet (for pending cases), or after the Filing Date (for Subsequent Cases), or after notice of transfer (for Tag-along cases).
- C. Except as provided in Section VI for the Converted Cases and Section VII for the 2002-2004 Cases, each plaintiff shall produce the documents requested by the Fact Sheet within 60 days after approval of the Fact Sheet (for pending cases), or 30 days after the Filing Date (for Subsequent Cases), or 30 days after notice of transfer (for Tagalong cases).

- D. If a plaintiff has provided sworn answers to prior interrogatories responding to all information requested in the Fact Sheet, that plaintiff shall not be required to provide answers to the Fact Sheet. If a plaintiff has provided sworn responses to prior interrogatories responding to some, but not all information requested in the Fact Sheet, that plaintiff shall answer those portions of the Fact Sheet for which no prior interrogatory response has been provided.
- E. A plaintiff's responses to the Fact Sheet shall conform to the standards applicable to interrogatory answers under Fed. R. Civ. P. 33 and production of documents responses under Fed. R. Civ. P. 34. Each plaintiff shall remain bound by the supplementation duties imposed by Fed.R.Civ.P. 26(e). The questions contained in the Fact Sheet shall not be limited by the restrictions of S.D.N.Y. Local Civil Rule 33.3(a).

#### X. NON-EXPERT DISCOVERY PROVIDED BY DEFENDANTS

A. On or before May 20, 2004, Plaintiffs' Coordinating Counsel and Defendants' Coordinating Counsel shall serve and file a proposed Master Set of Requests for Production of Documents and Interrogatories (the "Master Discovery") in a form suitable for all ephedra cases. (The limitations imposed by the S.D.N.Y. Local Rule 33.3(a) on the scope of initial interrogatories shall not here apply.) Plaintiffs' Coordinating Counsel and Defendants' Coordinating Counsel shall meet and confer regarding (a) the possibility of utilizing prior document production requests from the transferred actions to serve in whole or in part as document production requests in the Master Discovery, and (b) the possibility of utilizing prior document productions from the transferred actions to serve as responses to the document production requests in the Master Discovery. Parties to whom the proposed Master Discovery is propounded may

serve and file objections within five business days after service of the proposed requests.

After reviewing the proposal and any objections thereto, the Court shall order the final form of Master Discovery.

B. Defendants in the cases now before this Court shall produce documents responsive to the Master Discovery within 60 days after entry of the order approving the Master Discovery; defendants in Subsequent and Tag-along cases shall produce documents within 90 days after entry unless they apply for an extension of time to the Special Master within 30 days after the Filing Date or notice of transfer and such extension is granted for good cause shown. On the date for producing documents, the producing parties shall provide the information specified in Local Civil Rule 26.2 (the "privilege log") for each document withheld because of a claim of privilege. Objections to privileges are waived unless the objecting party requests a ruling by the Special Master within ten days after the information specified in Local Civil Rule 26.2 is provided in writing by the party claiming the privilege.

#### XI. THIRD-PARTY DOCUMENT DISCOVERY

#### A. Plaintiffs' Medical Records

Prior to using any medical authorization furnished by a plaintiff as part of that plaintiff's Fact Sheet submission in order to obtain medical records or other documents with respect to a plaintiff, the person using such authorization shall provide to the plaintiff's counsel, or to the plaintiff if unrepresented, the names and addresses of the persons to whom the authorizations will be addressed. A plaintiff shall have 15 days to object to the use of such authorization. If such an objection is made, the authorization shall not be used to request medical records from such medical provider or third person

until the objection is resolved by consent or ruling of the Special Master. Upon request of a plaintiff's counsel and/or a co-defendant's counsel, defendants' counsel shall provide copies of the records requested to plaintiff's counsel and/or co-defendant's counsel at a reasonable cost.

#### B. Third Party Document Production Requests

Subject to the limitations contained in the prior paragraph, any party may request production of documents by a third party through a subpoena duces tecum. The party initiating such discovery shall provide, at a reasonable cost, copies of all documents obtained to any counsel in that action who requests copies of the documents. No such subpoena may be served fewer than 60 days before the Fact Discovery Completion Date as provided below.

# XII. NON-EXPERT DEPOSITIONS, REQUESTS TO ADMIT, EXPEDITED DISCOVERY, DISCOVERY COMPLETION DATE

#### A. Non-Expert Depositions

- (1) All non-expert depositions must be completed by January 21, 2005, unless, in the case of Subsequent and Tag-along cases, the parties apply for an extension of time to the Special Master within 30 days after the Filing Date or notice of transfer and such extension is granted.
- (2) No witness shall be deposed on the same subject more than once. If a party seeks to take a second deposition of a witness, it shall provide the opposing party its basis for an exception to the rule along with a list of the subject matters as to which interrogation is sought.

#### B. Present and Former Employees of Defendants

- (1) The defendants shall make available any of their present employees requested by plaintiffs for deposition, subject to the defendants' rights to object to the taking of any particular employee's deposition for good cause shown.
- (2) Plaintiffs shall in good faith take only those depositions of defendants' present and former employees deemed reasonably necessary under the circumstances. Plaintiffs shall coordinate with each other to schedule only one deposition of the same present or former employee of a defendant.
- (3) Defendants shall take reasonable steps to make available requested former employees, to the extent possible. If a defendant is unable, despite its best efforts, to produce a former employee, then the defendant shall provide upon request the former employee's last known address and shall cooperate in any effort to obtain this Court's or another court's assistance to compel the former employee's attendance at deposition. Plaintiffs shall not contact former employees of a defendant without permission of the defendant who employed the former employee.
- (4) Absent consent of the parties or order of this Court, all depositions of former employees of TL shall be conducted in (a) New York City, (b) Long Island, New York, (c) Utah, or (d) the employee's state of residence, to be determined by the preference of the former employee. Absent consent of the parties or ruling by the Special Master, all depositions of former employees of other defendants shall be conducted in the state or country of the given defendant's principal place of business or the former employee's state of residence, to be determined by the preference of the former employee.

#### C. Requests for Admission

Requests for Admission, if any, must be served no later than 30 days before the Fact Discovery Completion Date as provided below.

#### D. Expedited Discovery

Expedited discovery of a plaintiff and of the plaintiff's health care provider(s), including depositions of the same, may be granted, at the option of the Special Master, where all of the following conditions exist: (a) the plaintiff or a member of the plaintiff's family is terminally ill; (b) there is an urgent need to record and preserve the testimony because of the gravity of the illness; (c) the plaintiff has fully completed and served a Fact Sheet and provided and fully complied with the execution of medical authorizations as required by the Fact Sheet; and (d) defendants have had an opportunity to conduct reasonable informal discovery prior to the taking of the deposition. If necessary, the defendants in an action may conduct expedited discovery of a plaintiff, including the deposition of the plaintiff, before the plaintiff has completed and served a Fact Sheet if the other conditions set forth above are satisfied.

#### E. Fact Discovery Completion Date

All fact discovery is to be completed within 9 months from the date of this Order (the "Fact Discovery Completion Date") unless, in the case of Subsequent and Tag-along cases, the parties apply for an extension of time to the Special Master within 30 days after filing or notice of transfer, and such extension is granted. Interim deadlines for other items set forth in this Order may be extended by the parties in an individual action on consent, provided the parties are <u>certain</u> they can still meet the discovery completion date set forth in this paragraph, which shall not be extended except upon a showing of

extraordinary circumstances. Expert discovery shall take place pursuant to the schedule set forth in Section XIV below.

#### XIII. CONFIDENTIALITY

The Court will enter a confidentiality order based on its standard Protective Order to protect confidential and competitively sensitive information. By e-mail application to the Special Master on or before April 30, 2004, the parties may request modifications in the Court's standard protective order, a copy of which will made available to counsel by the Special Master.

#### XIV. EXPERT DISCOVERY AND "DAUBERT" PROCEEDINGS

#### A. In General

Every party-proponent of a claim (including any counterclaim, cross-claim, or third-party claim) that intends to offer expert testimony in support of such claim and every party-opponent that intends to offer expert testimony in opposition to such claim must make the disclosures required by Fed. R. Civ. P. 26(a)(2) in accordance with the requirements of this provision. No expert testimony (whether designated as "rebuttal" or otherwise) will be permitted by other experts or beyond the scope of the opinions covered by the aforesaid disclosures except upon prior express permission of the Court, application for which must be made no later than 10 days after the date for disclosure of expert witnesses by party-opponents as set forth infra. All experts may be deposed, but such depositions must occur within the time limit for depositions set forth infra.

#### B. Generic Expert Witnesses

(1) On or before June 18, 2004, Plaintiffs' Coordinating Counsel shall identify, in a document served on all parties, generic expert witnesses who are reasonably

expected to testify for the plaintiffs on issues of general or widespread applicability ("Generic Experts"), including, but not limited to, experts who will testify on general causation. The plaintiffs may identify up to three Generic Experts with respect to each of the following categories of injuries: (a) ischemic stroke, (b) hemorrhagic stroke, (c) seizures, (d) cardiac injury, (e) psychotic injury, and (f) primary pulmonary hypertension. For each Generic Expert so identified, Plaintiffs' Coordinating Counsel, on behalf of all plaintiffs, shall serve upon all parties the disclosures required by Fed. R. Civ. P. 26(a)(2), except that such disclosures need not include such testimony, if any, that such Generic Expert is expected to offer relating only to liability or damages as to a particular plaintiff.

- (2) Each plaintiff in a Subsequent or Tag-along case shall identify
  Generic Experts and provide all disclosures required by Fed. R. Civ. P. 26(a)(2) (with the
  exceptions contained in the preceding paragraph), on or before July 16, 2004, unless such
  plaintiff applies for an extension of time to the Special Master within 30 days after the
  Filing Date or notice of transfer and such extension is granted.
- (3) Discovery depositions of all Generic Experts identified pursuant to paragraph (1) above shall be completed no later than September 10, 2004. Discovery depositions of all Generic Experts identified pursuant to paragraph (2) shall be completed no later than two months following the date for the designation of such experts.
- (4) On or before August 18, 2004, Defendants' Coordinating Counsel shall identify, in a document served on all parties, Generic Experts who are reasonably expected to testify in response to the Generic Experts disclosed by the plaintiffs pursuant to paragraph (1) above. The defendants may identify up to three Generic Experts with respect to each of the categories of injuries set forth in paragraph (1) above. For each

Generic Expert so identified, Defendants' Coordinating Counsel, on behalf of all defendants, shall serve upon all parties the disclosures required by Fed. R. Civ. P. 26(a)(2), except that such disclosures need not include such testimony, if any, that such Generic Expert is expected to offer relating only to liability or damages as to a particular plaintiff.

- (5) Each defendant in a Subsequent or Tag-along case shall identify
  Generic Experts and provide all disclosures required by Federal Rule of Civil Procedure
  26(a)(2) (with the exceptions contained in the preceding paragraph), on or before
  September 15, 2004, unless such defendant applies for an extension of time to the Special
  Master within 30 days after filing or notice of transfer and such extension is granted.
- (6) Discovery depositions of all Generic Experts identified pursuant to paragraph (4) above shall be completed no later than November 5, 2004. Discovery depositions of all Generic Experts identified pursuant to paragraph (5) above shall be completed no later than two months following the designation of such experts.
- (7) On or before December 10, 2004, every party wishing to challenge the admissibility of an identified Generic Expert's opinion in any case other than a Subsequent or Tag-along Case shall serve and file a motion to exclude such testimony (a "Daubert motion"). For Subsequent or Tag-along cases, every party wishing to challenge the admissibility of such Expert's opinion shall serve and file a Daubert motion no later than three months following the date for designation of such Expert. Answering papers shall be served within 14 days after service of the motion papers, and reply papers within 7 days after service of the answering papers.

- (8) The parties are instructed to avoid duplicative arguments in *Daubert* motions by coordinating arguments with Plaintiffs' Coordinating Counsel, Defendants' Coordinating Counsel, and each other.
- (9) Oral argument on the motions to exclude Generic Expert testimony in any but Subsequent or Tag-along Cases shall be held at 9 a.m. on January 6, 2005. If an evidentiary hearing is required, the Court will so notify counsel that time and set a date for such hearing.

#### C. <u>Case Specific Expert Witnesses</u>

- (1) On or before October 15, 2004, each plaintiff shall identify, in a document served on all parties, each Case Specific Expert witness who is expected to testify for that plaintiff and, except to the extent that such disclosures already have been made, shall make all expert disclosures required by Fed. R. Civ. P. 26(a)(2).
- (2) On or before December 10, 2004, each defendant shall identify, in a document served on all parties, each Case Specific Expert who is expected to testify for that defendant and, except to the extent that such disclosures already have been made, shall make all expert disclosures required by Fed. R. Civ. P. 26(a)(2).
- (3) Discovery depositions of plaintiffs' Case Specific Experts shall be completed by December 10, 2004. Discovery depositions of defendants' Case Specific Experts shall be completed by January 14, 2005.
- (4) On or before January 28, 2005, every party wishing to challenge the admissibility of a Case Specific Expert's opinion shall serve and file a *Daubert* motion. Answering papers shall be served within 14 days after service of the motion papers, and reply papers within 14 days after service of the answering papers. Oral

argument shall be held at 9 a.m. on March 3, 2005. If an evidentiary hearing is required, the Court will so notify counsel at that time and set a date for such hearing. The parties are instructed to avoid duplicative arguments in *Daubert* motions by coordinating arguments with Plaintiffs' Coordinating Counsel, Defendants' Coordinating Counsel, and each other.

#### XV. SUMMARY JUDGMENT MOTIONS

A. Motions for summary judgment may be filed no later than 14 days after the Court decides any relevant *Daubert* motions that were filed as provided above. Answering papers shall be served within 14 days after service of the motion papers, and reply papers within 7 days after service of the answering papers. Oral argument will be heard at the next status conference occurring at least ten business days after the last paper is due.

#### XVI. FINAL PRE-TRIAL CONFERENCE; REMAND; TRIAL

#### A. <u>Pretrial Order; Final Pre-Trial Conferences</u>

Counsel in the Twinlabs cases shall submit a joint pretrial consent order (in the form prescribed by Rule 4 of Judge Rakoff's Individual Rules) by May 2, 2005. The final pretrial conference will be held on May 5, 2005. Except where extensions of time have been granted as set forth above, the MDL and Tag-along cases will be remanded to their district of origin for trial.

#### B. Trial

Trial of the Twinlabs cases will commence on May 9, 2005, with a consolidated trial of all common issues.

#### C. Demand for Jury Trial

Where any complaint in a Twinlabs case transferred to this Court does not contain a demand for jury trial, the plaintiff may serve and file a jury demand within 30 days after

entry of this order or, for Subsequent Cases, after the Filing Date.

XVII. EFFECT ON BANKRUPTCY PROCEEDINGS

Nothing contained herein shall prohibit, hinder, impair or in any way affect the

rights of TL, the Official Committee of Unsecured Creditors, and the Official Committee

of Ephedra Claimants appointed in the Chapter 11 cases of TL, In re TL Administration

Corporation, et al. (f/k/a Twinlab Corporation, et al.), Case No. 03-15564 (Bankr.

S.D.N.Y.) (CB) (Jointly Administered) (the "Chapter 11 Cases"), filed in the United

States Bankruptcy Court for the Southern District of New York, to, inter alia, participate

in or propose a plan of liquidation or reorganization in the Chapter 11 Cases and/or any

pending actions in the Court relating to and encompassed by In re: Twinlabs Personal

Injury Cases, 21 M.C. 99 (S.D.N.Y.) (JSR) (the "District Court Actions"), and/or which

allocates, assigns, marshals, or otherwise treats any insurance coverages which may be

applicable to any such claims in the Chapter 11 Cases or causes of action in the District

Court Actions.

SO ORDERED.

Dated: New York, N.Y. April 26, 2004

27

#### **EXHIBIT A**

- 1. Alcantar v. Twin Laboratories Inc., et al., Case No. 03CV09169
- 2. Alexander v. Twin Laboratories Inc., et al., Case No. 04CV00095
- 3. Ashment v. Twinlab Corp., et al., Case No. 04CV00538
- 4. Baines v. Twin Laboratories Inc., et al., Case No. 04CV00094
- 5. Bonner, et al. v. O'Leary Health Food Distributor Co., Inc., et al., Case No. 03CV09275
- 6. Carrero, et al. v. Twin Laboratories Inc., et al., Case No. 04CV01276
- 7. Filardi, et al. v. Twin Laboratories Inc., et al., Case No. 04CV00093
- 8. Forbes, et al. v. Twin Laboratories Inc., et al., Case No. 03CV09272
- 9. Fulton v. Twin Laboratories Inc., et al., Case No. 03CV09271
- 10. Ganir v. FM, Inc., et al., Case No. 04CV01961
- 11. Garvin v. Twin Laboratories Inc., et al., Case No. 04CV02379
- 12. Garza v. Twin Laboratories Inc., et al., Case No. N/A
- 13. Greve, et al. v. Twin Laboratories Inc., et al., Case No. 03CV9278
- 14. Hawkinson v. Twin Laboratories Inc., et al., Case No. 04CV00539
- 15. Houghton, et al. v. Twin Laboratories Inc., et al., Case No. 03CV09290
- 16. Joyner-Wiggins v. Twin Laboratories Inc, et al., Case No. 03CV09285
- 17. Korizis, et al. v. Twin Laboratories Inc., et al., Case No. 03CV09284
- 18. McMillan, et al. v. Twin Laboratories Inc., et al., Case No. 03CV09267
- 19. Millan v. Twin Laboratories Inc., et al., Case No. 04CV01988
- 20. Parks v. Twin Laboratories Inc., et al., Case No. 04CV01037
- 21. Parsley, et al. v. Metabolife International, Inc., et al., Case No. 03CV09266
- 22. Petrovich, et al. v. Twin Laboratories Inc., et al., Case No. 04CV02498
- 23. Rawert, et al. v. Twin Laboratories Inc., et al., Case No. 03CV09265

#### 

- 24. Rodriguez, Ronnie v. Twin Laboratories Inc., et al., Case No. 03CV09283
- 25. Smith, Jahn v. Twin Laboratories Inc., et al., Case No. 04CV01039
- 26. Welch, et al. v. Twin Laboratories Inc., et al., Case No. 04CV00091
- 27. White, et al. v. Rexall Sundown, Inc., et al., Case No. 03CV09281

#### EXHIBIT C

In re TL Administration Corporation,
First Amended Plan of Reorganization and Confirmation Order

UNITED STATES DISTRICT COURT SOUTHERN DISTRICT OF NEW YORK		
In re	x :	0.4.N.D. 1500 (IGD)
EPHEDRA PRODUCTS LIABILITY LITIGATION.	:	04 MD 1598 (JSR)
PERTAINS TO ALL CASES	X	
UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK	<b>v</b>	
In re	:	
TL ADMINISTRATION CORPORATION,	:	Chapter 11 Case No.
et al. (f/k/a TWINLAB CORPORATION, et al.),	:	03-15564 (RDD)
Debtors.	•	(Jointly Administered)
	X	

FIRST AMENDED JOINT PLAN OF LIQUIDATION OF TL ADMINISTRATION CORPORATION (F/K/A TWINLAB CORPORATION), TL ADMINISTRATION, INC. (F/K/A TWIN LABORATORIES INC.), AND TL ADMINISTRATION (UK) LTD. (F/K/A TWIN LABORATORIES (UK) LTD.)

<u>UNDER CHAPTER 11 OF THE BANKRUPTCY CODE</u>

WEIL, GOTSHAL & MANGES LLP Attorneys for Debtors and Debtors in Possession 767 Fifth Avenue New York, New York 10153 (212) 310-8000

Dated: May 25, 2005

#### Casse 1::148-mdl-023419-FDS | Document 1387-2 | Filed 035/205/135 | Pagge 3: off 742

#### TABLE OF CONTENTS

	Page
SECTION 1	DEFINITIONS AND INTERPRETATION1
1.1	2002-2004 Ephedra PI Claim
1.2	2002-2004 Ephedra PI Settlement Agreement
1.3	Administrative Expense Claim2
1.4	Aggregate Allowed Consumer Class Action Claims2
1.5	Aggregate Third Party Settlement Amount
1.6	AISLIC2
1.7	Allowed2
1.8	Amended Complaint
1.9	Available Cash2
1.10	Avoidance Actions2
1.11	Bankruptcy Code2
1.12	Bankruptcy Court
1.13	Bankruptcy Rule s
1.14	Blechman Agreement
1.15	Blechmans
1.16	Business Day3
1.17	CPLR
1.18	Cash3
1.19	Causes of Action
1.20	Chapter 11 Cases
1.21	Claim
1.22	Class4
1.23	Class 4 Distribution
1.24	Collateral4
1.25	Commencement Date
1.26	Confirmation Date
1.27	Confirmation Hearing4
1.28	Confirmation Order
1.29	Consumer Class Action Claim
1.30	Creditor4
1.31	Creditors Committee
1.32	Debtors or Debtors in Possession
1.33	Declaratory Judgment Complaint
1.34	Disbursing Agent4

#### Casse 1::148-mdl-023419-FDS | Document 1387-2 | Filed 035/205/135 | Pagge 4 of 742

	Page
1.35	Disclosure Statement
1.36	Disclosure Statement Hearing5
1.37	Disclosure Statement Order
1.38	Disputed5
1.39	Distribution Record Date5
1.40	District Court
1.41	Effective Date5
1.42	Entity5
1.43	Ephedra Claimants Committee
1.44	Ephedra Indemnification Claim6
1.45	Ephedra Insurance Policies6
1.46	Ephedra Personal Injury Claim6
1.47	Ephedra Personal Injury Trust6
1.48	Ephedra Personal Injury Trust Agreement
1.49	Ephedra Personal Injury Trustee
1.50	Equity Interest6
1.51	Estates6
1.52	Final Order7
1.53	GNC7
1.54	General Unsecured Claim
1.55	Holdings7
1.56	Ideasphere
1.57	Ideasphere Agreement
1.58	Impaired
1.59	Indenture
1.60	Indenture Trustee
1.61	Initial Cash
1.62	Non-Ephedra Claims
1.63	Objection Deadline
1.64	Other Participating Ephedra Claims
1.65	Person8
1.66	Plan8
1.67	Plan Administrator8
1.68	Pre-2002 Ephedra PI Claim8

#### Casse 1::148-mdl-023419-FDS | Document 1387-2 | Filed 033/205/135 | Pagge 54 off 742

		Page
1.69	Priority Non-Tax Claim	8
1.70	Priority Tax Claim	8
1.71	Pro Rata Share	8
1.72	Punitive Damages	8
1.73	Punitive Damage Claim	9
1.74	Released Parties	9
1.75	Schedules	9
1.76	Secured Claim	9
1.77	Settlement Escrow Agent	9
1.78	Settling Third Party	9
1.79	Tax Code	9
1.80	Third Party Settlement Amount	9
1.81	Treasury Regulations	9
1.82	U.S. Trustee	9
1.83	Voting Procedures Order	9
SECTION 2	ADMINISTRATIVE EXPENSE CLAIMS AND PRIORITY TAX CLAIMS	10
2.1	Administrative Expense Claims	10
2.2	Priority Tax Claims	10
SECTION 3	CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS	10
3.1	Classification of Claims and Equity Interests	10
3.2	Bifurcation of Claims Asserting Punitive Damages	11
SECTION 4	TREATMENT OF CLAIMS AND EQUITY INTERESTS	11
4.1	Class 1 –Secured Claims	11
4.2	Class 2 - Priority Non-Tax Claims	12
4.3	Class 3 – Pre-2002 Ephedra PI Claims	12
4.4	Class 4 – 2002-2004 Ephedra PI Claims	13
4.5	Class 5 – General Unsecured Claims	13
4.6	Class 6 – Punitive Damage Claims	14
4.7	Class 7 – Equity Interests	14
SECTION 5	THE EPHEDRA PERSONAL INJURY TRUST	14
5.1	Responsibilities of the Ephedra Personal Injury Trust	14
5.2	Resolution of Minors' Claims by the Ephedra Personal Injury Trust	14
5.3	Transfer of Property to the Ephedra Personal Injury Trust	15

#### Casse 1::148-mdl-023419-FDS | Document 1387-2 | Filed 035/205/135 | Pagge 6 off 742

		Page
5.4	Tax Treatment of Ephedra Personal Injury Trust	15
5.5	Appointment of Ephedra Personal Injury Trustee	15
SECTION 6	PROVISIONS GOVERNING DISTRIBUTIONS	16
6.1	Distribution Record Date	16
6.2	Delivery of Distributions and Undeliverable Distributions	16
6.3	Surrender of Instruments	16
6.4	Setoffs	17
6.5	Transactions on Business Days	17
6.6	Allocation of Plan Distribution Between Principal and Interest	17
6.7	Withholding and Reporting Requirements	17
SECTION 7	MEANS FOR IMPLEMENTATION AND EXECUTION OF THE PLAN	17
7.1	Deemed Consolidation of Debtors for Plan Purposes Only	17
7.2	The Plan Administrator	18
7.3	Directors/Officers/Assets of the Debtors on the Effective Date	18
7.4	Duties and Powers of the Plan Administrator	19
7.5	Method of Distributions Under the Plan	21
7.6	Procedures for Treating Disputed Claims	22
7.7	Closing of the Chapter 11 Cases	23
7.8	Cancellation of Existing Securities and Agreements	23
7.9	Dissolution	24
7.10	Books and Records	24
7.11	Corporate Action	24
7.12	Effectuating Documents and Further Transactions	24
SECTION 8	EXECUTORY CONTRACTS AND UNEXPIRED LEASES	25
8.1	Executory Contracts and Unexpired Leases	25
8.2	Approval of Rejection of Executory Contracts and Unexpired Leases	25
8.3	Rejection Claims	25
SECTION 9	EFFECTIVENESS OF THE PLAN	25
9.1	Condition Precedent to Confirmation of Plan	25
9.2	Conditions Precedent to Effective Date	25
9.3	Satisfaction of Conditions	26
9.4	Effect of Nonoccurrence of Conditions to Consummation	26
9.5	Waiver of Conditions	27

#### Casse 1::148-mdl-023419-FDS | Document 1387-2 | Filed 035/205/135 | Pagge 75 off 742

		Page
SECTION 10	EFFECT OF CONFIRMATION	27
10.1	Vesting of Assets	27
10.2	Binding Effect.	27
10.3	Retention of Causes of Action	27
10.4	Exculpations and Limitation of Liability	28
10.5	Releases	28
10.6	Injunction	29
10.7	United States Exception	30
10.8	Terms of Pre-Plan Injunction and Stays	31
SECTION 11	RETENTION OF JURISDICTION	31
11.1	Jurisdiction of Bankruptcy Court	31
SECTION 12	MISCELLANEOUS PROVISIONS	32
12.1	Dissolution of Creditors Committee and Ephedra Claimants Committee	32
12.2	Substantial Consummation	33
12.3	Payment of Statutory Fees.	33
12.4	Effectuating Documents and Further Transactions	33
12.5	Exemption from Transfer Taxes	33
12.6	Post-Confirmation Date Fees and Expenses of Professionals	33
12.7	Elimination of Vacant Classes	33
12.8	Nonconsensual Confirmation	34
12.9	Modification of Plan.	34
12.10	Revocation or Withdrawal of Plan	34
12.11	Courts of Competent Jurisdiction	34
12.12	Severability	34
12.13	Governing Law	35
12.14	Exhibits	35
12.15	Successors and Assigns	35
12.16	Time	35
12.17	Notices	35

SOUTHERN DISTRICT OF NEW YORK		
In re	x :	
EPHEDRA PRODUCTS LIABILITY LITIGATION.	•	04 MD 1598 (JSR)
PERTAINS TO ALL CASES	X	
UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK	v	
In re	:	
TILL A DA MANAGERO A TINONA GOOD DODA TINONA	:	Chapter 11 Case No.
TL ADMINISTRATION CORPORATION,	:	0.0.1.0.1.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0
et al. (f/k/a TWINLAB CORPORATION, et al.),	:	03-15564 (RDD)
	:	
	:	(Jointly Administered)
Debtors.	:	
	·X	

LINITED STATES DISTRICT COURT

FIRST AMENDED JOINT PLAN OF LIQUIDATION OF TL ADMINISTRATION CORPORATION (F/K/A TWINLAB CORPORATION), TL ADMINISTRATION, INC. (F/K/A TWIN LABORATORIES INC.), AND TL ADMINISTRATION (UK) LTD. (F/K/A TWIN LABORATORIES (UK) LTD.) UNDER CHAPTER 11 OF THE BANKRUPTCY CODE

TL Administration Corporation (f/k/a Twinlab Corporation), TL Administration Inc. (f/k/a Twin Laboratories, Inc.), and TL Administration (UK) Ltd. (f/k/a Twin Laboratories (UK) Ltd.), the above-captioned debtors and debtors in possession, propose the following chapter 11 plan of liquidation pursuant to section 1121(a) of title 11 of the United States Code:

#### **SECTION 1**

#### **DEFINITIONS AND INTERPRETATION**

**<u>DEFINITIONS</u>**. The following terms used herein shall have the respective meanings defined below (such meanings to be equally applicable to both the singular and plural):

- **1.1** 2002-2004 Ephedra PI Claim means any Ephedra Personal Injury Claim that is not a Pre-2002 Ephedra PI Claim.
- 1.2 <u>2002-2004 Ephedra PI Settlement Agreement</u> means that certain agreement by and among the holders of 2002-2004 Ephedra PI Claims, the holders of Other Participating Ephedra Claims, the Settling Third Parties, and the Debtors, substantially in the form of the agreement attached hereto as Exhibit A.

- Administrative Expense Claim means any right to payment constituting a cost or expense of administration of any of the Chapter 11 Cases allowed under sections 503(b), 507(a)(1), and 1114(e) of the Bankruptcy Code, including, without limitation, any actual and necessary costs and expenses of preserving the Debtors' estates, any actual and necessary costs and expenses of operating the Debtors' businesses, any indebtedness or obligations incurred or assumed by the Debtors, as debtors in possession, during the Chapter 11 Cases, including, without limitation, for the acquisition or lease of property or an interest in property or the rendition of services, any allowances of compensation and reimbursement of expenses to the extent allowed by Final Order under sections 330 or 503 of the Bankruptcy Code, and any fees or charges assessed against the estates of the Debtors under section 1930 of chapter 123 of title 28 of the United States Code.
- 1.4 Aggregate Allowed Consumer Class Action Claims means the aggregate Allowed value, if any, of the Consumer Class Action Claims where the Allowed value of each such Claim is determined by estimation by (i) the Debtors following a settlement agreement with the applicable claimant(s) or (ii) the Bankruptcy Court pursuant to section 502(c) of the Bankruptcy Code.
- 1.5 <u>Aggregate Third Party Settlement Amount</u> means the sum of all Third Party Settlement Amounts.
  - **1.6 AISLIC** means American International Specialty Lines Insurance Company.
- **1.7 Allowed** means, with reference to any Claim, (i) any Claim against any Debtor which has been listed by such Debtor in the Schedules, as such Schedules may be amended by the Debtors from time to time in accordance with Bankruptcy Rule 1009, as liquidated in amount and not disputed or contingent and for which no contrary proof of claim has been filed, (ii) any timely filed Claim as to which no objection to allowance has been interposed in accordance with Section 7.6(a) hereof or such other applicable period of limitation fixed by the Bankruptcy Code, the Bankruptcy Rules, or the Bankruptcy Court, or as to which any objection has been determined by a Final Order to the extent such objection is determined in favor of the respective holder, (iii) any Claim expressly allowed by a Final Order or hereunder, or (iv) any 2002-2004 Ephedra PI Claim or Other Participating Ephedra Claim finally resolved in accordance with the procedures of the Ephedra Personal Injury Trust Agreement.
- **1.8** Amended Complaint means the amended complaint filed by the Creditors Committee and Indenture Trustee on May 7, 2004 in the Bankruptcy Court, Adversary Proceeding No. 04-02334(CB).
  - **1.9 Available Cash** means the Initial Cash minus the Class 4 Distribution.
- **1.10** Avoidance Actions means any actions that may be commenced, before or after the Effective Date, pursuant to sections 544, 545, 547, 548, or 549 of the Bankruptcy Code that existed or may have existed against vendors, suppliers, or customers of the Debtors.
- **1.11** Bankruptcy Code means title 11 of the United States Code, as amended from time to time, as applicable to the Chapter 11 Cases.

- 1.12 <u>Bankruptcy Court</u> means the United States District Court for the Southern District of New York, having jurisdiction over the Chapter 11 Cases and, to the extent of any reference made under section 157 of title 28 of the United States Code, the bankruptcy court unit of such District Court having jurisdiction over the Chapter 11 Cases under section 151 of title 28 of the United States Code.
- 1.13 <u>Bankruptcy Rules</u> means the Federal Rules of Bankruptcy Procedure as promulgated by the United States Supreme Court under section 2075 of title 28 of the United States Code, as amended from time to time, applicable to the Chapter 11 Cases, and any Local Rules of the Bankruptcy Court.
- 1.14 <u>Blechman Agreement</u> means that certain Stipulation and Agreement Regarding Compromise and Settlement of Adversary Proceeding, dated as of November 12, 2004, by and among the Debtors, the Creditors Committee, the Blechmans, John Danhakl, Jonathan Sokoloff, Leonard Schutzman, William Westerfield, U.S. Bank National Association as Indenture Trustee, and National Union Fire Insurance Company of Pittsburgh.
- **1.15** <u>Blechmans</u> means Brian Blechman, Dean Blechman, Neil Blechman, Steve Blechman, Ross Blechman, Robin Blechman, Sharon Blechman, Helena Blechman, Linda Blechman, and Elyse Blechman.
- **1.16** Business Day means any day other than a Saturday, a Sunday, or any other day on which banking institutions in New York, New York are required or authorized to close by law or executive order.
  - **1.17 CPLR** means the Civil Practice Law and Rules of New York.
  - **1.18** Cash means legal tender of the United States of America.
- 1.19 <u>Causes of Action</u> means any and all actions, causes of action, liabilities, obligations, rights, suits, damages, judgments, claims, and demands whatsoever, whether known or unknown, existing or hereafter arising, in law, equity, or otherwise, based in whole or in part upon any act or omission or other event occurring prior to the Commencement Date or during the course of the Chapter 11 Cases, including through the Effective Date, but not including the Avoidance Actions.
- 1.20 <u>Chapter 11 Cases</u> means the jointly administered cases under chapter 11 of the Bankruptcy Code commenced by the Debtors on September 4, 2003 in the Bankruptcy Court and styled *In re TL Administration Corporation (f/k/a Twinlab Corporation), et al.*, Ch. 11 Case No. 03-15564 (RDD) (Jointly Administered), which represents the following three cases: *In re TL Administration Corporation (f/k/a Twinlab Corporation)*, Ch. 11 Case No. 03-15564 (RDD); *In re TL Administration, Inc. (f/k/a Twin Laboratories, Inc.)*, Ch. 11 Case No. 03-15566 (RDD); and *In re TL Administration (UK) Ltd. (f/k/a Twin Laboratories (UK) Ltd.)*, Ch. 11 Case No. 03-15563 (RDD).
  - **1.21** Claim has the meaning set forth in section 101 of the Bankruptcy Code.

- **1.22** <u>Class</u> means any group of Claims or Equity Interests classified by the Plan pursuant to section 1122(a)(1) of the Bankruptcy Code.
  - **Class 4 Distribution** means \$3,550,000.
- **1.24** Collateral means any property or interest in property of the estate of any Debtor subject to a lien, charge, or other encumbrance to secure the payment or performance of a Claim, which lien, charge, or other encumbrance is not subject to avoidance under the Bankruptcy Code.
  - **1.25** Commencement Date means September 4, 2003.
- 1.26 <u>Confirmation Date</u> means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order.
- **1.27** Confirmation Hearing means the hearing held by the Bankruptcy Court pursuant to section 1128 of the Bankruptcy Code to consider confirmation of the Plan.
- **1.28** <u>Confirmation Order</u> means the order of the Bankruptcy Court confirming the Plan pursuant to section 1129 of the Bankruptcy Code.
- **1.29** Consumer Class Action Claim means any General Unsecured Claim listed in Exhibit B hereto.
- **1.30** <u>Creditor</u> means any Entity that holds a Claim against a Debtor or Debtors (whether or not such Entity has asserted a Claim against any Debtors).
- **1.31** Creditors Committee means the official committee of unsecured creditors appointed by the U.S. Trustee in the Chapter 11 Cases pursuant to section 1102 of the Bankruptcy Code.
- **1.32 Debtors or Debtors in Possession** means TL Administration Corporation (f/k/a Twinlab Corporation), TL Administration Inc. (f/k/a Twin Laboratories, Inc.) and TL Administration (UK) Ltd. (f/k/a Twin Laboratories (UK) Ltd.).
- **1.33** Declaratory Judgment Complaint means the complaint seeking declaratory judgment filed by the Blechmans on February 6, 2004 in the Bankruptcy Court, Adversary Proceeding No. 04-02323 (CB).
- **1.34 Disbursing Agent** means any Entity in its capacity as a disbursing agent under Section 7.5(a) of the Plan.
- **1.35** <u>Disclosure Statement</u> means the disclosure statement relating to the Plan, including, without limitation, all exhibits thereto, as approved by the Bankruptcy Court pursuant to section 1125 of the Bankruptcy Code.

- 1.36 <u>Disclosure Statement Hearing</u> means the hearing held by the Bankruptcy Court pursuant to section 1125 of the Bankruptcy Code to consider approval of the Disclosure Statement.
- **1.37** <u>Disclosure Statement Order</u> means the order of the Bankruptcy Court entered pursuant to section 1125 of the Bankruptcy Code approving the Disclosure Statement.
- **1.38 Disputed** means, with respect to any Claim which has not been Allowed pursuant to the Plan or a Final Order:
- (a) if no proof of claim has been filed by the applicable deadline: (i) a Claim that has been or hereafter is listed on the Schedules as disputed, contingent, or unliquidated; or (ii) a Claim that has been or hereafter is listed on the Schedules as other than disputed, contingent, or unliquidated, but as to which the Debtors or any other party in interest has interposed an objection or request for estimation which has not been withdrawn or determined by a Final Order;
- (b) if a proof of claim or request for payment of an Administrative Expense Claim has been filed by the applicable deadline: (i) a Claim for which no corresponding Claim has been or hereafter is listed on the Schedules; (ii) a Claim for which a corresponding Claim has been or hereafter is listed on the Schedules as other than disputed, contingent, or unliquidated, but the nature or amount of the Claim as asserted in the proof of claim varies from the nature and amount of such Claim as listed on the Schedules; (iii) a Claim for which a corresponding Claim has been or hereafter is listed on the Schedules as disputed, contingent, or unliquidated; or (iv) a Claim for which a timely objection or request for estimation is interposed by the Debtors or other party in interest which has not been withdrawn or determined by a Final Order.

Without limiting the foregoing and for the avoidance of any doubt, any Ephedra Personal Injury Claim that has not been Allowed prior to the Effective Date pursuant to an order of the District Court is a Disputed Claim.

- **1.39 Distribution Record Date** means the Confirmation Date.
- **1.40** <u>District Court</u> means the United States District Court for the Southern District of New York.
- **1.41** Effective Date means the date on which the Plan shall become effective, after the conditions to effectiveness of the Plan set forth in Section 9 hereof have been satisfied or waived.
- **1.42** Entity means an individual, corporation, partnership, limited liability company, association, joint stock company, joint venture, estate, trust, unincorporated organization, or government or any political subdivision thereof, or other person or entity.
- **1.43** Ephedra Claimants Committee means the official committee of ephedra claimants appointed by the U.S. Trustee in the Chapter 11 Cases pursuant to section 1102 of the Bankruptcy Code.

- 1.44 Ephedra Indemnification Claim means any Claim against any of the Debtors, that is not a Punitive Damages Claim, that is (i) held by an Entity (or any assignee or transferee of such Entity) who has been, is, or may be a defendant or otherwise liable party in a Cause of Action seeking damages for wrongful death or personal injury (whether physical, emotional, or otherwise) to the extent caused or allegedly caused, directly or indirectly, by the ingestion of ephedra or ephedra-containing products and (ii) on account of alleged liability of one or more of the Debtors for reimbursement, indemnification, subrogation, or contribution of all or any portion of (x) any damages such Entity has paid or may pay to one or more plaintiffs or other claimants in such Cause of Action and/or (y) any expenses, including attorney fees in connection with such Causes of Action.
- **1.45** Ephedra Insurance Policies means any and all policies of insurance in which the Debtors are or may be covered in respect of the 2002-2004 Ephedra PI Claims, to the full remaining extent of such coverage; provided, however, that the Ephedra Insurance Policies shall not include policies written by AISLIC.
- 1.46 Ephedra Personal Injury Claim means (i) any Claim against any of the Debtors that is not a Punitive Damage Claim, which arises out of events which occurred, in whole or in part, prior to the Commencement Date, and alleges wrongful death or personal injury (whether physical, emotional, or otherwise) to the extent caused or allegedly caused, directly or indirectly, by the ingestion of ephedra or ephedra-containing products designed, manufactured, fabricated, sold, supplied, produced, or distributed by one or more of the Debtors, including, without limitation, any Claim for compensatory damages and actual pecuniary loss, and (ii) any Ephedra Indemnification Claim. Without limitation of the foregoing, Ephedra Personal Injury Claims shall include all (i) Pre-2002 Ephedra PI Claims and (ii) 2002-2004 Ephedra PI Claims.
- **1.47** Ephedra Personal Injury Trust means the trust established pursuant to the Ephedra Personal Injury Trust Agreement.
- **1.48** Ephedra Personal Injury Trust Agreement means the Ephedra Personal Injury Trust Agreement executed by the Debtors, the Settlement Escrow Agent, and the Ephedra Personal Injury Trustee, substantially in the form of the agreement attached hereto as Exhibit C.
- 1.49 <u>Ephedra Personal Injury Trustee</u> means the person confirmed by the Bankruptcy Court to serve as trustee of the Ephedra Personal Injury Trust, pursuant to the terms of the Ephedra Personal Injury Trust Agreement, or as subsequently may be appointed pursuant to the terms of the Ephedra Personal Injury Trust Agreement.
- **1.50** Equity Interest means the interest of any holder of an equity security of Holdings represented by any issued and outstanding shares of common or preferred stock or other instrument evidencing a present ownership interest in Holdings, whether or not transferable, or any option, warrant, or right, contractual or otherwise, to acquire any such interest.
- **1.51** Estates means the three estates of the Debtors created pursuant to section 541 of the Bankruptcy Code upon the commencement of the Chapter 11 Cases.

- 1.52 Final Order means an order or judgment of the Bankruptcy Court entered by the Clerk of the Bankruptcy Court on the docket in the Chapter 11 Cases which has not been reversed, vacated, or stayed and as to which (i) the time to appeal, petition for *certiorari*, or move for a new trial, reargument, or rehearing has expired and as to which no appeal, petition for *certiorari*, or other proceeding for a new trial, reargument, or rehearing shall then be pending, or (ii) if an appeal, writ of *certiorari*, new trial, reargument, or rehearing thereof has been sought, such order or judgment of the Bankruptcy Court shall have been affirmed by the highest court to which such order was appealed, or *certiorari* shall have been denied, or a new trial, reargument, or rehearing shall have been denied or resulted in no modification of such order, and the time to take any further appeal, petition for *certiorari*, or move for a new trial, reargument, or rehearing shall have expired.
- **1.53** GNC means GNC Oldco, Inc. (f/k/a General Nutrition Companies, Inc.) and its affiliates and assigns, and their insurer, Zurich American Insurance Company.
- **1.54** General Unsecured Claim means any Claim against any of the Debtors that is not an Administrative Expense Claim, Priority Tax Claim, Priority Non-Tax Claim, Secured Claim, Ephedra Personal Injury Claim, or Punitive Damage Claim.
  - **1.55 Holdings** means TL Administration Corporation (f/k/a Twinlab Corporation).
- **1.56** <u>Ideasphere</u> means, collectively, Ideasphere, Inc. and Twinlab Corporation (f/k/a TL Acquisition Corp).
- **1.57** <u>Ideasphere Agreement</u> means that certain Asset Purchase Agreement between Ideasphere and the Debtors, dated as of September 3, 2003, as amended by (i) that certain Amendment No. 1, dated December 5, 2003, (ii) that certain Amendment No. 2, dated December 19, 2003, and (iii) that certain Amendment No. 3, dated December 19, 2003.
- **1.58 Impaired** means any Class that is impaired within the meaning of section 1124 of the Bankruptcy Code.
- **1.59** Indenture means that certain Indenture, dated as of May 7, 1996, between Holdings and US Bank Corporate Trust Services, as trustee, as such Indenture may have been amended or modified, pursuant to which \$100,000,000 in principal amount of 10-1/4% Senior Subordinated Notes due May 2006 were issued and under which \$39,915,000 principal amount of such notes remains outstanding.
- **1.60** <u>Indenture Trustee</u> means the trustee under the Indenture, and any successors or predecessors thereto.
- 1.61 <u>Initial Cash</u> means, as of the Effective Date, all Cash and Cash equivalents (not including restricted cash or the Aggregate Third Party Settlement Amount) of the Debtors, less the amount of Cash estimated and retained by the Debtors to fund adequately the administration of the Plan and the Chapter 11 Cases on and after the Effective Date and the payment of Administrative Expense Claims, Priority Tax Claims, Priority Non-Tax Claims, and, if applicable, Secured Claims.

- **1.62** Non-Ephedra Claims means any Claim that is not an Ephedra Personal Injury Claim.
- **1.63** Objection Deadline means the date set forth in the order of the Bankruptcy Court by which a holder of a Claim or a holder of an Equity Interest or other party in interest may file an objection to confirmation of the Plan.
- 1.64 Other Participating Ephedra Claims means, in connection with the lawsuits listed in Exhibit D, the claims against the Settling Third Parties seeking damages for wrongful death or personal injury (whether physical, emotional, or otherwise) to the extent caused or allegedly caused, directly or indirectly, by the ingestion of ephedra or ephedra-containing products designed, manufactured, fabricated, sold, supplied, produced, or distributed by one or more of the Debtors.
  - **Person** has the meaning set forth in section 101(41) of the Bankruptcy Code.
- **1.66** Plan means this first amended chapter 11 joint plan of liquidation, dated May 25, 2005, including the exhibits hereto, as the same may be amended or modified from time to time in accordance with the provisions of the Bankruptcy Code and the terms hereof.
- **1.67 Plan Administrator** means the person or entity charged with administering the Plan, initially selected by the Debtors and named in the Confirmation Order, and any successor thereto appointed in accordance with section 7.2 hereof.
- **1.68** Pre-2002 Ephedra PI Claim means any Ephedra Personal Injury Claim listed in Exhibit E hereto and any Ephedra Indemnification Claim in connection with a Pre-2002 Ephedra PI Claim.
- **1.69** Priority Non-Tax Claim means any Claim, other than an Administrative Expense Claim or a Priority Tax Claim, entitled to priority in payment as specified in section 507(a)(3), (4), (5), (6), (7), or (9) of the Bankruptcy Code.
- **1.70** Priority Tax Claim means any Claim of a governmental unit of the kind entitled to priority in payment as specified in sections 502(i) and 507(a)(8) of the Bankruptcy Code.
- **1.71 Pro Rata Share** means a proportionate share, so that the ratio of the consideration distributed on account of an Allowed Claim in a Class to the amount of such Allowed Claim is the same as the ratio of the consideration distributed on account of all Allowed Claims in such Class to the amount of all Allowed Claims in such Class.
- **1.72 Punitive Damages** mean any fine, penalty, forfeiture, attorneys' fees (to the extent such attorneys' fees are punitive in nature), or for multiple, exemplary, or punitive damages, to the extent that such fine, penalty, forfeiture, attorneys' fees, or damages is not for compensatory damages or actual pecuniary loss.

- **1.73** Punitive Damage Claim means, to the maximum extent permitted by law, any Claim against any of the Debtors, whether secured or unsecured, for Punitive Damages or for reimbursement, indemnification, subrogation, or contribution of Punitive Damages.
- **1.74** Released Parties means the Settling Third Parties and any current or former employee, agent, representative, officer, director, member, partner, manager, accountant, attorney, financial advisor, other professional, successor, and assign of any Settling Third Party, and any Entity claimed to be liable derivatively through any of the foregoing.
- 1.75 <u>Schedules</u> means the schedules of assets and liabilities and the statements of financial affairs filed by the Debtors under section 521 of the Bankruptcy Code, Bankruptcy Rule 1007, and the Official Bankruptcy Forms of the Bankruptcy Rules as such schedules and statements have been or may be supplemented or amended through the Confirmation Date.
- 1.76 <u>Secured Claim</u> means a Claim (i) secured by Collateral, to the extent of the value of such Collateral (a) as set forth in the Plan, (b) as agreed to by the holder of such Claim and the Debtors, or (c) as determined by a Final Order in accordance with section 506(a) of the Bankruptcy Code, or (ii) secured by the amount of any rights of setoff of the holder thereof under section 553 of the Bankruptcy Code.
- **1.77** Settlement Escrow Agent means the escrow agent appointed pursuant to the 2002-2004 Ephedra PI Settlement Agreement.
- **1.78** Settling Third Party means each of AISLIC and the Entities listed on Exhibit F hereto, each of which is a party to the 2002-2004 Ephedra PI Settlement Agreement pursuant to which such Entity agreed to pay a Third Party Settlement Amount.
- 1.79 <u>Tax Code</u> means title 26 of the United States Code, as amended from time to time.
- **1.80** Third Party Settlement Amount means the amount pursuant to the 2002-2004 Ephedra PI Settlement Agreement that a Settling Third Party agrees to pay to the Settlement Escrow Agent for distribution to the Ephedra Personal Injury Trust in accordance with the Plan.
- **1.81** <u>Treasury Regulations</u> means regulations (including temporary and proposed) promulgated under the Tax Code by the United States Treasury Department, as amended from time to time.
- **1.82** <u>U.S. Trustee</u> means the Office of the United States Trustee for the Southern District of New York.
- **1.83 Voting Procedures Order** means an order of the Bankruptcy Court approving procedures relating to the solicitation and tabulation of votes with respect to the Plan, which order may be included and be a part of the Disclosure Statement Order.

## INTERPRETATION; APPLICATION OF DEFINITIONS AND RULES OF CONSTRUCTION

The words "herein," "hereof," "hereto," "hereunder," and other words of similar import refer to the Plan as a whole and not to any particular section, subsection, or clause contained therein. Wherever from the context it appears appropriate, each term stated in either the singular or the plural shall include the singular and the plural and pronouns stated in the masculine, feminine, or neuter gender shall include the masculine, the feminine, and the neuter. A term used herein that is not defined herein shall have the meaning assigned to that term in the Bankruptcy Code. The rules of construction contained in section 102 of the Bankruptcy Code shall apply to the Plan. The headings in the Plan are for convenience of reference only and shall not limit or otherwise affect the provisions hereof.

#### **SECTION 2**

#### ADMINISTRATIVE EXPENSE CLAIMS AND PRIORITY TAX CLAIMS

- 2.1 Administrative Expense Claims. Except to the extent that a holder of an Allowed Administrative Expense Claim agrees to a different treatment, on the later of the Effective Date and the date such Administrative Expense Claim becomes an Allowed Administrative Expense Claim, or as soon thereafter as is reasonably practicable, the Debtors shall pay to each holder of an Allowed Administrative Expense Claim, in full satisfaction of such Claim, an amount in Cash equal to the Allowed amount of such Claim.
- **2.2** Priority Tax Claims. Except to the extent that a holder of an Allowed Priority Tax Claim agrees to a different treatment, on the later of the Effective Date and the date such Priority Tax Claim becomes an Allowed Priority Tax Claim, or as soon thereafter as is reasonably practicable, the Debtors shall pay to each holder of an Allowed Priority Tax Claim, in full satisfaction of such Claim, an amount in Cash equal to the Allowed amount of such Claim.

#### **SECTION 3**

### **CLASSIFICATION OF CLAIMS AND EQUITY INTERESTS**

3.1 <u>Classification of Claims and Equity Interests</u>. Claims, other than Administrative Expense Claims and Priority Tax Claims, and Equity Interests are classified for all purposes, including voting (unless otherwise specified), confirmation, and distribution pursuant to the Plan, as follows:

Class	Designation	<u>Status</u>	Entitled to Vote
Class 1	Secured Claims	Unimpaired	No
Class 2	Priority Non-Tax Claims	Unimpaired	No
Class 3	Pre-2002 Ephedra PI Claims	Unimpaired	No
Class 4	2002-2004 Ephedra PI Claims	Impaired	Yes
Class 5	General Unsecured Claims	Impaired	Yes
Class 6	Punitive Damage Claims	Impaired	No
Class 7	<b>Equity Interests</b>	Impaired	No

**3.2** <u>Bifurcation of Claims Asserting Punitive Damages.</u> Any Claim, including without limitation Pre-2002 Ephedra PI Claims, 2002-2004 Ephedra PI Claims, and Ephedra Indemnification Claims which asserts, alleges, or includes a claim for Punitive Damages or seeks reimbursement, indemnification, subrogation, or contribution of Punitive Damages shall be bifurcated into two Claims, a Claim for the amount asserted or alleged exclusive of Punitive Damages and a Punitive Damage Claim for the amount asserted or alleged as Punitive Damages (or for reimbursement, indemnification, subrogation, or contribution of Punitive Damages), whether or not asserted or alleged in a fixed or unfixed amount.

#### **SECTION 4**

# TREATMENT OF CLAIMS AND EQUITY INTERESTS

## 4.1 Class 1 – Secured Claims.

Claim agrees to less favorable treatment or that such Allowed Secured Claim has previously been satisfied by the Debtors, on the later of the Effective Date or the date such Secured Claim becomes an Allowed Secured Claim, or as soon thereafter as is reasonably practicable, each holder of an Allowed Secured Claim shall receive, at the option of the Debtors, and in full satisfaction of such Claim, either (i) Cash in an amount equal to one hundred percent (100%) of the unpaid amount of such Allowed Secured Claim, (ii) the proceeds of the sale or disposition of the Collateral securing such Allowed Secured Claim to the extent of the value of the holder's secured interest in the Allowed Secured Claim, net of the costs of disposition of such Collateral, (iii) the Collateral securing such Allowed Secured Claim, (iv) treatment that leaves unaltered the legal, equitable, and contractual rights to which the holder of such Allowed Secured Claim is entitled, or (v) such other distribution and/or treatment as necessary to satisfy the requirements of the Bankruptcy Code for classifying a Class of secured Claims as unimpaired. In the event the Debtors treat a Claim under clause (i) or (ii) of this Section, the liens securing such Secured Claim shall be deemed released.

(b) <u>Impairment and Voting</u>. Class 1 is unimpaired by the Plan. Each holder of a Secured Claim in Class 1 is not entitled to vote to accept or reject the Plan and shall be conclusively deemed to have accepted the Plan.

### 4.2 Class 2 - Priority Non-Tax Claims.

- (a) <u>Distributions</u>. Except to the extent that a holder of an Allowed Priority Non-Tax Claim agrees to a different treatment of such Claim, on the later of the Effective Date and the date such Priority Non-Tax Claim becomes an Allowed Priority Non-Tax Claim, or as soon thereafter as is reasonably practicable, each holder of an Allowed Priority Non-Tax Claim shall receive, in full satisfaction of such Claim, an amount in Cash equal to the Allowed amount of such Priority Non-Tax Claim.
- (b) <u>Impairment and Voting</u>. Class 2 is unimpaired by the Plan. Each holder of a Priority Non-Tax Claim in Class 2 is not entitled to vote to accept or reject the Plan and shall be conclusively deemed to have accepted the Plan.

# 4.3 <u>Class 3 – Pre-2002 Ephedra PI Claims</u>.

- (a) <u>Distributions</u>. Subject to subsection (b) below, each Pre-2002 Ephedra PI Claim shall be satisfied in full in the ordinary course of business from the proceeds of the Debtors' applicable insurance policy or policies, as the case may be, at such time as such Pre-2002 Ephedra PI Claim becomes an Allowed Claim.
- Liquidation of Pre-2002 Ephedra PI Claims. All Pre-2002 Ephedra PI Claims are Disputed Claims and shall be determined, resolved, or adjudicated, as the case may be, in a manner as if the Chapter 11 Cases had not been commenced and shall survive the Effective Date as if the Chapter 11 Cases had not been commenced. Upon the determination, resolution or adjudication of any such Claim as provided herein, such Claim shall be deemed to be an Allowed Pre-2002 Ephedra PI Claim, in the amount or in the manner determined by a Final Order or by a binding award, agreement or settlement; provided, however, that in addition to the Debtors' preservation of all rights and defenses respecting any Pre-2002 Ephedra PI Claim that exist under applicable non-bankruptcy law, (i) any rejection, avoidance, recovery, or other power or defense available to the Debtors under sections 365, 510 (except subordination), 542, 543, 544, 545, 547, 548, 549, 550, 533 or 724 of the Bankruptcy Code is preserved, and (ii) the Debtors may object under section 502 of the Bankruptcy Code to any Pre-2002 Ephedra PI Claim on the ground that (A) such Pre-2002 Ephedra PI Claim was not timely asserted in the Chapter 11 Cases, (B) such Pre-2002 Ephedra PI Claim is subject to any power or defense reserved in clause (i) of this sentence and/or is disallowable under section 502(d) of the Bankruptcy Code, or (C) such Pre-2002 Ephedra PI Claim is disallowable under section 502(e) of the Bankruptcy Code, to the extent that such section is relied on to ensure that there is no duplication in the Claim of an allegedly subrogated claim, on the one hand, and the underlying claimant whose Claim allegedly gave rise to the subrogated claim, on the other. Subject to the foregoing, all Pre-2002 Ephedra PI Claims shall be determined and liquidated under applicable non-bankruptcy law in the administrative or judicial tribunal in which they are pending as of the Effective Date, or if no such action is pending on the Effective Date, in any administrative or

judicial tribunal of appropriate jurisdiction. To effectuate the foregoing, the entry of the Confirmation Order shall, effective as of the Effective Date, constitute a modification of any stay or injunction under the Bankruptcy Code that would otherwise preclude the determination, resolution or adjudication of any Pre-2002 Ephedra PI Claim. Nothing contained in this Section will constitute or be deemed to constitute a waiver or release of any (i) claim, right or Cause of Action that the Debtors may have against any Entity in connection with or arising out of any Pre-2002 Ephedra PI Claim, including, but not limited to, any rights under section 157(b) of title 28, United States Code, or (ii) defense in any action or proceeding in any administrative or judicial tribunal, including, but not limited to, with respect to the jurisdiction of such administrative or judicial tribunal, except a defense to a Claim that was timely asserted in the Chapter 11 Cases and that constitutes a Pre-2002 Ephedra PI Claim where such defense is based on the discharge of section 1141(d) of the Bankruptcy Code or any release, discharge, or injunction contained in this Plan or the Confirmation Order. Further, nothing contained in this Section or in the Plan shall affect the rights of any Settling Third Party to coverage under the Debtors' applicable insurance policy or policies with respect to the Pre-2002 Ephedra PI Claims.

(c) <u>Impairment and Voting</u>. Class 3 is unimpaired by the Plan. Each holder of a Pre-2002 Ephedra PI Claim in Class 3 is not entitled to vote to accept or reject the Plan and shall be conclusively deemed to have accepted the Plan.

# 4.4 <u>Class 4 – 2002-2004 Ephedra PI Claims</u>.

(a) <u>Distributions</u>. All 2002-2004 Ephedra PI Claims shall be determined and paid pursuant to the terms, provisions, and procedures of the Ephedra Personal Injury Trust and the Ephedra Personal Injury Trust Agreement. The Ephedra Personal Injury Trust shall be funded in accordance with the provisions of section 5.3 of the Plan.

All 2002-2004 Ephedra PI Claims are Disputed Claims. The Ephedra Personal Injury Trust shall assume full responsibility for determining the Allowed amount of all 2002-2004 Ephedra PI Claims pursuant to the Ephedra Personal Injury Trust Agreement (incorporated herein and made a part hereof), and for making payments on account of 2002-2004 Ephedra PI Claims that become Allowed 2002-2004 Ephedra PI Claims under the conditions set forth in the Ephedra Personal Injury Trust Agreement. Distributions to holders of 2002-2004 Ephedra PI Claims are limited to such payments by the Ephedra Personal Injury Trust.

(b) <u>Impairment and Voting</u>. Class 4 is impaired by the Plan. Each holder of a 2002-2004 Ephedra PI Claim is entitled to vote to accept or reject the Plan to the extent and in the manner provided in the Voting Procedures Order or in any other order or orders of the Bankruptcy Court temporarily allowing one or more 2002-2004 Ephedra PI Claims for voting purposes pursuant to Bankruptcy Rule 3018(a).

### 4.5 Class 5 – General Unsecured Claims.

(a) <u>Distributions</u>. On the Effective Date, or as soon thereafter as is reasonably practicable, each holder of an Allowed General Unsecured Claim shall receive its Pro Rata Share of Available Cash until such Claim has been paid or otherwise satisfied in full.

(b) <u>Impairment and Voting</u>. Class 5 is impaired by the Plan. To the extent and in the manner provided in the Voting Procedures Order, each holder of a General Unsecured Claim is entitled to vote to accept or reject the Plan.

### 4.6 Class 6 – Punitive Damage Claims.

- (a) <u>Distributions</u>. Each holder of a Punitive Damage Claim shall (i) be subordinated in full to the Claims in Class 1, Class 2, Class 3, Class 4, and Class 5 and (ii) receive no distribution under the Plan.
- (b) <u>Impairment and Voting</u>. Class 6 is impaired by the Plan. Holders of Punitive Damage Claims are not entitled to vote on the Plan and shall be conclusively deemed to have rejected the Plan.

## 4.7 Class 7 – Equity Interests.

- (a) <u>Distributions</u>. On the Effective Date, or as soon thereafter as is reasonably practicable, all Equity Interests shall be canceled, at which time one new share of Holdings common stock shall be issued to the Plan Administrator who shall hold such share as custodian for the benefit of holders of Equity Interests consistent with their former economic entitlement. Such holders' beneficial interests in such share of new common stock shall be noncertified and nontransferable.
- (b) <u>Impairment and Voting</u>. Class 7 is impaired by the Plan. Holders of Class 7 Equity Interests are not entitled to vote and shall be conclusively deemed to have rejected the Plan.

#### SECTION 5

#### THE EPHEDRA PERSONAL INJURY TRUST

- Personal Injury Trust will assume full responsibility for determining the Allowed amount of all 2002-2004 Ephedra PI Claims and Other Participating Ephedra Claims pursuant to the Ephedra Personal Injury Trust Agreement (incorporated herein and made a part hereof), for making payments on account of 2002-2004 Ephedra PI Claims and Other Participating Ephedra Claims that become Allowed Claims under the conditions set forth in the Ephedra Personal Injury Trust Agreement, for fulfilling its other obligations under the Ephedra Personal Injury Trust Agreement, and for paying its own costs and expenses, all as set forth more fully hereon and in the Ephedra Personal Injury Trust Agreement.
- 5.2 Resolution of Minors' Claims by the Ephedra Personal Injury Trust. In connection with any 2002-2004 Ephedra PI Claim or Other Participating Ephedra Claim in which the plaintiff is a minor, the proposed distribution to the plaintiff shall be submitted to the District Court for approval in accordance with Rule 83.2 of the Local Civil Rules of the Southern and Eastern Districts of New York and Rule 1208 of the CPLR. The parent, guardian, guardian ad litem, adult spouse, next friend, or other representative of the minor shall move for approval

of the proposed distribution and shall file, with such motion (a) an affidavit of the representative that complies in all respects with the requirements set forth in CPLR 1208(a); (b) an affidavit of the minor's or the representative's attorney, which complies in all respects with the requirements set forth in CPLR 1208(b); and (c) a copy of relevant medical or hospital records, as required by CPLR 1208(c). The District Court shall hold a hearing on the motion, and the movant, the minor plaintiff, and the attorney shall attend such hearing. The District Court shall determine whether the proposed distribution is in the minor plaintiff's best interests and, if the proposed distribution is approved, that determination shall be embodied in an order which shall have the effect of a judgment. Upon approval of the proposed distribution to a minor plaintiff, the District Court shall authorize payment to the minor's counsel of fees and disbursements to be paid from the distribution and further shall order that the remainder of the distribution be distributed in a manner that will best protect the interest of the minor.

Effective Date, or as soon thereafter as is reasonably practicable, the Debtors will, in full satisfaction and discharge of all Claims in Class 4 (i.e. 2002-2004 Ephedra PI Claims), cause the following to occur: (a) the execution and delivery by the Debtors and the Ephedra Personal Injury Trustee of the Ephedra Personal Injury Trust Agreement, thereby establishing the Ephedra Personal Injury Trust; (b) the payment by the Debtors to the Ephedra Personal Injury Trust of the amount of the Class 4 Distribution; (c) the payment by the Settlement Escrow Agent to the Ephedra Personal Injury Trust of the Aggregate Third Party Settlement Amount; and (d) the transfers with respect to any Ephedra Insurance Policies as provided directly below.

On the Effective Date, or as soon thereafter as is reasonably practicable, the Debtors will transfer, assign, and deliver to the Ephedra Personal Injury Trust, for the benefit of the Ephedra Personal Injury Trust as an insured party thereunder, any of the Ephedra Insurance Policies which are assignable without cancellation or reduction of coverage and which the Debtors elect to so transfer, assign, and deliver; and in respect of such Ephedra Insurance Policies which the Debtors do not elect to so transfer, assign, and deliver, the Debtors will take all such actions as are necessary or desirable to enable the Ephedra Personal Injury Trust, at its expense, to obtain and enjoy the benefits of coverage thereunder, including, without limitation, pursuit in the name of the Debtors (subject to control by, and at the expense of, the Ephedra Personal Injury Trust) of all claims under the Ephedra Insurance Policies and assignment by the Debtors to the Ephedra Personal Injury Trust of all recoveries thereunder.

- 5.4 <u>Tax Treatment of Ephedra Personal Injury Trust.</u> Subject to a determination by the Internal Revenue Service or a court of competent jurisdiction to the contrary, the Ephedra Personal Injury Trust shall be treated as a "qualified settlement fund" within the meaning of Treasury Regulation section 1.468B-1, and all parties (including the Debtors, the Ephedra Personal Injury Trustee and the holders of 2002-2004 Ephedra PI Claims) shall report consistent therewith.
- 5.5 <u>Appointment of Ephedra Personal Injury Trustee</u>. Prior to the Effective Date, the Ephedra Claimants Committee shall appoint the individual to serve as the Ephedra Personal Injury Trustee for the Ephedra Personal Injury Trust; provided that no individual shall be appointed as the Ephedra Personal Injury Trustee if such individual would be treated as, or

would cause the Ephedra Personal Injury Trust to be treated as, a "related person" (within the meaning of Treasury Regulation section 1.468B-1(d)(2)) to any of the Debtors.

#### **SECTION 6**

### PROVISIONS GOVERNING DISTRIBUTIONS

- Classes of Claims or Equity Interests as maintained by the Debtors, or their claims agent, shall be deemed closed, and there shall be no further changes in the record holders of any of the Claims or Equity Interests; (ii) the Debtors and the Ephedra Personal Injury Trustee, as the case may be, shall have no obligation to recognize any transfer of Claims or Equity Interests entered on the docket of the Bankruptcy Court on or after the Distribution Record Date; and (iii) the Debtors and the Ephedra Personal Injury Trustee, as the case may be, shall be entitled to recognize and deal for all purposes hereunder only with those record holders stated on the transfer ledgers as of the close of business on the Distribution Record Date, to the extent applicable. Distributions to holders of General Unsecured Claims under the Indenture will be made pursuant to the surrender of securities as described in section 6.3 of the Plan.
- 6.2 **Delivery of Distributions and Undeliverable Distributions.** Subject to Bankruptcy Rule 9010, all distributions to any holder of an Allowed Claim shall be made at the address of such holder as set forth on the Schedules filed with the Bankruptcy Court or on the books and records of the Debtors or their agents or in a letter of transmittal unless the Debtors and/or the Ephedra Personal Injury Trustee, as the case may be, have been notified in writing of a change of address, including, without limitation, by the filing of a proof of claim by such holder that contains an address for such holder different from the address reflected on such Schedules for such holder. Any distribution to be made to a holder of a General Unsecured Claim under the Indenture shall be made to the Indenture Trustee to assert its charging lien against such distributions. Any distribution to the Indenture Trustee shall be deemed a distribution to the respective holder of a General Unsecured Claim under the Indenture. In the event that any distribution to any holder is returned as undeliverable, no further distributions to such holder shall be made unless and until the Debtors or the Ephedra Personal Injury Trustee, as the case may be, are notified of such holder's then-current address, at which time all missed distributions shall be made to such holder, without interest. All demands for undeliverable distributions shall be made on or before ninety (90) days after the date such undeliverable distribution was initially made. Thereafter, in accordance with section 347(b) of the Bankruptcy Code, the amount represented by such undeliverable distribution shall irrevocably revert to the Debtors or the Ephedra Personal Injury Trust, as the case may be, and any Claim in respect of such undeliverable distribution shall be discharged and forever barred from assertion against the Ephedra Personal Injury Trust, the Debtors, and their property.
- **6.3 Surrender of Instruments.** As a condition to receiving any distribution under the Plan, each holder of a note under the Indenture must surrender such note held by it to the Indenture Trustee or its designee. Any holder of such note that fails to (i) surrender such note, or (ii) execute and deliver an affidavit of loss and/or indemnity reasonably satisfactory to

the Indenture Trustee and furnish a bond in form, substance, and amount reasonably satisfactory to the Indenture Trustee before the first anniversary of the Effective Date shall be deemed to have forfeited all rights and Claims and may not participate in any distribution under the Plan. Any distribution so forfeited shall become property of the Debtors.

- 6.4 Setoffs. The Debtors may, but shall not be required to, set off against any Claim (for purposes of determining the Allowed amount of such Claim on which distribution shall be made), any claims of any nature whatsoever that the Debtors may have against the holder of such Claim, but neither the failure to do so nor the allowance of any Claim hereunder shall constitute a waiver or release by the Debtors of any such claim the Debtors may have against the holder of such Claim.
- **6.5** Transactions on Business Days. If the Effective Date or any other date on which a transaction may occur under the Plan shall occur on a day that is not a Business Day, the transactions contemplated by the Plan to occur on such day shall instead occur on the next succeeding Business Day but shall be deemed to have been completed as of the required date.
- 6.6 <u>Allocation of Plan Distribution Between Principal and Interest.</u> All distributions in respect of any Allowed Claim shall be allocated first to the principal amount of such Allowed Claim, as determined for federal income tax purposes, and thereafter, to the remaining portion of such Claim comprising interest, if any (but solely to the extent that interest is an allowable portion of such Allowed Claim).
- 6.7 Withholding and Reporting Requirements. The Plan Administrator shall comply with all applicable withholding and reporting requirements imposed by any federal, state, local, or foreign tax authority, and all distributions under the Plan shall be subject to any such withholding and reporting requirements. Notwithstanding the above, each holder of an Allowed Claim that is to receive a distribution under the Plan shall have the sole and exclusive responsibility for the satisfaction and payment of any tax obligations imposed by any governmental unit, including income, withholding, and other tax obligations, on account of such distribution.

#### SECTION 7

### MEANS FOR IMPLEMENTATION AND EXECUTION OF THE PLAN

Effective Date, the Debtors shall be deemed consolidated for voting and distribution purposes under the Plan. Accordingly, for such purposes: (i) all guaranties of any Debtor of the payment, performance, or collection of obligations of another Debtor shall be eliminated and canceled, (ii) any obligation of any Debtor and all guaranties thereof executed by one or more of the other Debtors shall be treated as a single obligation, and such guaranties shall be deemed a single Claim against the deemed consolidated Debtors, (iii) all joint obligations of two or more Debtors and all multiple Claims against such entities on account of such joint obligations shall be treated and allowed only as a single Claim against the deemed consolidated Debtors, (iv) all Claims between or among the Debtors shall be canceled, and (v) each Claim filed in the Chapter 11 Case of any Debtor shall be deemed filed against the deemed consolidated Debtors and a single

obligation of the deemed consolidated Debtors on and after the Effective Date. Such deemed consolidation, however, shall not affect the legal and organizational structure of the Debtors and each of TL Administration Inc. and TL Administration (UK) ltd. shall be governed by their respective certificate of incorporation and by-laws.

### 7.2 The Plan Administrator.

- (a) The Confirmation Order shall name the Plan Administrator to implement, or cause to be implemented, the terms of the Plan with respect to the consolidated Debtors.
- (b) The Plan Administrator shall be designated by the Debtors in consultation with the Creditors Committee.
- (c) The salient terms of the Plan Administrator's employment, including the Plan Administrator's duties and compensation (which compensation shall be negotiated by the Plan Administrator, the Debtors, and the Creditors Committee), to the extent not set forth in the Plan, shall be set forth in the Confirmation Order. The Plan Administrator shall be entitled to reasonable compensation in an amount consistent with that of similar functionaries in similar types of bankruptcy proceedings.
- (d) The Plan Administrator shall have no liability to the Debtors, their creditors, or their equity holders. The Debtors shall indemnify and hold harmless the Plan Administrator for any losses incurred in its capacity as such, except to the extent such losses were the result of such person's gross negligence or willful misconduct.
- (e) In the event the Plan Administrator dies, is terminated, or resigns for any reason, the U.S. Trustee shall designate a successor. The Plan Administrator shall be required to disclose its connections, if any, with the Debtors, their creditors, any other party in interest, and the U.S. Trustee.

### 7.3 Directors/Officers/Assets of the Debtors on the Effective Date.

- (a) On the Effective Date, the authority, power and incumbency of the persons then acting as directors and officers of the Debtors shall be terminated and such directors and officers shall be deemed to have resigned or to have been removed without cause.
- (b) On the Effective Date, the Plan Administrator shall be appointed as and become, and shall succeed to such powers as would have been applicable to Holding's officers and directors. In addition, the Plan Administrator shall be appointed as the initial director and officer of TL Administration Inc. and TL Administration (UK) Ltd., to serve in accordance with the respective certificates of incorporation and by-laws of such companies.
- (c) Notwithstanding anything to the contrary herein, all assets of the Debtors shall remain assets of the respective Debtor until they are transferred or distributed, or caused to be transferred or distributed, by the Plan Administrator pursuant to the terms of this Plan.

- After the Effective Date, the Plan Administrator may decide, in its sole discretion, to maintain, or cause to be maintained, any or all of the Debtors as corporations in good standing until such time as all aspects of the Plan (other than those aspects that are the responsibility of the Ephedra Personal Injury Trust) have been completed (provided, however, that the Debtors shall conduct no business except as necessary or appropriate to implement the Plan) or, at such time as the Plan Administrator considers appropriate and consistent with the implementation of the Plan, may dissolve, or cause to be dissolved, the Debtors and complete the winding up of the Debtors in accordance with applicable law. In connection with the liquidation of the Debtors, the Plan Administrator may transfer, or caused to be transferred, all assets of the Debtors to a trust or other entity created to make distributions in accordance with the Plan to holders of Allowed Claims and the Ephedra Personal Injury Trust. The Plan Administrator may act as trustee of such a trust if and as permitted by law or in the capacity of management of any other entity created to make such distributions. As soon as practicable after all aspects of the Plan (other than those aspects that are the responsibility of the Ephedra Personal Injury Trust) have been completed, the Debtors shall be dissolved and wound up in accordance with applicable law.
- (e) As of the Effective Date, the certificate of incorporation and by-laws of each of the Debtors shall be amended to the extent necessary to carry out the provisions of the Plan. The amended certificates and by-laws, if any, of each of the Debtors shall be included in the Plan Supplement.
- 7.4 <u>Duties and Powers of the Plan Administrator</u>. The Plan Administrator, together with its representatives and professionals, shall administer, or cause to be administered, the Plan. The duties and powers of the Plan Administrator shall include all powers necessary to implement the Plan and administrate and liquidate, or cause to be administered and liquidated, the assets of the Debtors, including, without limitation, the duties and powers listed herein.
- (a) <u>Authority</u>. The Plan Administrator may exercise all power and authority that may be or could have been exercised, commence all proceedings that may be or could have been commenced and take all actions that may be or could have been taken by an officer, director, or shareholder of Holdings with like effect as if authorized, exercised, and taken by unanimous action of such officers and directors, without shareholder approval, including, without limitation, amendment of the certificate of incorporation and by-laws of Holdings and the dissolution of Holdings.
- (b) <u>Claims</u>. The Plan Administrator may object to, seek to subordinate, compromise, or settle any or all Claims against the Debtors except such Claims that are the responsibility of the Ephedra Personal Injury Trust; provided, however, that the Creditors Committee shall have the exclusive right to continue to litigate, compromise, or settle any litigation commenced by the Creditors Committee on behalf of the Debtors prior to the Effective Date.
- (c) <u>Liquidation of Assets</u>. The Plan Administrator shall, or shall cause the appropriate Debtor to, in an expeditious but orderly manner, liquidate and convert to cash the assets of the consolidated Debtors, make timely distributions, and not unduly prolong the

duration of the Debtors. In so doing, the Plan Administrator shall exercise its reasonable business judgment in liquidating the assets of the consolidated Debtors to maximize recoveries. The liquidation of such assets may be accomplished through the sale of the assets (in whole or in combination, and including the sale of any Causes of Action, other than Causes of Action released under the Plan), through the prosecution, compromise and settlement, abandonment, or dismissal of any or all claims or Causes of Action (other than Causes of Action released under the Plan), or otherwise.

- (d) <u>Abandoning Assets</u>. The Plan Administrator may, or may cause the appropriate Debtor to, abandon in any commercially reasonably manner, including abandonment or donation to a charitable organization of its choice, any assets, if it concludes that they are of no benefit to the Estates.
- (e) <u>Causes of Action</u>. The Plan Administrator may pursue, or cause to be pursued, Causes of Action (other than Causes of Action released under the Plan) of the consolidated Debtors. The Plan Administrator shall have discretion to elect whether or not to pursue any and all such Causes of Action and whether and when to compromise, settle, abandon, dismiss, or otherwise dispose of any such Causes of Action, as the Plan Administrator may determine is in the best interests of the holders of Claims against the Debtors, and the Plan Administrator shall have no liability to any of the Debtors, their Estates, their creditors, the Creditors Committee or its members, the Ephedra Claimants Committee or its members, or any other party for the outcome of its decisions in this regard.
- (f) <u>Retention of Professionals</u>. The Plan Administrator may retain professionals to assist it in performing its duties hereunder.
- (g) <u>Books and Records</u>. The Plan Administrator shall maintain the Debtors' books and records, maintain accounts, make distributions, and take other actions consistent with the Plan and the implementation hereof.
- (h) <u>Agreements</u>. The Plan Administrator may, or may cause the appropriate Debtor to, enter into any agreement or execute any document required by or consistent with the Plan and performall of obligations of the respective Debtors thereunder.
- (i) <u>Investment Power.</u> The right and power of the Plan Administrator to invest, or cause to be invested, any of the Debtors' Cash, including cash proceeds from the liquidation of any assets and the realization or disposition of any Causes of Action, and any income earned by the Debtors shall be limited to the right and power to invest such Cash in United States Treasury Bills, interest-bearing certificates of deposit, tax exempt securities having at the time of acquisition an investment grade credit rating as assigned by a nationally recognized credit rating service or investments permitted by section 345 of the Bankruptcy Code or otherwise authorized by the Bankruptcy Court, using prudent efforts to enhance the rates of interest earned on such Cash without inordinate credit risk or interest rate risk; <u>provided</u>, <u>however</u>, that the Plan Administrator may expend the Cash of the consolidated Debtors to effectuate the provisions of the Plan.

- (j) <u>Tax Obligations</u>. The Plan Administrator shall have the powers of administration regarding all of the tax obligations of the Debtors, including filing of returns. The Plan Administrator shall (i) endeavor to complete and file, within ninety (90) days after the dissolution of the Debtors, the Debtors final federal, state and local tax returns, (ii) request, if necessary, an expedited determination of any unpaid tax liability of the Debtors or their Estates under Bankruptcy Code section 505(b) for all taxable periods of the consolidated Debtors ending after the Commencement Date through the dissolution of the Debtors as determined under applicable tax laws and (iii) represent the interest and account of the Debtors or their Estates before any taxing authority in all matters including, without limitation, any action, suit, proceeding or audit.
- (k) <u>Reporting Duties</u>. The Plan Administrator shall be responsible for filing informational returns on behalf of the Debtors and paying any tax liability of the Debtors. The Plan Administrator shall file (or cause to be filed) any other statements, returns or disclosures relating to the Debtors that are required by any governmental unit or applicable law.
- (l) <u>Reasonable Fees and Expenses</u>. The Plan Administrator may incur any reasonable and necessary expenses in connection with the performance of its duties under the Plan.
- (m) <u>Other Actions</u>. The Plan Administrator may take all other actions not inconsistent with the provisions of the Plan which the Plan Administrator deems reasonably necessary or desirable with respect to administering the Plan.

# 7.5 Method of Distributions Under the Plan.

- Personal Injury Trust and holders of Allowed Non-Ephedra Claims against the Debtors shall be made in accordance with the terms of the Plan and to the extent provided by the Plan by the Plan Administrator as Disbursing Agent or such other Entity designated by the Plan Administrator as Disbursing Agent. A Disbursing Agent shall not be required to provide any bond, surety, or other security for the performance of its duties, unless otherwise ordered by the Bankruptcy Court; and, in the event that a Disbursing Agent is so otherwise ordered, all costs and expenses of procuring any such bond, surety, or other security shall be borne by the Debtors.
- (b) <u>Effective Date Payments and Transfers</u>. On the Effective Date, or, as soon thereafter as reasonably practicable, the Plan Administrator shall (i) remit, to each holder of an Allowed Non-Ephedra Claim (as of the Distribution Record Date) against the Debtors, the distribution provided for such Claim under the Plan and (ii) pay and transfer to the Ephedra Personal Injury Trust (or cause the Settlement Escrow Agent to pay and transfer to the Ephedra Personal Injury Trust) the Cash and assets to be paid and transferred to the Ephedra Personal Injury Trust pursuant to section 5.3 of the Plan.
- (c) <u>Distribution of Cash</u>. At the option of the Plan Administrator or the Ephedra Personal Injury Trustee, as the case may be, any Cash payment to be made hereunder may be by check or wire transfer or as otherwise required or provided in applicable agreements.

- **Retention of Cash.** In making the distributions under the Plan, the (d) Debtors shall, at all times, retain sufficient Cash and/or beneficial assets as reasonably necessary to (i) satisfy Administrative Expense Claims, Priority Tax Claims, Priority Non-Tax Claims, and, if applicable, Secured Claims, to the extent such claims are not paid on or prior to the Effective Date, (ii) meet the reasonably necessary administrative expenses of the Debtors after the Effective Date, including contingent liabilities and the cost of continuation of litigation commenced by the Creditors Committee on behalf of the Debtors prior to the Effective Date, (iii) pay reasonable administrative expenses of the Estate that have not been paid (including the Debtors' professional fees and expenses and the fees of the U.S. Trustee) or have not been Allowed as of the Effective Date but which are subsequently Allowed, (iv) satisfy all Disputed Claims (including, without limitation and as applicable, Consumer Class Action Claims, but not including Disputed Ephedra Personal Injury Claims) estimated by the Bankruptcy Court pursuant to section 502(c) of the Bankruptcy Code in the amount so estimated, (v) satisfy all Disputed Claims (but not including Disputed Ephedra Personal Injury Claims or Disputed Claims estimated by the Bankruptcy Court pursuant to section 502(c) of the Bankruptcy Code) in the event that all such Claims were to become Allowed Claims, (vi) satisfy other liabilities incurred by the Debtors in accordance with the Plan, and (vii) otherwise perform the functions and take the actions provided for or permitted herein or in any other agreement executed by the Plan Administrator pursuant to the Plan. The amounts retained pursuant to this paragraph shall not be segregated from the other assets of the Debtors.
- (e) <u>Final Distributions and Transfers</u>. As soon as reasonably practicable after the resolution of all Disputed Non-Ephedra Claims and after all assets of the consolidated Debtors have been liquidated and the resolution of all Causes of Action, the Plan Administrator shall (x) to the extent not paid, pay all Allowed Administrative Expense Claims, Allowed Priority Tax Claims, Allowed Priority Non-Tax Claims, and wind down costs and expenses including the fees of the U.S. Trustee; (y) to the extent the Debtors have remaining Cash after the payments made in accordance with (x) above, distribute to each holder of an Allowed General Unsecured Claim its Pro Rata share of such remaining Cash; but in no event shall the amount of such distribution, when added to the amount of prior distributions received by such holder, exceed the amount of such Allowed Claim; and (z) complete the wind up of the affairs of the Debtors (or the successor to the Debtors).

## 7.6 Procedures for Treating Disputed Claims.

- (a) <u>Objections</u>. Any objections to Non-Ephedra Claims shall be served and filed on or before the later of (a) 120 days after the Effective Date and (b) such later date as may be fixed by the Bankruptcy Court.
- (b) <u>No Distributions Pending Allowance</u>. Notwithstanding any other provision hereof, if any portion of a Non-Ephedra Claim is a Disputed Claim, no payment or distribution provided hereunder shall be made on account of such Claim unless and until such Disputed Claim becomes an Allowed Claim.
- (c) <u>Estimation of Claims</u>. The Plan Administrator may at any time request that the Bankruptcy Court estimate any contingent, unliquidated, or Disputed Non-Ephedra

Claim including, without limitation, any Consumer Class Action Claim, pursuant to section 502(c) of the Bankruptcy Code or otherwise regardless of whether the Debtors previously objected to such Claim or whether the Bankruptcy Court has ruled on any such objection, and the Bankruptcy Court will retain exclusive jurisdiction to estimate any Claim at any time during litigation concerning any objection to any Claim, including without limitation, during the pendency of any appeal relating to any such objection. In the event that the Bankruptcy Court estimates any contingent, unliquidated, or Disputed Non-Ephedra Claim, the amount so estimated shall constitute either the Allowed amount of such Claim or a maximum limitation on such Claim, as determined by the Bankruptcy Court. If the estimated amount constitutes a maximum limitation on the amount of such Claim, the Plan Administrator may pursue supplementary proceedings to object to the allowance of such Claim. All of the aforementioned objection, estimation, and resolution procedures are intended to be cumulative and not exclusive of one another. Non-Ephedra Claims may be estimated and subsequently disallowed, reduced, compromised, settled, withdrawn, or resolved by any mechanism approved by the Bankruptcy Court.

## (d) Resolution of Disputed Claims.

- (i) Within sixty (60) days of a Disputed Non-Ephedra Claim becoming an Allowed Claim, the Plan Administrator shall remit to the holder of such Allowed Claim Cash equal to the amount such holder would have received as of that date under the Plan if the Allowed portion of the Disputed Claim had been an Allowed Claim as of the Effective Date.
- (ii) To the extent that a Disputed Non-Ephedra Claim is not Allowed or becomes an Allowed Claim in an amount less than the amount of the Disputed Claim set forth in the proof of claim, or as previously estimated by the Bankruptcy Court, the excess of the amount of Cash that would have been distributed to the holder of the Disputed Claim if the Claim had been Allowed in full over the amount of Cash actually distributed on account of such Disputed Claim shall be Available Cash.
- (iii) Holders of Disputed Non-Ephedra Claims shall not be entitled to interest if such Disputed Claim becomes an Allowed Claim except to the extent such holder is entitled to interest under the Plan as a holder of an Allowed Claim.
- 7.7 Closing of the Chapter 11 Cases. When all Disputed Non-Ephedra Claims filed against the Debtors have become Allowed Claims or have been disallowed by Final Order and all remaining assets of the consolidated Debtors have been liquidated and converted into Cash or abandoned and such Cash has been distributed in accordance with the Plan, the Plan Administrator shall seek authority from the Bankruptcy Court to close the Chapter 11 Cases in accordance with the Bankruptcy Code and the Bankruptcy Rules.
- **7.8** Cancellation of Existing Securities and Agreements. Except for purposes of evidencing a right to distributions under the Plan or otherwise provided hereunder, on the Effective Date all the agreements and other documents evidencing the Claims or rights of any holder of a Claim against the Debtors, including the Indenture and any notes evidencing such

Claims shall be canceled; <u>provided</u>, <u>however</u>, that the Indenture shall continue in effect solely for the purposes of (i) allowing the Indenture Trustee to make any distributions on account of Allowed General Unsecured Claims of noteholders under the Indenture pursuant to the Plan and to perform such other necessary administrative functions with respect thereto, and (ii) permitting the Indenture Trustee to maintain any rights or liens they may have for fees, costs, and expenses under the Indenture.

- 7.9 <u>Dissolution</u> Within thirty (30) days after completion of the acts required by the Plan, or as soon thereafter as is practicable, each Debtor shall be deemed dissolved for all purposes without the necessity for any other or further actions to be taken by or on behalf of each Debtor; <u>provided</u>, <u>however</u>, that each Debtor shall file with the office of the Secretary of State or other appropriate office for the state of its organization a certificate of cancellation or dissolution.
- 8. The Plan Administrator shall have the responsibility of storing and maintaining the Debtors' books and records until one year after the date the Debtors are dissolved in accordance with Section 7.9 hereof, after which time such books and records may be abandoned or destroyed without further Bankruptcy Court order. For purposes of this Section, books and records include computer generated or computer maintained books and records and computer data, as well as electronically generated or maintained books and records or data, along with books and records of the Debtors maintained by or in possession of third parties and all of the claims and rights of the Debtors in and to their books and records, wherever located.
- 7.11 **Corporate Action.** Upon the Effective Date, the Debtors shall perform each of the actions and effect each of the transfers required by the terms of the Plan, in the time period allocated therefor, and all matters provided for under the Plan that would otherwise require approval of the stockholders, partners, members, directors, or comparable governing bodies of the Debtors shall be deemed to have occurred and shall be in effect from and after the Effective Date pursuant to the applicable general corporation law (or other applicable governing law) of the states in which the Debtors are incorporated or organized, without any requirement of further action by the stockholders, members, or directors (or other governing body) of the Debtors. Each of the Debtors shall be authorized and directed, following the completion of all disbursements, other transfers, and other actions required of the Debtors by the Plan, to file its certificate of cancellation or dissolution as contemplated by Section 7.9 hereof. The filing of such certificates of cancellation, dissolution, or merger shall be authorized and approved in all respects without further action under applicable law, regulation, order, or rule, including, without express or implied limitation, any action by the stockholders, members, or directors (or other governing body) of the Debtors.
- **7.12** Effectuating Documents and Further Transactions. Each of the officers of each of the Debtors is authorized and directed to execute, deliver, file, or record such contracts, instruments, releases, indentures, and other agreements or documents and take such actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.

#### SECTION 8

## EXECUTORY CONTRACTS AND UNEXPIRED LEASES

- **8.1** Executory Contracts and Unexpired Leases. On the Effective Date, all executory contracts and unexpired leases to which any of the Debtors are parties shall be deemed rejected as of the Effective Date, except for an executory contract or unexpired lease that (i) has been assumed or rejected pursuant to Final Order of the Bankruptcy Court entered prior to the Effective Date or (ii) is the subject of a separate motion to assume or reject filed under section 365 of the Bankruptcy Code by the Debtors prior to the Effective Date.
- **8.2** Approval of Rejection of Executory Contracts and Unexpired Leases. Entry of the Confirmation Order shall constitute the approval, pursuant to section 365(a) of the Bankruptcy Code, of the rejection of the executory contracts and unexpired leases rejected as of the Effective Date pursuant to the Plan.
- **8.3** Rejection Claims. In the event that the rejection of an executory contract or unexpired lease by any of the Debtors pursuant to the Plan results in damages to the other party or parties to such contract or lease, a Claim for such damages, if not heretofore evidenced by a filed proof of claim, shall be forever barred and shall not be enforceable against the Debtors, or any property to be distributed under the Plan unless a proof of claim is filed with the Bankruptcy Court and served upon the Plan Administrator on or before the date that is thirty (30) days after the Confirmation Date.

#### SECTION 9

#### **EFFECTIVENESS OF THE PLAN**

- **9.1** Condition Precedent to Confirmation of Plan. The following is a condition precedent to the confirmation of the Plan:
- (a) The Bankruptcy Court shall have entered a Confirmation Order in form and substance satisfactory to the Debtors.
- (b) (i) The Aggregate Allowed Consumer Class Action Claims shall be no greater than such amount that will yield an aggregate distribution to all such class claimants of \$350,000 including attorney fees or (ii) the aggregate Allowed amount of Class 5 Claims (General Unsecured Claims) shall be no greater than \$68,300,000.
- (c) The Bankruptcy Court shall have entered a Confirmation Order confirming the Plan which includes the Plan Release and Injunction in Favor of Settling Third Parties as provided in paragraph 4 of the 2002-2004 Ephedra PI Settlement Agreement.
- **9.2** <u>Conditions Precedent to Effective Date</u>. The following are conditions precedent to the Effective Date of the Plan:
  - (a) No stay of the Confirmation Order shall then be in effect;

- (b) The Ephedra Personal Injury Trust Agreement shall have been executed and made effective:
- (c) The Settlement Escrow Agent shall have paid to the Ephedra Personal Injury Trust the Third Party Settlement Amounts aggregating the Aggregate Third Party Settlement Amount;
- (d) The Debtors shall have paid to the Ephedra Personal Injury Trust the amount of the Class 4 Distribution; and
- The Debtors shall have sufficient available Cash and/or beneficial assets to pay the sum of (i) satisfying Administrative Expense Claims, Priority Tax Claims, Priority Non-Tax Claims, and Secured Claims, if applicable and to the extent such claims have not been paid, (ii) the reasonably necessary administrative expenses of the Debtors after the Effective Date. including contingent liabilities and the cost of continuation of litigation commenced by the Creditors Committee on behalf of the Debtors prior to the Effective Date, (iii) the reasonable administrative expenses of the Estate that have not been paid (including the Debtors' professional fees and expenses and the fees of the U.S. Trustee), (iv) satisfying all Disputed Claims (including, without limitation and as applicable, Consumer Class Action Claims, but not including Disputed Ephedra Personal Injury Claims) estimated by the Bankruptcy Court pursuant to section 502(c) of the Bankruptcy Code in the amount so estimated, (v) satisfying all Disputed Claims (but not including Disputed Ephedra Personal Injury Claims or Disputed Claims estimated by the Bankruptcy Court pursuant to section 502(c) of the Bankruptcy Code) in the event that all such Claims were to become Allowed Claims, (vi) other liabilities incurred by the Debtors in accordance with the Plan, and (vii) the amount required to otherwise perform the functions and take the actions provided for or permitted herein or in any other agreement executed by the Plan Administrator pursuant to the Plan.
- 9.3 <u>Satisfaction of Conditions</u>. Any actions required to be taken on the Effective Date shall take place and shall be deemed to have occurred simultaneously, and no such action shall be deemed to have occurred prior to the taking of any other such action. If the Debtors decide, after consultation with the Creditors Committee and the Ephedra Claimants Committee, that one of the conditions precedent set forth in Section 9.2 hereof cannot be satisfied and the occurrence of such condition is not waived or cannot be waived, then the Debtors shall file a notice of the failure of the Effective Date with the Bankruptcy Court.
- 9.4 Effect of Nonoccurrence of Conditions to Consummation. If each of the conditions to consummation and the occurrence of the Effective Date has not been satisfied or duly waived on or before the first Business Day that is one hundred twenty (120) days after the Confirmation Date, or such later date as shall be agreed by the Debtors and the Ephedra Claimants Committee, or by order of the Bankruptcy Court, the Confirmation Order may be vacated by the Bankruptcy Court. If the Confirmation Order is vacated pursuant to this Section, the Plan shall be null and void in all respects, and nothing contained in the Plan shall constitute a waiver or release of any Claims against any of the Debtors.

**9.5 Waiver of Conditions.** The Debtors and the Ephedra Claimants Committee may together (but not individually) waive by a writing signed by an authorized representative of the respective party and subsequently filed with the Bankruptcy Court, one or more of the conditions precedent set forth in Section 9.1 and Section 9.2 hereof, other than the conditions set forth in Sections 9.1(a), 9.1(c), 9.2(a), 9.2(b), and 9.2(e) hereof. The Debtors and the Settling Third Parties may together (but not individually) waive by writing signed by an authorized representative of the respective party and subsequently filed with the Bankruptcy Court the condition set forth in Section 9.1(c) hereto.

#### **SECTION 10**

### **EFFECT OF CONFIRMATION**

## 10.1 Vesting of Assets.

- (a) As of the Effective Date, the property of the Estates shall vest in the Debtors or such other entity as provided in the Plan.
- (b) From and after the Effective Date, the Plan Administrator and the Ephedra Personal Injury Trustee may dispose of, or cause to be disposed of, the assets of the Debtors and the Ephedra Personal Injury Trust respectively, free of any restrictions of the Bankruptcy Code, but in accordance with the provisions of the Plan and the Ephedra Personal Injury Trust Agreement.
- (c) As of the Effective Date, all assets of the Debtors and the Ephedra Personal Injury Trust shall be free and clear of all Claims, except as provided in the Plan or the Confirmation Order.
- **10.2 Binding Effect.** Except as otherwise provided in section 1141(d)(3) of the Bankruptcy Code, on and after the Confirmation Date, the provisions of the Plan shall bind any holder of a Claim against, or Equity Interest in, the Debtors and their respective successors and assigns, whether or not the Claim or Equity Interest of such holder is impaired under the Plan and whether or not such holder has accepted the Plan.
- Retention of Causes of Action Except as otherwise provided in the Plan, on and after the Effective Date, the Debtors will have the exclusive right to enforce any and all Causes of Action against any person. The Debtors may pursue, abandon, settle, or release any or all Causes of Action (other than Causes of Action released under the Plan), as it deems appropriate, without the need to obtain approval or any other or further relief from the Bankruptcy Court. The Plan Administrator may pursue, abandon, settle, or release any or all such Causes of Action as it deems appropriate, without the need to obtain approval or any other or further relief from the Bankruptcy Court. The Debtors may, in their sole discretion, offset any such claim held against a person against any payment due such person under the Plan; provided, however, that any claims of the Debtors arising before the Commencement Date shall first be offset against Claims against the Debtors arising before the Commencement Date.

**10.4** Exculpations and Limitation of Liability. Notwithstanding anything to the contrary contained herein, as of the Effective Date, none of (i) the Debtors and the Debtors' officers, directors, and employees, (ii) the Creditors Committee, (iii) the Ephedra Claimants Committee, and (iv) the members, representatives, accountants, financial advisors, investment bankers, consultants, and attorneys of the Entities described in (i) through (iii) of this paragraph shall have or incur any liability to any person for any act taken or omission, after the Commencement Date, in connection with or related to the Chapter 11 Cases or the operations of the Debtors' businesses during the Chapter 11 Cases, including but not limited to (i) formulating, preparing, disseminating, implementing, confirming, consummating or administrating the Plan (including soliciting acceptances or rejections thereof); (ii) the Disclosure Statement or any contract, instrument, release or other agreement or document entered into or any action taken or omitted to be taken in connection with the Plan or the Disclosure Statement; or (iii) any distributions made pursuant to the Plan, except for acts constituting willful misconduct or gross negligence.

# 10.5 Releases.

- Releases of the Settling Third Parties. In consideration of the 2002-(a) 2004 Ephedra PI Settlement Agreement and the Third Party Settlement Amounts, on the Effective Date, and except for the obligations arising under or in connection with this Plan, the Debtors, the Creditors Committee, the Ephedra Claimants Committee, and all Creditors, for themselves, their successors and assigns, and the Estates, shall fully, finally, and irrevocably release and forever discharge the Released Parties of and from any and all past, present, and future claims, rights, actions, demands, injuries, damages, compensation, or Causes of Action of whatsoever kind or nature, arising out of or otherwise relating to the manufacture, marketing, sale, distribution, fabrication, advertising, supply, production, use, or ingestion of ephedra or ephedra -containing products designed, manufactured, fabricated, sold, supplied, produced, or distributed by one or more of the Debtors, whether foreseen or unforeseen, known or unknown, contingent or actual, liquidated or unliquidated, whether in tort or contract, whether statutory or at common law (including, without limitation, claims under section 506(c) of the Bankruptcy Code) which they have or may have as of the Confirmation Date, arising from the beginning of the world up to the Confirmation Date, provided, however, that such release does not include claims against the Settling Third Parties in connection with the Pre-2002 Ephedra PI Claims; provided, further, that such release does limit the liability of GNC in each of the Pre-2002 Ephedra PI Claims to the proceeds of the Debtors' applicable insurance policy or policies and further limits the recovery of any holder of a Pre-2002 Ephedra PI Claim solely to such insurance proceeds in connection with any claim against GNC.
- Agreement, on the Effective Date, and except for the obligations arising under or in connection with this Plan, the Debtors, the Creditors Committee, the Ephedra Claimants Committee, and all Creditors, for themselves, their successors and assigns, the Estates, and any person claiming by, through, or under any of the foregoing shall fully, finally, and irrevocably release and forever discharge the Blechmans of and from any and all claims, rights, actions, demands, injuries, damages, compensation, or Causes of Action of every

kind and nature, whether foreseen or unforeseen, known or unknown, contingent or actual, liquidated or unliquidated, whether in tort or contract, whether statutory or at common law (including, without limitation, claims under section 506(c) of the Bankruptcy Code and any and all claims of creditors or shareholders of the Debtors) (i) asserted or that might have been asserted in the Amended Complaint or the Declaratory Judgment Complaint, (ii) arising out of or related to the facts and circumstances alleged in the Amended Complaint or Declaratory Judgment Complaint, (iii) arising from all alleged acts or omissions of every kind, nature, and description by any of the Blechmans arising out of their management of the Debtors or related to any of the Debtors, and (iv) based on allegations of product liability, personal injury (whether physical, emotional, or otherwise), wrongful death, or other statutory claims relating to products manufactured, fabricated, supplied, advertised, marketed, distributed, and/or sold by the Debtors) which they have or may have as of the Confirmation Date relating to the Debtors, arising from the beginning of the world up to the Confirmation Date.

(c) Releases by the Settling Third Parties and the Blechmans. For good and valuable consideration, on the Effective Date, and except for the obligations arising under or in connection with this Plan, the Blechmans and the Released Parties, for themselves, their successors and assigns, and any person claiming by, through, or under any of the foregoing shall fully, finally, and irrevocably release and forever discharge the Debtors and the Estates of and from any and all claims, rights, actions, demands, injuries, damages, compensation, or Causes of Action which they have or may have as of the Confirmation Date relating to the Debtors, arising from the beginning of the world up to the Confirmation Date; provided, however, that such release does not include release of Ephedra Indemnification Claims in connection with Pre-2002 Ephedra PI Claims, which claims will be treated in accordance with section 4.3 of the Plan.

### 10.6 Injunction.

All Entities who have held, hold, or may hold Claims against or (a) Equity Interests in any or all of the Debtors and other parties in interest, along with their respective present or former employees, agents, officers, directors, or principals, are permanently enjoined, on and after the Effective Date, with respect to all Claims against, and Equity Interests in, any of the Debtors from (i) commencing, conducting, or continuing in any manner, directly or indirectly, any Cause of Action or other proceeding of any kind (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against or affecting the Debtors, the Estates, the Plan Administrator, Ephedra Personal Injury Trust, the Ephedra Personal Injury Trustee, Ideasphere, the Released Parties, the Blechmans, or the property of any of the foregoing, (ii) enforcing, levying, attaching (including, without limitation, any prejudgment attachment), collecting, or otherwise recovering by any manner or means, whether directly or indirectly, any judgment, award, decree, or order against the Debtors, the Estates, the Plan Administrator, the Ephedra Personal Injury Trust, the Ephedra Personal Injury Trustee, Ideasphere, the Released Parties, the Blechmans, or the property of any of the foregoing, (iii) creating, perfecting, or otherwise enforcing in any manner, directly or indirectly, any encumbrance of any kind against the Debtors, the Estates, the Plan Administrator, the Ephedra Personal Injury Trust, the Ephedra Personal Injury Trustee, Ideasphere, the

Released Parties, the Blechmans, or against the property or interests in property of the Debtors, (iv) asserting any right of setoff, directly or indirectly, against any obligation due the Debtors, the Estates, the Plan Administrator, the Ephedra Personal Injury Trust, the Ephedra Personal Injury Trustee, Ideasphere, the Released Parties, the Blechmans, or any of the property of the foregoing, except as contemplated or allowed by the Plan; (v) acting or proceeding in any manner, in any place whatsoever, that does not conform to or comply with the provisions of the Plan; (vi) commencing, continuing or asserting in any manner any Cause of Action or other proceeding of any kind with respect to any Claims and Causes of Action which are extinguished or released pursuant to the Plan, and (vii) taking any actions to interfere with the implementation or consummation of the Plan.

Notwithstanding the foregoing, nothing contained in this paragraph shall prohibit holders of Pre-2002 Ephedra PI Claims from prosecuting such claims in accordance with section 4.3 of the Plan.

- All persons are permanently enjoined, on and after the Effective Date, (b) from asserting any Claim (x) which is released by such person under the Plan or (y) for which the party against whom such Claim is being asserted has received exculpation under the Plan, including: (i) commencing, conducting, or continuing in any manner, directly or indirectly, any Cause of Action or other proceeding of any kind (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) on account of such Claim, (ii) enforcing, levying attaching (including, without limitation, any prejudgment attachment), collecting, or otherwise recovering by any manner or means, whether directly or indirectly, any judgment, award, decree, or order on account of such Claim, (iii) creating, perfecting, or otherwise enforcing in any manner, directly or indirectly, any encumbrance of any kind on account of such Claim (iv) asserting any right of setoff, directly or indirectly, against any obligation on account of such claim (v) acting or proceeding in any manner, in any place whatsoever, that does not conform to or comply with the provisions of the Plan; (vi) commencing, continuing or asserting in any manner any Cause of Action or other proceeding of any kind with respect to any such Claim, and (vii) taking any actions to interfere with the implementation or consummation of the Plan.
- 10.7 **United States Exception.** Notwithstanding any provision of the Plan or any document or order associated therewith, nothing shall be deemed to waive, release, discharge, affect, or terminate any liability of, debt of, or Claim against the Debtors or any non-Debtor in connection with any criminal action or proceeding or civil forfeiture action or proceeding, or any action pursuant to the United States' police and regulatory power concerning conduct at any time by the Debtors (or its agents or former employees) and nothing herein shall release the Debtors or any non-Debtor from the criminal laws or civil forfeiture laws of the United States; provided that, except in connection with a properly filed proof of claim and to the extent that the Claim evidenced thereby is treated in accordance with the Plan, the foregoing provision does not permit the United States, any state of the United States, or any of their respective police or regulatory agencies to obtain any monetary recovery from or forfeit any assets of the Debtors or their property or interest in property with respect to any such Claim or other debt or liability that is discharged or any Equity Interest or other right of equity interest that is terminated or cancelled pursuant to the Plan, including, without limitation, any monetary claim, forfeiture, or penalty in furtherance of a police or regulatory power.

# 10.8 Terms of Pre-Plan Injunction and Stays.

- (a) Unless otherwise provided in the Plan, the Confirmation Order, or a separate order of the Bankruptcy Court, all injunctions or stays arising under or entered during the Chapter 11 Cases under section 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the later of the Effective Date and the date indicated in such applicable order.
- (b) Unless otherwise ordered by the Bankruptcy Court, on and after the Confirmation Date, any "fifty percent shareholder" within the meaning of section 382(g)(4)(D) of the Tax Code shall be enjoined from claiming a worthless stock deduction with respect to any Equity Interest held by such shareholder for any taxable year of such shareholder ending prior to the Effective Date.

#### **SECTION 11**

#### **RETENTION OF JURISDICTION**

- **11.1 Jurisdiction of Bankruptcy Court.** The Bankruptcy Court shall retain jurisdiction of all matters arising under, arising out of, or related to the Chapter 11 Cases and the Plan pursuant to, and for the purposes of, sections 105(a) and 1142 of the Bankruptcy Code and for, among other things, the following purposes:
- (a) To hear and determine motions for the assumption, assumption and assignment, or rejection of executory contracts or unexpired leases and the allowance of Claims resulting therefrom;
- (b) To determine any motion, adversary proceeding, application, contested matter, and other litigated matter pending on or commenced after the Confirmation Date, including, without limitation, any proceeding to recover a Cause of Action or Avoidance Action;
- (c) To ensure that distributions to holders of Allowed Claims are accomplished as provided herein;
- (d) Except as to the 2002-2004 Ephedra PI Claims but subject to sections 5.3 or 11.1(i) herein, to consider Claims or the allowance, classification, priority, compromise, estimation, or payment of any Claim;
- (e) To enter, implement, or enforce such orders as may be appropriate in the event the Confirmation Order is for any reason stayed, reversed, revoked, modified, or vacated;
- (f) To issue injunctions, enter and implement other orders, and take such other actions as may be necessary or appropriate to restrain interference by any person with the consummation, implementation, or enforcement of the Plan, the Confirmation Order, or any other order of the Bankruptcy Court;

- (g) To hear and determine any application to modify the Plan in accordance with section 1127 of the Bankruptcy Code, to remedy any defect or omission or reconcile any inconsistency in the Plan, the Disclosure Statement, or any order of the Bankruptcy Court, including the Confirmation Order, in such a manner as may be necessary to carry out the purposes and effects thereof;
- (h) To hear and determine all applications under sections 330, 331, and 503(b) of the Bankruptcy Code for awards of compensation for services rendered and reimbursement of expenses incurred prior to the Confirmation Date;
- (i) To hear and determine disputes arising in connection with the interpretation, implementation, or enforcement of the Plan, the Ephedra Personal Injury Trust, the Ephedra Personal Injury Trust Agreement, the Confirmation Order, any transactions or payments contemplated by any of the foregoing, or any agreement, instrument, or other document governing or relating to any of the foregoing;
- (j) To take any action and issue such orders as may be necessary to construe, enforce, implement, execute, and consummate the Plan or to maintain the integrity of the Plan following consummation;
- (k) To recover all assets of the Debtors and property of the Estates, wherever located;
- (l) To determine such other matters and for such other purposes as may be provided in the Confirmation Order;
- (m) To hear and determine matters concerning state, local, and federal taxes in accordance with sections 346, 505, and 1146 of the Bankruptcy Code (including, without limitation, matters with respect to any taxes payable by a trust established in furtherance of the Plan);
- (n) To hear and determine any other matters related hereto and not inconsistent with the Bankruptcy Code and title 28 of the United States Code; and
  - (o) To enter a final decree closing the Chapter 11 Cases.

#### **SECTION 12**

# MISCELLANEOUS PROVISIONS

12.1 <u>Dissolution of Creditors Committee and Ephedra Claimants Committee.</u>
The Creditors Committee and the Ephedra Claimants Committee shall dissolve on the Effective Date, or as soon thereafter as is reasonably practicable to permit the (i) Creditors Committee and the Ephedra Claimants Committee and their respective professionals the right to pursue, review, and object to any applications for compensation and reimbursement of expenses filed in accordance with Section 11.1(h) hereof and (ii) Creditors Committee the right to continue to

pursue, or compromise and settle, any litigation commenced by the Creditors Committee on behalf of the Debtors prior to the Effective Date.

- **12.2** <u>Substantial Consummation</u>. On the Effective Date, the Plan shall be deemed to be substantially consummated under sections 1101 and 1127(b) of the Bankruptcy Code.
- Payment of Statutory Fees. On the Effective Date, and thereafter as may be required, the Debtors shall pay all fees payable pursuant to section 1930 of chapter 123 of title 28 of the United States Code. Notwithstanding the deemed substantive consolidation of the Debtors for purposes of the Plan, each Debtor's calculation of and liability for quarterly fees shall remain unaffected, and each Debtor shall continue to pay all quarterly fees pursuant to section 1930 of chapter 123 of title 28 of the United States Code that become due after the Effective Date and, with respect to each Debtor, on or before the date that the Chapter 11 Case concerning such Debtor is (i) closed, (ii) dismissed, or (iii) converted to a case under Chapter 7 of the Bankruptcy Code.
- 12.4 <u>Effectuating Documents and Further Transactions</u>. Denis O'Connor, Operating Officer of Holdings, and an officer of each of the Debtors, is authorized and directed to execute, deliver, file, or record such contracts, instruments, releases, indentures, and other agreements or documents and take such actions as may be reasonably necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.
- 12.5 Exemption from Transfer Taxes. Pursuant to section 1146(c) of the Bankruptcy Code, the assignment or surrender of any lease or sublease, or the delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with the Plan, including any deeds, bills of sale, or assignments executed in connection with any disposition of assets contemplated by the Plan (including transfers of assets to and by the Ephedra Personal Injury Trust and transfers and assignments of assets to Ideasphere pursuant to the Ideasphere Agreement) shall not be subject to any stamp, real estate transfer, mortgage recording, sales, use, or other similar tax.
- shall, in the ordinary course of business and without the necessity for any approval by the Bankruptcy Court, pay the reasonable fees and expenses, incurred after the Confirmation Date, of the professional persons employed by the Debtors, the Ephedra Claimants Committee, and the Creditors Committee in connection with the implementation and consummation of the Plan, the claims reconciliation process, and any other matters as to which such professionals may be engaged. The fees and expenses of such professionals shall be paid within ten (10) Business Days after submission of a detailed invoice therefor. If the Plan Administrator disputes the reasonableness of any such invoice, the Plan Administrator shall timely pay the undisputed portion of such invoice, and the Plan Administrator or the affected professional may submit such dispute to the Bankruptcy Court for a determination of the reasonableness of such invoice.
- **12.7** Elimination of Vacant Classes. Any Class that is not occupied as of the commencement of the Confirmation Hearing by an Allowed Claim or a Claim temporarily

allowed under Bankruptcy Rule 3018 or as to which no vote is cast shall be deemed eliminated from the Plan for purposes of voting to accept or reject the Plan by such Class pursuant to section 1129(a)(8) of the Bankruptcy Code.

- **12.8 Nonconsensual Confirmation.** If any Impaired Class entitled to vote shall not accept the Plan by the requisite statutory majorities provided in section 1126(c) of the Bankruptcy Code, the Debtors reserve the right to request the Bankruptcy Court confirm the Plan under section 1129(b) of the Bankruptcy Code.
- 12.9 **Modification of Plan.** Upon reasonable notice to the Creditors Committee and the Ephedra Claimants Committee, the Plan may be amended, modified, or supplemented by the Debtors in the manner provided for by section 1127 of the Bankruptcy Code or as otherwise permitted by law without additional disclosure pursuant to section 1125 of the Bankruptcy Code, except as the Bankruptcy Court may otherwise direct. In addition, after the Confirmation Date, so long as such action does not materially adversely affect the treatment of holders of Claims or Equity Interests under the Plan, the Debtors may institute proceedings in the Bankruptcy Court to remedy any defect or omission or reconcile any inconsistencies in the Plan or the Confirmation Order, with respect to such matters as may be necessary to carry out the purposes and effects of the Plan. Prior to the Effective Date, the Debtors may make appropriate technical adjustments and modifications to the Plan without further order or approval of the Bankruptcy Court, provided that such technical adjustments and modifications do not adversely affect in a material way the treatment of holders of Claims or Equity Interests. Notwithstanding the foregoing, without the consent of the Settling Third Parties, the Plan may not be amended pursuant to this paragraph if after such amendment, the Plan would not include the Plan Release and Injunction in Favor of Settling Third Parties as provided in paragraph 4 of the 2002-2004 Ephedra PI Settlement Agreement.
- 12.10 <u>Revocation or Withdrawal of Plan</u> The Debtors reserve the right to revoke or withdraw the Plan at any time prior to the Confirmation Date. If the Debtors take such action, the Plan shall be deemed null and void. In such event, nothing contained herein shall be deemed to constitute a waiver or release of any Claim by or against the Debtors or any other person or to prejudice in any manner the rights of the Debtors or any other person in any further proceedings involving the Debtors.
- **12.11** Courts of Competent Jurisdiction. If the Bankruptcy Court abstains from exercising, or declines to exercise, jurisdiction or is otherwise without jurisdiction over any matter arising out of the Plan, such abstention, refusal, or failure of jurisdiction shall have no effect upon and shall not control, prohibit, or limit the exercise of jurisdiction by any other court having competent jurisdiction with respect to such matter.
- **12.12 Severability.** If, prior to the entry of the Confirmation Order, any term or provision of the Plan is held by the Bankruptcy Court to be invalid, void, or unenforceable, the Bankruptcy Court, at the request of the Debtors, shall have the power to alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void, or unenforceable, and such term or provision shall then be applicable as altered or interpreted. Notwithstanding any

such holding, alteration, or interpretation, the remainder of the terms and provisions of the Plan will remain in full force and effect and will in no way be affected, impaired, or invalidated by such holding, alteration, or interpretation. The Confirmation Order shall constitute a judicial determination and shall provide that each term and provision of the Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.

- 12.13 <u>Governing Law</u>. Except to the extent the Bankruptcy Code or other federal law is applicable, the rights, duties, and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with, the laws of the State of New York, without giving effect to the principles of conflicts of law thereof.
- **12.14 Exhibits.** The exhibits to the Plan are incorporated into and are a part of the Plan as if set forth in full herein.
- **12.15** <u>Successors and Assigns</u>. All the rights, benefits, and obligations of any person named or referred to in the Plan shall be binding on, and shall inure to the benefit of, the heirs, executors, administrators, successors, and/or assigns of such person.
- **12.16** <u>Time</u>. In computing any period of time prescribed or allowed by the Plan, unless otherwise set forth herein or determined by the Bankruptcy Court, the provisions of Bankruptcy Rule 9006 shall apply.
- 12.17 <u>Notices</u>. All notices, requests, and demands to or upon the Debtors to be effective shall be in writing (including by facsimile transmission) and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when actually delivered or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows:

If to the Debtors, to:

TL Administration Corporation c/o AlixPartners
9 West 57th Street, Suite 3420
New York, New York 10019
Attn: Denis O'Connor

Telephone: (646) 746-2465 Telecopier: (646) 746-2490

-and-

Weil, Gotshal & Manges LLP 767 Fifth Avenue New York, New York 10153 Attn: Michael P. Kessler, Esq. Telephone: (212) 310-8000

Telecopier: (212) 310-8007

Dated: New York, New York May 25, 2005

Respectfully submitted,

TL ADMINISTRATION CORPORATION (F/K/A TWINLAB CORPORATION)

By: /s/ Denis O'Connor

Name: Denis O'Connor Title: Vice President

TL ADMINISTRATION INC. (F/K/A TWIN LABORATORIES INC.)

By: /s/ Denis O'Connor

Name: Denis O'Connor Title: Vice President

TL ADMINISTRATION (UK) LTD. (F/K/A TWIN LABORATORIES (UK) LTD.)

By: /s/ Denis O'Connor

Name: Denis O'Connor Title: Vice President

UNITED STATES DISTRICT COURT SOUTHERN DISTRICT OF NEW YORK	
In re	: : 04 MD 1598 (JSR)
EPHEDRA PRODUCTS LIABILITY LITIGATION.	: :
PERTAINS TO ALL CASES	
UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK	
In re	: Chapter 11 Case No.
TL ADMINISTRATION CORPORATION, et al.),	: 03-15564 (RDD) : (Jointly Administered)
Debtorsx	:

ORDER CONFIRMING FIRST AMENDED JOINT PLAN
OF LIQUIDATION OF TL ADMINISTRATION CORPORATION
(F/K/A TWINLAB CORPORATION), TL ADMINISTRATION, INC.
(F/K/A TWIN LABORATORIES INC.), AND TL ADMINISTRATION (UK) LTD.
(F/K/A TWIN LABORATORIES (UK) LTD.)
UNDER CHAPTER 11 OF THE BANKRUPTCY CODE

The First Amended Joint Plan of Liquidation of TL Administration Corporation (f/k/a Twinlab Corporation), TL Administration, Inc. (f/k/a Twin Laboratories Inc.), and TL Administration (UK) Ltd. (f/k/a Twin Laboratories (UK) Ltd.) Under Chapter 11 of the Bankruptcy Code, dated May 25, 2005 (collectively, and as may be modified and amended, the "Plan") <sup>1</sup> and the First Amended Disclosure Statement thereto (the "Disclosure Statement") having been filed with the Court (defined herein) by TL Administration Corporation (f/k/a Twin Laboratories, Inc.)

<sup>&</sup>lt;sup>1</sup> Unless otherwise defined herein, capitalized terms shall have the meanings ascribed to such terms in the Plan. A copy of the Plan is attached hereto as Exhibit "A."

("Laboratories") and TL Administration (UK) Ltd. (f/k/a Twin Laboratories (UK) Ltd.) ("TL UK" and collectively with Holdings and Laboratories, the "Debtors"), as debtors and debtors in possession, and as proponents of the Plan within the meaning of section 1129 of title 11 of the United States Code (the "Bankruptcy Code"); and the Court having entered, after due notice and a hearing, an Order, dated June 3, 2005, (i) Approving First Amended Disclosure Statement, (ii) Fixing a Record Date, (iii) Approving Solicitation Packages and Procedures for Distribution Thereof, (iv) Approving Forms of Ballots and Procedures for Voting on the Debtors' First Amended Joint Plan of Liquidation, (v) Scheduling a Hearing and Establishing Notice and Objection Procedures in Respect of Confirmation of the Debtors' First Amended Joint Plan of Liquidation, and (vi) Authorizing the Retention of Bankruptcy Services, LLC and Financial Balloting Group LLC as Voting and Tabulation Agents (the "Disclosure Statement Order"); and the Disclosure Statement and Plan having been distributed to holders of Claims against and Equity Interests in the Debtors and other parties in interest as provided in the Disclosure Statement Order; and due notice of the Plan and the Confirmation Hearing having been provided to holders of Claims against and Equity Interests in the Debtors and other parties in interest in accordance with the Disclosure Statement Order, the Bankruptcy Code, and the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), as established by the affidavits of service, mailing, and/or publication filed with the Court (the 'Notice Affidavits'), <sup>2</sup> and such notice being sufficient under the circumstances and no further notice being required; and after full

<sup>&</sup>lt;sup>2</sup> The Notice Affidavits include the following: (i) Affidavit of Mailing of Rachel Brown, dated May 3, 2005 (District Court MDL Docket No. 1516); (ii) Affidavit of Kathy Gerber, dated June 15, 2005 (District Court MDL Docket No. 1645); (iii) Certification of Publication, dated June 14, 2005(District Court MDL Docket No. 1665); (iv) the affidavits filed by the Debtors' Court-appointed voting and tabulation agents, Bankruptcy Services LLC ("BSI") and Financial Balloting Group LLC ("FBG"), dated July 15, 2005 (as amended July 18, 2005) (the "Voting Affidavits") (District Court MDL Docket Nos. 1762 and 1763).

consideration of (i) the Debtors' memorandum of law in support of confirmation of the Plan, dated July 19, 2005 (the "Confirmation Memorandum"), (ii) the declarations in support of confirmation of the Plan, sworn to by (a) Denis O'Connor (the "O'Connor Declaration"), and (b) Rex A. Littrell (the "Littrell Declaration"), (iii) the Voting Affidavits regarding the solicitation of votes and methodology applied to the tabulation of the voting results with respect to the Plan, and the results of the tabulation of votes, (iv) the limited objection filed by the Ephedra Official Committee of Ephedra Claimants (the "Ephedra Claimants Committee"), and the joinders thereto filed by (a) the Plaintiffs' Coordinating Counsel appointed in the above-captioned Multi District Litigation, (b) Sallie Cooper, (c) Timothy and Sharon Sallis, (d) Joseph Michalowski, (e) Beverly A. Greenberg, (f) Jason Vickrey, and (g) Kenneth and Patricia Lowen, and (h) the "Barr Class" (collectively, the "Limited Objections"), (v) the objection filed by Clarence Lackowski and Christine Felts (the "Lackowski and Felts Objection"); (vi) the objection filed by the "Cirak Class" (the "Cirak Objection"), and the joinder thereto filed by the "Barr Class" (the "Barr Objection" and together with the Lackowski and Felts Objection and the Cirak Objection, the "Consumer Class Action Objections," (vii) the limited oral objection made at the Confirmation Hearing (as defined below) by the Indenture Trustee (together with the Limited Objections and the Consumer Class Objections, the "Objections"), and (viii) the Debtors' responses to the Objections contained in the Confirmation Memorandum; and the hearing to consider confirmation of the Plan having been held before the Court on July 21, 2005 (the "Confirmation" Hearing"); and the Court having considered the evidence adduced at the Confirmation Hearing, including the testimony of Messrs. O'Connor and Littrell, and Ms. Sullivan and Gerber, through their declarations, which were admitted into evidence, and the exhibits admitted into evidence, the arguments, representations, and agreements of counsel on the record; and all Objections

having been withdrawn, resolved, granted, or otherwise overruled as set forth herein; and upon all of the proceedings had before the Court and upon the arguments of counsel and all of the evidence adduced at the Confirmation Hearing; and the Court having determined based upon all of the foregoing that the Plan should be confirmed, as reflected by the Court's rulings made herein and at the Confirmation Hearing; and after due deliberation and sufficient cause appearing therefor, the Court hereby FINDS, DETERMINES, AND CONCLUDES that:

- A. <u>Findings and Conclusions</u>. The findings and conclusions set forth herein and in the record of the Confirmation Hearing constitute the Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to this proceeding pursuant to Bankruptcy Rule 9014. To the extent any of the following findings of fact constitute conclusions of law, they are adopted as such. To the extent any of the following conclusions of law constitute findings of fact, they are adopted as such.
- B. <u>Jurisdiction, Core Proceeding, Venue</u>. The Court has jurisdiction over the Debtors' chapter 11 cases, the Confirmation Hearing, the relief requested therein, including confirmation of the Plan, and the Objections, in accordance with 28 U.S.C. §§ 157(b) and 1334 and the Standing Order of Referral of Cases to Bankruptcy Court Judges of the District Court for the Southern District of New York (the "<u>District Court</u>"), dated July 19, 1984 (Ward, Acting C.J.). Consideration of the Confirmation Memorandum, the relief requested therein, and the Objections thereto is a core proceeding pursuant to 28 U.S.C. § 157(b). By Order dated March 22, 2005 (the "<u>Reference Order</u>" as amended by the District Court's Status Order (Including Case Management Order No. 12) dated April 15, 2005 and Status Order (Including Case Management Order No. 13) dated May 10, 2005), the District Court withdrew the bankruptcy reference with respect to the Debtors' chapter 11 cases from the United States Bankruptcy Court

for the Southern District of New York (the "Bankruptcy Court" and together with the District Court, the "Court") and assumed direct jurisdiction over all aspects of the Debtors' chapter 11 cases, except those specifically reserved to the Bankruptcy Court as set forth in the Reference Order and such additional referrals as the District Court may order in the future. The Reference Order provided that the District Court and the Bankruptcy Court "shall jointly retain jurisdiction over, and jointly hear...all hearings on the approval of a disclosure statement and confirmation of a chapter 11 plan of liquidation." Venue is proper before the Court pursuant to 28 U.S.C. §§ 1408 and 1409.

- C. <u>Commencement of the Chapter 11 Cases</u>. On September 4, 2003 (the "<u>Commencement Date</u>"), each of the Debtors commenced a case under chapter 11 of the Bankruptcy Code in the Bankruptcy Court (Case Nos. 03-15564 (Holdings), 03-15566 (Laboratories), and 03-15563 (TL (UK))). The Debtors' chapter 11 cases were consolidated for procedural purposes only and have been jointly administered pursuant to an order of the Bankruptcy Court dated September 4, 2004. The Debtors continue to manage their properties as debtors in possession pursuant to sections 1107(a) and 1108 of the Bankruptcy Code. No trustee has been appointed in these cases.
- D. <u>Judicial Notice</u>. The Court takes judicial notice of the dockets of the Debtors' chapter 11 cases and related and adversary proceedings maintained by the Clerk of the Bankruptcy Court, including, without limitation, all pleadings and other documents filed, all orders entered, and all evidence and arguments made, proffered, or adduced at the hearings held before the Bankruptcy Court during the pendency of the chapter 11 cases. Further, the Court takes judicial notice of the dockets of the cases in the above-captioned Multi District Litigation maintained by the Clerk of the District Court, including, without limitation, all pleadings and

other documents filed, all orders entered, and all evidence and arguments made, proffered, or adduced at the hearings held before the District Court during the pendency of the chapter 11 cases.

- E. <u>Burden of Proof.</u> The Debtors have the burden of proving the elements of sections 1129(a) and (b) of the Bankruptcy Code by a preponderance of the evidence.
- Solicitation and Notice. On June 3, 2005, the Court entered the Disclosure Statement Order, which, among other things, approved the Disclosure Statement, finding that it contained "adequate information" within the meaning of section 1125 of the Bankruptcy Code, and established procedures for the Debtors' solicitation of votes on the Plan. The Disclosure Statement, the Plan, the ballots, the notice of the Confirmation Hearing, and the Disclosure Statement Order (as applicable, the "Solicitation Materials") were served in compliance with the Bankruptcy Rules and the Disclosure Statement Order. As described in the Notice Affidavits (i) the service of the Solicitation Materials was adequate and sufficient under the circumstances of these Chapter 11 Cases and (ii) adequate and sufficient notice of the Confirmation Hearing, and other requirements, deadlines, hearings, and matters described in the Disclosure Statement Order was timely provided in compliance with the Bankruptcy Rules, the Disclosure Statement Order, and due process to all parties in interest including.
- G. <u>Voting</u>. Votes on the Plan were solicited after disclosure to holders of Claims against and Equity Interests in the Debtors of "adequate information" as defined in section 1125 of the Bankruptcy Code. As evidenced by the Voting Affidavits, votes to accept or reject the Plan have been solicited and tabulated fairly, in good faith, and in a manner consistent with the Disclosure Statement Order, the Bankruptcy Code, the Bankruptcy Rules, and the Local Bankruptcy Rules for the Southern District of New York (the "<u>Local Rules</u>").

- H. <u>Conditions Precedent to Confirmation</u>. As established by, among other things, the O'Connor Declaration, the Debtors have satisfied all conditions precedent to confirmation set forth in section 9.1(b) of the Plan.
- I. <u>Plan Compliance with the Bankruptcy Code (11 U.S.C. § 1129(a)(1))</u>. The Plan complies with the requirements of Bankruptcy Code sections 1122 and 1123 as well as other applicable provisions of the Bankruptcy Code.
- J. Proper Classification (11 U.S.C. §§ 1122). In accordance with Bankruptcy Code section 1122, the Plan provides for the separate classification of Claims against and Equity Interests in the Debtors based upon differences in the legal nature and/or priority of such Claims and Equity Interests. Section 3 of the Plan provides for the following six (6) Classes of Claims and one (1) Class of Equity Interests: Class 1 (Secured Claims), Class 2 (Priority Non-Tax Claims), Class 3 (Pre-2002 Ephedra PI Claims), Class 4 (2002-2004 Ephedra PI Claims), Class 5 (General Unsecured Claims), Class 6 (Punitive Damage Claims), and Class 7 (Equity Interests). Each of the Claims or Equity Interests in each particular Class is substantially similar to the other Claims or Equity Interests in such Class.
- K. <u>Designation of Classes of Claims and Equity Interests 11 U.S.C.</u>

  § 1123(a)(1)). Section 2 of the Plan provides for the treatment of Administrative Expense

  Claims and Priority Tax Claims and Section 3 of the Plan designates Classes of Claims and

  Classes of Equity Interests as required by Bankruptcy Code section 1123(a)(1).
- L. <u>Specified Unimpaired Classes (11 U.S.C. § 1123(a)(2))</u>. As required by Bankruptcy Code section 1123(a)(2), Section 4 of the Plan specifies that Class 1 (Secured Claims), Class 2 (Priority Non-Tax Claims), and Class 3 (Pre-2002 Ephedra PI Claims) are unimpaired under the Plan.

- M. Specified Treatment of Impaired Classes (11 U.S.C. § 1123(a)(3)). As required by Bankruptcy Code section 1123(a)(3), Section 4 of the Plan specifies the treatment of the Claims in Class 4 (2002-2004 Ephedra PI Claims), Class 5 (General Unsecured Claims), and Class 6 (Punitive Damage Claims), and the Equity Interests in Class 7 (Equity Interests), each of which is impaired under the Plan.
- N. <u>No Discrimination (11 U.S.C. § 1123(a)(4))</u>. Section 4 of the Plan provides that the treatment of each Claim or Equity Interest in each particular Class is the same as the treatment of each other Claim or Equity Interest in such Class, as required by Bankruptcy Code section 1123(a)(4).
- O. Implementation of the Plan (11 U.S.C. § 1123(a)(5)). As required by Bankruptcy Code section 1123(a)(5), the Plan provides adequate means for the Plan's implementation. Sections 5, 6, 7, 8 and various other provisions of the Plan set forth the means for implementation of the Plan, including the transfer of property to the Ephedra Personal Injury Trust, the appointment of an Ephedra Personal Injury Trustee, the appointment of a Plan Administrator, the methods for distribution to be made under the Plan for payment of claims and the rejection of executory contracts and unexpired leases.
- P. <u>Non-Voting Equity Securities / Allocation of Voting Power (11 U.S.C. § 1123(a)(6))</u>. Bankruptcy Code section 1123(a)(6) is inapplicable to the Plan because the Plan is a liquidating plan that provides for the dissolution of the Debtors and no distributions to Equity Interests and no distribution of new equity to any party in interest, except to the Plan Administrator and solely for purposes of winding down and liquidating the Debtors.
- Q. <u>Designation of Directors and Officers (11 U.S.C. § 1123(a)(7))</u>. Sections 7.2 and 7.3 of the Plan provide for the appointment on the Effective Date of the Plan

Administrator as the sole officer and director to serve the Debtors post-confirmation. This appointment meets the requirements of Bankruptcy Code section 1123(a)(7), is consistent with the interests of creditors and with public policy, and was approved by the Creditors Committee.

- R. <u>Additional Plan Provisions (11 U.S.C. § 1123(b))</u>. The additional provisions of the Plan are appropriate and consistent with the applicable provisions of the Bankruptcy Code, thereby satisfying section 1123(b) of the Bankruptcy Code.
- S. Impairment/Unimpairment of Classes of Claims or Equity Interests (§ 1123(b)(1)). As permitted under section 1123(b)(1) of the Bankruptcy Code, Sections 3 and 4 of the Plan render Class 1 (Secured Claims), Class 2 (Priority Non-Tax Claims), and Class 3 (Pre-2002 Ephedra PI Claims) unimpaired, and Class 4 (2002-2004 Ephedra PI Claims), Class 5 (General Unsecured Claims), Class 6 (Punitive Damage Claims), and Class 7 (Equity Interests) impaired.
- T. <u>Assumption and Rejection of Executory Contracts (11 U.S.C. §</u>

  1123(b)(2)). As permitted under section 1123(b)(2) of the Bankruptcy Code, Section 8 of the Plan provides for the rejection of executory contracts and unexpired leases of the Debtors not previously assumed or rejected under section 365 of the Bankruptcy Code.
- U. <u>Settlement or Adjustment of Claims and Provisions Not Inconsistent with</u>

  the Bankruptcy Code (11 U.S.C. §§ 1123(b)(3)(A) and 1123(b)(6)). As contemplated by sections 1123(b)(3)(A) and 1123(b)(6), the Plan incorporates the compromise and settlement of the 2002-2004 Ephedra PI Claims and Other Participating Ephedra Claims through the 2002-2004 Ephedra PI Settlement Agreement and certain releases and injunctions related and necessary thereto.

- V. <u>Debtors' Compliance with the Bankruptcy Code (11 U.S.C. § 1129(a)(2))</u>. Each of the Debtors is an eligible debtor under section 109 of the Bankruptcy Code. The Debtors have complied with the applicable provisions of the Bankruptcy Code, including the provisions of sections 1125 and 1126 of the Bankruptcy Code, and except as otherwise provided or permitted by orders of the Court. The Debtors have complied with the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules, and the Bankruptcy Court's Local Rules in transmitting the Plan, the Disclosure Statement, the Ballots, and related documents and notices and in soliciting and tabulating the votes on the Plan.
- W. Plan Proposed in Good Faith (11 U.S.C. § 1129(a)(3)). The Debtors have proposed the Plan in good faith and not by any means forbidden by law, thereby satisfying section 1129(a)(3) of the Bankruptcy Code. No objections have been filed to the Plan on the grounds of lack of good faith. Pursuant to Bankruptcy Rule 3020(b)(2), the Court may therefore determine that the Plan has been proposed in good faith and not by any means forbidden by law without receiving evidence on such issues. The Debtors' good faith is evident, however, from the facts and record of these chapter 11 cases, the Disclosure Statement, and the record of the Confirmation Hearing and other proceedings held in the Debtors' chapter 11 cases and the Multi District Litigation. The Plan provides for the liquidation of the Debtors, with distribution to creditors in accordance with the priorities afforded under the Bankruptcy Code. The Plan is the culmination of extensive good faith, arms-length negotiations among the Debtors, the Creditors Committee and the Ephedra Claimants Committee, among others. The Plan incorporates the 2002-2004 Ephedra PI Settlement Agreement, which itself was negotiated in accordance with the directives of the Court and is supported by the Debtors, their insurers, the Creditors Committee, the Ephedra Claimants Committee, the Settling Third Parties, and the individual holders of 2002-

2004 Ephedra PI Claims and Other Participating Ephedra Claims. The Plan was proposed with the legitimate and honest purpose of maximizing the value of the Debtors' estates for distribution to their creditors.

- X. Payment for Services or Costs and Expenses (11 U.S.C. § 1129(a)(4)).

  Any payment made or to be made by the Debtors for services or for costs and expenses in connection with their chapter 11 cases, or in connection with the Plan and incident to their chapter 11 cases, has been approved by, or is subject to the approval of, the Court as reasonable, thereby satisfying section 1129(a)(4) of the Bankruptcy Code.
- Y. <u>Directors, Officers, and Insiders (11 U.S.C. § 1129(a)(5))</u>. The Debtors have satisfied the requirements of section 1129(a)(5) of the Bankruptcy Code. On July 15, the Debtors filed a notice disclosing the identity of Denis O'Connor as the Plan Administrator and also describing his compensation to be earned as Plan Administrator.
- Z. <u>No Rate Changes (11 U.S.C. § 1129(a)(6))</u>. Section 1129(a)(6) of the Bankruptcy Code is inapplicable to the Plan. The Debtors are liquidating and no governmental regulatory commission will have jurisdiction over the rates of the Debtors.
- AA. Best Interest of Creditors (11 U.S.C. § 1129(a)(7)). The Plan satisfies section 1129(a)(7) of the Bankruptcy Code. The updated liquidation analysis attached to the O'Connor Declaration and supported by the O'Connor Declaration, and other evidence proffered or adduced at the Confirmation Hearing, (i) is persuasive and credible, (ii) has not been controverted by other evidence, and (iii) establishes that each holder of an impaired Claim or Equity Interest either has accepted the Plan or will receive or retain under the Plan, on account of such Claim or Equity Interest, property of a value, as of the Effective Date, that is not less than

the amount that such holder would receive or retain if the applicable Debtor were liquidated under chapter 7 of the Bankruptcy Code on such date.

BB. Acceptance by Certain Classes (11 U.S.C. § 1129(a)(8)). Class 1 (Secured Claims), Class 2 (Priority Non-Tax Claims), and Class 3 (Pre-2002 Ephedra PI Claims) are

Classes of unimpaired Claims that are conclusively presumed to have accepted the Plan pursuant to section 1126(f) of the Bankruptcy Code. Class 4 (2002-2004 Ephedra PI Claims)

(unanimously) and Class 5 (General Unsecured Claims) (by a substantial majority) have voted to accept the Plan in accordance with sections 1126(b) and (c) of the Bankruptcy Code. Class 6

(Punitive Damage Claims) and Class 7 (Equity Interests) are impaired by the Plan and are not entitled to receive or retain any property under the Plan and, therefore, are deemed to have rejected the Plan pursuant to section 1126(g) of the Bankruptcy Code. Although section 1129(a)(8) is not satisfied with respect to the deemed rejection of Class 6 and Class 7, the Plan may nevertheless be confirmed because the Plan satisfies section 1129(b) of the Bankruptcy Code with respect to Class 6 and Class 7.

CC. Treatment of Administrative Expense Claims, Tax Claims, and Other

Priority Claims (11 U.S.C. § 1129(a)(9)). The Plan satisfies the requirements of section

1129(a)(9) of the Bankruptcy Code. In accordance with section 1129(a)(9)(A), the Plan provides
that, on the Effective Date, the Debtors shall pay each holder of an Allowed Administrative

Expense Claim an amount in Cash equal to the Allowed amount of such Claim. In accordance
with section 1129(a)(9)(B), the Plan provides that each holder of an Allowed Priority Non-Tax

Claim shall be paid an amount in Cash equal to the Allowed amount of such Priority Non-Tax

Claim. In accordance with section 1129(a)(9)(C), the Plan provides that each holder of an

Allowed Priority Tax Claim shall be paid an amount in Cash equal to the amount of such Claim.

All Allowed Priority Tax Claims not due and payable on or before the Effective Date shall be paid in the ordinary course of business as such obligations become due.

DD. Acceptance By Impaired Class of Claims (11 U.S.C. § 1129(a)(10)). As set forth in the Voting Affidavits, at least one Class of impaired claims against each Debtor has accepted the Plan, determined without including any acceptance of the Plan by insiders, thereby satisfying the requirements of section 1129(a)(10) of the Bankruptcy Code. Specifically, both Classes of impaired Claims entitled to vote on the Plan (Class 4 and Class 5) have voted to accept the Plan by the requisite majorities mandated under section 1126 of the Bankruptcy Code, determined without including any acceptance of the Plan by any insider. The evidence set forth in the Voting Affidavits (i) is persuasive and credible, (ii) has not been controverted by other evidence, and (iii) establishes the voting results with respect to the Plan.

EE. Feasibility (11 U.S.C. § 1129(a)(11)). Because the Plan provides for the liquidation of the Debtors, the Plan is feasible if the Debtors will be able to satisfy the conditions precedent to the Effective Date (as defined in the Plan) and otherwise have sufficient funds to meet their post-Confirmation Date obligations to pay for the costs of administering and fully consummating the Plan and closing the chapter 11 cases. The evidence set forth in the O'Connor Declaration and otherwise proffered or adduced at the Confirmation Hearing (i) is persuasive and credible, (ii) has not been controverted by other evidence, and (iii) establishes that the Plan is feasible and that the Debtors will have sufficient funds to meet their post-confirmation date obligations, thereby satisfying the requirements of section 1129(a)(11) of the Bankruptcy Code.

FF. Payment of Fees (11 U.S.C. § 1129(a)(12)). In accordance with sections 507 and 1129(a)(12) of the Bankruptcy Code, the Plan provides that all fees due and payable

pursuant to section 1930 of chapter 123 of title 28 of the United States Code have been or will be paid on the Effective Date or as soon thereafter as is practicable.

GG. <u>Continuation of Retiree Benefits (11 U.S.C. § 1129(a)(13))</u>. The Debtors do not have obligations to provide retiree benefits. Accordingly, the Bankruptcy Code section 1129(a)(13) is inapplicable to the Plan.

HH. Subordination of Punitive Damage Claims (11 U.S.C §§ 726(a)(4), 1129(a)(7), and 105(a)). Punitive Damage Claims are properly subordinated by the Plan based on (i) the interplay between sections 724(a)(4) and 1129(a)(7) of the Bankruptcy Code, and/or (ii) the equitable powers provided under section 105(a) of the Bankruptcy Code. The Debtors' estates are not sufficient to pay all unsecured claims in full. Under a chapter 7 liquidation, no Punitive Damage Claims would be payable out of the Debtors' estates. Assuming section 726(a)(4) applies in analyzing the best interests test, the Plan would not pass the best interests test if it did not separately classify and subordinate Punitive Damage Claims or if it proposed to pay any Punitive Damages to creditors (absent unanimous consent by all voting creditors). In addition, Punitive Damage Claims serve neither a deterrent nor a punishment purpose in the Debtors' cases. The Debtors ceased operating their businesses upon the closing of the sale of substantially all of the their assets in December 2003 and are now liquidating. Moreover, distributions on account of Punitive Damage Claims would visit a hardship on creditors for wrongs attributed to others and would be inequitable. Furthermore, no objections were filed with respect to the separate classification and subordination of Punitive Damage Claims.

II. No Unfair Discrimination; Fair and Equitable (11 U.S.C. § 1129(b)).

Class 6 and Class 7 are deemed to have rejected the Plan. Based upon the pleadings submitted and the evidence proffered, adduced, and/or presented by the Debtors at or before the

Confirmation Hearing, the Plan does not discriminate unfairly and is fair and equitable with respect to Class 6 (Punitive Damage Claims) and Class 7 (Equity Interests), as required by sections 1129(b)(1) and (b)(2) of the Bankruptcy Code. The Plan does not discriminate against Class 6 because (i) Class 6 is comprised of Claims dissimilar to the Claims and Equity Interests in all other Classes and (ii) taking into account the particular facts and circumstances of the case, there is a reasonable basis for disparate treatment. See paragraph HH above. Class 7 is the only Class consisting of Equity Interests under the Plan. Accordingly, the Plan does not discriminate unfairly as between any similarly situated holders of equity interests in the Debtors. Equity Interests in Class 7 are not entitled to receive a distribution under the Plan. Therefore, no Claims or Equity Interests junior to Class 6 Claims will receive property under the Plan and the Plan satisfies the absolute priority rule with respect to Class 6 pursuant to section 1129(b)(2)(B) of the Bankruptcy Code. There are no Classes junior to Class 7, and creditors in the Classes senior to Class 7 will not be paid in full. The Plan therefore satisfies the absolute priority rule of section 1129(b)(2)(C) of the Bankruptcy Code with respect to Class 7.

JJ. Approval of the 2002-2004 Ephedra PI Settlement Agreement

(Bankruptcy Rule 9019(a), 11 U.S.C. § 1123(b)(6)). The evidence and arguments made,

proffered, or adduced at the hearings held before the Court during the pendency of the chapter 11

cases, including the Disclosure Statement, the O'Connor Declaration, and the Littrell

Declaration, and representations of counsel, demonstrate that (i) the Debtors' exercised their sound business judgment in entering into the 2002-2004 Ephedra PI Settlement Agreement, which was the result of arms length, good faith and arduous multi-party Court-ordered mediation, (ii) the 2002-2004 Ephedra PI Settlement Agreement is fair and equitable, reasonable, and in the best interests of the Debtors' estates, and (iii) clearly falls above the lowest range of

reasonableness. The evidence and arguments are persuasive and credible and have not been controverted by other evidence. The Ephedra PI Settlement Agreement was negotiated in good faith, at arms length, in accordance with the directives of the Court, and is supported by the Debtors, their insurers, the Creditors Committee, the Ephedra Claimants Committee, the Settling Third Parties, and the individual holders of 2002-2004 Ephedra PI Claims and Other Participating Ephedra Claims. The 2002-2004 Ephedra PI Settlement Agreement will allow the Debtors to fairly and efficiently liquidate their estates by resolving their largest outstanding contingencies. The establishment of the Ephedra Personal Injury Trust and the channeling of Class 4 Claims to the Ephedra Personal Injury Trust is a valid plan mechanism for resolving unliquidated claims to the substantial benefit of all impaired Claims. Accordingly, the Court approves the 2002-2004 Ephedra PI Settlement Agreement

KK. Plan Releases and Injunctions (Bankruptcy Rule 9019(a), 11 U.S.C. §§

1123(b)(3)(A) and 1123(b)(6)). The Court has jurisdiction pursuant to 28 U.S.C. §§ 1334(b) and
157(a) to approve the releases and injunctions set forth in Sections 10.5 and 10.6 of the Plan.

Additionally, the Court has jurisdiction over each of the parties in the Multi District Litigation.

The evidence and arguments made, proffered, or adduced at the hearings held before the Court during the pendency of the chapter 11 cases, including the Disclosure Statement, the O'Connor Declaration, and the Littrell Declaration, demonstrates that the incorporation of the releases in Section 10.5 of the Plan (as modified by this Confirmation Order), including of the Released Parties pursuant to Section 10.5(a) of the Plan, was (i) fair and equitable, reasonable, and in the best interests of the Debtors' estates and creditors, and (ii) supported by special circumstances justifying such unusual relief. The release of the Released Parties pursuant to Section 10.5(a) of the Plan was in consideration for the 2002-2004 Ephedra PI Settlement Agreement and Third

Party Settlement Amounts, which form the basis for, and constitute the primary source for the funding of, the Plan. Section 105(a) of the Bankruptcy Code permits the issuance of injunctions and approval of the releases set forth in Sections 10.5 (as modified by this Confirmation Order) and 10.6 of the Plan if, as has been established here, such provisions are essential to the formulation and implementation of the Plan, are made in return for substantial consideration channeled to a settlement fund, and therefore, confer a material benefit on the Debtors' estates and creditors. The third-party release and injunction provisions of the Plan (as modified by this Confirmation Order) approved by this Confirmation Order are consistent with sections 105, 1123, and 1129 of the Bankruptcy Code. They were negotiated at arms length during Court-ordered mediation of the central issues in these chapter 11 cases, and are the quid pro quo for a settlement fund that materially benefits all unsecured creditors of the estate, without which it is unlikely that any chapter 11 plan would be confirmed.

LL. Notwithstanding the Court's denial of the Debtors' request for approval of the release of the Blechmans contained in Section 10.5(b) of the Plan and related injunctive relief, for the reasons stated in the Court's bench ruling, including that the proposed release of the Blechmans was not, unlike the release and injunction for the benefit of the Released Parties, necessary to the formulation and implementation of the Plan, the Debtors and the Creditors Committee have complied with their obligations under the Blechman Agreement with respect thereto.

MM. <u>Satisfaction of Confirmation Requirements</u>. Based upon the foregoing, the Plan satisfies the requirements for confirmation set forth in section 1129 of the Bankruptcy Code.

NN. <u>Retention of Jurisdiction</u>. The Court may properly, and upon the Effective Date shall, retain jurisdiction over the matters set forth in Section 11 of the Plan and section 1142 of the Bankruptcy Code.

ACCORDINGLY, IT IS HEREBY ORDERED, ADJUDGED AND DECREED THAT:

- 1. <u>Solicitation and Notice</u>. Notice of the Confirmation Hearing complied with the terms of the Disclosure Statement Order, was appropriate and satisfactory based upon the circumstances of the Debtors' Chapter 11 Cases, and was in compliance with the provisions of the Bankruptcy Code, the Bankruptcy Rules, and the Local Rules. The solicitation of votes on the Plan complied with the solicitation procedures in the Disclosure Statement Order, was appropriate and satisfactory based upon the circumstances of the Debtors' Chapter 11 Cases, and complied with the provisions of the Bankruptcy Code, the Bankruptcy Rules, and the Local Rules.
- 2. <u>Confirmation</u>. The Plan and each of its provisions (except as modified by this Confirmation Order) are hereby confirmed under section 1129 of the Bankruptcy Code. The terms of the Plan (except as modified by this Confirmation Order) and the exhibits thereto, including the 2002-2004 Ephedra PI Settlement Agreement, are incorporated by reference into and are an integral part of the Plan and this Confirmation Order.
- 3. <u>Binding Effect</u>. Except as otherwise provided in section 1141(d)(3) of the Bankruptcy Code, on or after entry of this Confirmation Order and subject to the occurrence of the Effective Date, the provisions of the Plan shall bind any holder of a Claim against or Equity Interest in the Debtors and their respective successors and assigns, whether or not such Claim or

Equity Interest of such holder is impaired under the Plan and whether or not such holder has accepted the Plan.

- 4. <u>Plan Settlements</u>. To the extent that the Plan provides for the compromise or settlement of Claims and controversies resolved pursuant to the Plan, this Confirmation Order shall constitute an order under Bankruptcy Rule 9019 approving such compromise. The 2002-2004 Ephedra PI Settlement Agreement is hereby approved in its entirety.
- 5. <u>Dismissal of Cases</u>. On the date this Confirmation Order becomes a Final Order, each of the cases subject to the 2002-2004 Ephedra PI Settlement Agreement (listed on Exhibit "B" attached hereto) shall be dismissed with prejudice; provided, however, that such dismissals shall not include the dismissal of claims against Metabolife International, Inc. In accordance with paragraph 5 of the 2002-2004 Ephedra PI Settlement Agreement, as soon as is practicable, counsel for the Plaintiffs (as that term is defined in the 2002-2004 Ephedra PI Settlement Agreement) shall serve upon the Special Master the confidential breakdown of the settlement amount to be paid to each individual plaintiff and the amount to be paid in fees and disbursements to the plaintiff's individual counsel and to coordinating counsel.
- 6. Exculpations and Limitation of Liability. As of the Effective Date, none of (i) the Debtors and the Debtors' officers, directors, and employees, (ii) the Creditors

  Committee, (iii) the Ephedra Claimants Committee, and (iv) the members, ad hoc members, representatives, accountants, financial advisors, investment bankers, consultants, and attorneys of the Entities described in (i) through (iii) of this paragraph shall have or incur any liability to any person for any act taken or omission in such capacity, after the Commencement Date, in connection with or related to the Chapter 11 Cases or the operations of the Debtors' businesses during the Chapter 11 Cases, including but not limited to (i) formulating, preparing,

disseminating, implementing, confirming, consummating or administrating the Plan (including soliciting acceptances or rejections thereof); (ii) the Disclosure Statement or any contract, instrument, release or other agreement or document entered into or any action taken or omitted to be taken in connection with the Plan or the Disclosure Statement; or (iii) any distributions made pursuant to the Plan. Nothing in the foregoing or Section 10.4 of the Plan shall (i) be construed to release or exculpate any non-Debtor from fraud, gross negligence, willful misconduct, malpractice, criminal conduct, unauthorized use of confidential information that causes damages, or ultra vires acts, or (ii) limit the liability of the professionals of the Debtors to their clients pursuant to DR 6-102 of the Code of Professional Responsibility.

7. Releases of the Released Parties. Pursuant to Section 10.5(a) of the Plan, on the Effective Date, and except for the obligations arising under or in connection with the Plan, the Debtors, the Creditors Committee, the Ephedra Claimants Committee, and all Creditors, for themselves, their successors and assigns, and the Estates, shall fully, finally, and irrevocably release and forever discharge the Released Parties of and from any and all past, present, and future claims, rights, actions, demands, injuries, damages, compensation, or Causes of Action of whatsoever kind or nature, arising out of or otherwise relating to the manufacture, marketing, sale, distribution, fabrication, advertising, supply, production, use, or ingestion of ephedra or ephedra-containing products designed, manufactured, fabricated, sold, supplied, produced, or distributed by one or more of the Debtors, whether foreseen or unforeseen, known or unknown, contingent or actual, liquidated or unliquidated, whether in tort or contract, whether statutory or at common law (including, without limitation, claims under section 506(c) of the Bankruptcy Code) which they have or may have as of the Confirmation Date, arising from the beginning of the world up to the Confirmation Date, provided, however, that such release does not include

claims against the Settling Third Parties in connection with the Pre-2002 Ephedra PI Claims; provided, further, that such release does limit the liability of GNC in each of the Pre-2002 Ephedra PI Claims to the proceeds of the Debtors' applicable insurance policy or policies and further limits the recovery of any holder of a Pre-2002 Ephedra PI Claim solely to such insurance proceeds in connection with any claim against GNC; provided, further, however, that the foregoing limitations contained herein and in Section 10.5(a) of the Plan in connection with the Pre-2002 Ephedra PI Claims and on the recovery of any holder of a Pre-2002 Ephedra PI Claim against GNC shall not apply to, encompass, preclude, or limit the liability of GNC for or to any claim for Punitive Damages asserted in or by the Pre-2002 Ephedra PI Claims against GNC.

- 8. Releases by the Released Parties. Pursuant to Section 10.5(c) of the Plan, on the Effective Date, and except for the obligations arising under or in connection with this Plan, the Released Parties, for themselves, their successors and assigns, and any person claiming by, through, or under any of the foregoing shall fully, finally, and irrevocably release and forever discharge the Debtors and the Estates of and from any and all claims, rights, actions, demands, injuries, damages, compensation, or Causes of Action which they have or may have as of the Confirmation Date relating to the Debtors, arising from the beginning of the world up to the Confirmation Date; provided, however, that such release does not include release of Ephedra Indemnification Claims in connection with Pre-2002 Ephedra PI Claims, which claims will be treated in accordance with section 4.3 of the Plan.
- 9. <u>Injunction</u>. Pursuant to Section 10.6(a) of the Plan, all Entities who have held, hold, or may hold Claims against or Equity Interests in any or all of the Debtors and other parties in interest, along with their respective present or former employees, agents, officers, directors, or principals, are permanently enjoined, on and after the Effective Date, with respect to

all Claims against, and Equity Interests in, any of the Debtors from (i) commencing, conducting, or continuing in any manner, directly or indirectly, any Cause of Action or other proceeding of any kind (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) against or affecting the Debtors, the Estates, the Plan Administrator, Ephedra Personal Injury Trust, the Ephedra Personal Injury Trustee, Ideasphere, the Released Parties, or the property of any of the foregoing, (ii) enforcing, levying, attaching (including, without limitation, any prejudgment attachment), collecting, or otherwise recovering by any manner or means, whether directly or indirectly, any judgment, award, decree, or order against the Debtors. the Estates, the Plan Administrator, the Ephedra Personal Injury Trust, the Ephedra Personal Injury Trustee, Ideasphere, the Released Parties, or the property of any of the foregoing, (iii) creating, perfecting, or otherwise enforcing in any manner, directly or indirectly, any encumbrance of any kind against the Debtors, the Estates, the Plan Administrator, the Ephedra Personal Injury Trust, the Ephedra Personal Injury Trustee, Ideasphere, the Released Parties, or against the property or interests in property of the Debtors, (iv) asserting any right of setoff, directly or indirectly, against any obligation due the Debtors, the Estates, the Plan Administrator, the Ephedra Personal Injury Trust, the Ephedra Personal Injury Trustee, Ideasphere, the Released Parties, or any of the property of the foregoing, except as contemplated or allowed by the Plan; (v) acting or proceeding in any manner, in any place whatsoever, that does not conform to or comply with the provisions of the Plan; (vi) commencing, continuing or asserting in any manner any Cause of Action or other proceeding of any kind with respect to any Claims and Causes of Action which are extinguished or released pursuant to the Plan, and (vii) taking any actions to interfere with the implementation or consummation of the Plan. Notwithstanding the foregoing, nothing contained in this paragraph shall prohibit holders of Pre-2002 Ephedra PI Claims from

prosecuting such claims in accordance with section 4.3 of the Plan. Nothing contained in this Confirmation Order or the Plan shall affect, impair, or modify the coverage obligations of any of the Debtors' insurers other than the obligations of AISLIC in connection with the 2002-2004 Ephedra PI Claims which are released in accordance with the terms of the Plan, the 2002-2004 Ephedra PI Settlement Agreement, and this Confirmation Order.

- 10. Pursuant to Section 10.6(b) of the Plan, all persons are permanently enjoined, on and after the Effective Date, from asserting any Claim (x) which is released by such person under the Plan (as modified by this Confirmation Order) or (y) for which the party against whom such Claim is being asserted has received exculpation under the Plan, including: (i) commencing, conducting, or continuing in any manner, directly or indirectly, any Cause of Action or other proceeding of any kind (including, without limitation, any proceeding in a judicial, arbitral, administrative or other forum) on account of such Claim, (ii) enforcing, levying attaching (including, without limitation, any prejudgment attachment), collecting, or otherwise recovering by any manner or means, whether directly or indirectly, any judgment, award, decree, or order on account of such Claim, (iii) creating, perfecting, or otherwise enforcing in any manner, directly or indirectly, any encumbrance of any kind on account of such Claim (iv) asserting any right of setoff, directly or indirectly, against any obligation on account of such claim (v) acting or proceeding in any manner, in any place whatsoever, that does not conform to or comply with the provisions of the Plan; (vi) commencing, continuing or asserting in any manner any Cause of Action or other proceeding of any kind with respect to any such Claim, and (vii) taking any actions to interfere with the implementation or consummation of the Plan.
- 11. <u>Terms of Pre-Plan Injunctions and Stays</u>. Pursuant to Section 10.8 of the Plan, unless otherwise provided in the Plan, the Confirmation Order, or a separate order of the

Bankruptcy Court, all injunctions or stays arising under or entered during the Chapter 11 Cases under section 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the later of the Effective Date and the date indicated in such applicable order. Unless otherwise ordered by the Bankruptcy Court, on and after the Confirmation Date, any "fifty percent shareholder" within the meaning of section 382(g)(4)(D) of the Tax Code shall be enjoined from claiming a worthless stock deduction with respect to any Equity Interest held by such shareholder for any taxable year of such shareholder ending prior to the Effective Date.

Deemed Consolidation for Plan Purposes. Pursuant to Section 7.1 of the Plan, on the Effective Date, (i) all guaranties of any Debtor of the payment, performance, or collection of obligations of another Debtor shall be eliminated and canceled, (ii) any obligation of any Debtor and all guaranties thereof executed by one or more of the other Debtors shall be treated as a single obligation, and such guaranties shall be deemed a single Claim against the deemed consolidated Debtors, (iii) all joint obligations of two or more Debtors and all multiple Claims against such entities on account of such joint obligations shall be treated and allowed only as a single Claim against the deemed consolidated Debtors, (iv) all Claims between or among the Debtors shall be canceled, and (v) each Claim filed in the Chapter 11 Case of any Debtor shall be deemed filed against the deemed consolidated Debtors and a single obligation of the deemed consolidated Debtors on and after the Effective Date. Such deemed consolidation, however, shall not affect the legal and organizational structure of the Debtors and each of TL Administration Corporation, TL Administration Inc., and TL Administration (UK) ltd. shall be governed by their respective certificate of incorporation and by-laws.

- 13. <u>Plan Administrator</u>. Pursuant to Section 7.2 of the Plan, Denis O'Connor, the Debtors' current Operating Officer and a Managing Director at AlixPartners, LLC, the Debtors current financial advisors, is hereby appointed to serve without bond as the Plan Administrator to implement or cause to be implemented the terms of the Plan with respect to the consolidated Debtors; <u>provided</u> that Mr. O'Connor's performance in making Plan distributions is covered by insurance of AlixPartners, LLC. The Plan Administrator shall be compensated according to terms substantially similar to the terms approved by orders of the Bankruptcy Court dated September 4, 2003, and July 14, 2004, for the retention of the Debtors' financial advisors, FTI Consulting, LLC and AlixPartners, LLC, respectively (i.e., at the normal hourly billing rates and disbursement charges billed to his other clients in the ordinary course).
- each of the actions and effect each of the transfers required by the terms of the Plan, in the time period allocated therefor, and all matters provided for under the Plan that would otherwise require approval of the stockholders, partners, members, directors, or comparable governing bodies of the Debtors shall be deemed to have occurred and shall be in effect from and after the Effective Date pursuant to the applicable general corporation law (or other applicable governing law) of the states in which the Debtors are incorporated or organized, without any requirement of further action by the stockholders, members, or directors (or other governing body) of the Debtors. Each of the Debtors shall be authorized and directed, following the completion of all disbursements, other transfers, and other actions required of the Debtors by the Plan, to file its certificate of cancellation or dissolution as contemplated by Section 7.9 of the Plan. The filing of such certificates of cancellation, dissolution, or merger shall be authorized and approved in all respects without further action under applicable law, regulation, order, or rule, including, without

express or implied limitation, any action by the stockholders, members, or directors (or other governing body) of the Debtors.

- 15. <u>Effectuating Documents and Further Transactions</u>. Each of the officers of each of the Debtors is authorized and directed to execute, deliver, file, or record such contracts, instruments, releases, indentures, and other agreements or documents and take such actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan.
- 16. Vesting of Assets (11 U.S.C. § 1141(b), (c)). Pursuant to Section 10.1 of the Plan, as of the Effective Date, and pursuant to sections 1141(b) and (c) of the Bankruptcy Code, all property of the Debtors' estates shall vest in the Debtors or such other entity as provided in the Plan, free and clear of all Claims, liens, encumbrances, charges, and other interests, except as provided in the Plan or in this Confirmation Order. From and after the Effective Date, the Plan Administrator and the Ephedra Personal Injury Trustee may dispose of, or cause to be disposed of, the assets of the Debtors and the Ephedra Personal Injury Trustee, free of any restrictions of the Bankruptcy Code, but in accordance with the provisions of the Plan and the Ephedra Personal Injury Trust Agreement.
- 17. Effective Date Payments and Transfers. On the Effective Date, or, as soon thereafter as reasonably practicable, the Plan Administrator shall (i) remit, to each holder of an Allowed Non-Ephedra Claim (as of the Distribution Record Date) against the Debtors, the distribution provided for such Claim under the Plan and (ii) pay and transfer to the Ephedra Personal Injury Trust (or cause the Settlement Escrow Agent to pay and transfer to the Ephedra Personal Injury Trust) the Cash and assets to be paid and transferred to the Ephedra Personal Injury Trust pursuant to section 5.3 of the Plan.

- 18. The Objections. Having examined the proposed third-party releases of the Blechmans contained in the Plan and having considered the Cirak Objection and the argument of counsel for Lackowski and Felts, (i) the releases and injunctions in favor of the Blechmans contained in the Plan are not approved; (ii) Section 10.5(b) of the Plan shall be void and without effect; (iii) the injunctions provided for in Section 10.6 of the Plan in favor of the Blechmans are void and without effect; and (iv) insofar as the Cirak Objection and argument of other counsel sought disapproval of the proposed third-party releases in favor of the Blechmans in the Plan, such objection is granted. The remainder of the Cirak Objection and all other Objections, responses to, and statements and comments in opposition to the Plan, other than those withdrawn with prejudice in their entirety prior to the Confirmation Hearing or otherwise resolved, shall be, and hereby are, overruled for the reasons stated on the record.
- 19. Retention of Causes of Action. Pursuant to Section 10.3 of the Plan and except as otherwise provided in the Plan (including section 12.1 of the Plan), on and after the Effective Date, the Debtors will have the exclusive right to enforce any and all Causes of Action against any person. The Debtors may pursue, abandon, settle, or release any or all Causes of Action (other than Causes of Action released under the Plan), as they deem appropriate, without the need to obtain approval or any other or further relief from the Bankruptcy Court. The Plan Administrator may pursue, abandon, settle, or release any or all such Causes of Action as it deems appropriate, without the need to obtain approval or any other or further relief from the Bankruptcy Court. The Debtors may, in their sole discretion, offset any such claim held against a person against any payment due such person under the Plan; provided, however, that any claims of the Debtors arising before the Commencement Date shall first be offset against Claims against the Debtors arising before the Commencement Date.

- 20. Objections to Claims. Pursuant to Section 7.6 of the Plan, any objections to Non-Ephedra Claims shall be served and filed on or before the later of (a) 120 days after the Effective Date and (b) such later date as may be fixed by the Court. If any portion of a Non-Ephedra Claim is a Disputed Claim, no payment or distribution provided under the Plan shall be made on account of such Claim unless and until such Disputed Claim becomes an Allowed Claim. Pursuant to Section 5.1 of the Plan, 2002-2004 Ephedra PI Claims and Other Participating Ephedra Claims shall be resolved in accordance with the terms of the Ephedra PI Trust Agreement and the Ephedra PI Claims Resolution Facility.
- 21. Estimation of Claims. The Plan Administrator may at any time request that the Bankruptcy Court estimate any contingent, unliquidated, or Disputed Non-Ephedra Claim including, without limitation, any Consumer Class Action Claim, pursuant to section 502(c) of the Bankruptcy Code or otherwise regardless of whether the Debtors previously objected to such Claim or whether the Bankruptcy Court has ruled on any such objection, and the Bankruptcy Court will retain exclusive jurisdiction to estimate any Claim at any time during litigation concerning any objection to any Claim, including without limitation, during the pendency of any appeal relating to any such objection. In the event that the Bankruptcy Court estimates any contingent, unliquidated, or Disputed Non-Ephedra Claim, the amount so estimated shall constitute either the Allowed amount of such Claim or a maximum limitation on such Claim, as determined by the Bankruptcy Court. If the estimated amount constitutes a maximum limitation on the amount of such Claim, the Plan Administrator may pursue supplementary proceedings to object to the allowance of such Claim. All of the aforementioned objection, estimation, and resolution procedures are intended to be cumulative and not exclusive of one another. Non-Ephedra Claims may be estimated and subsequently disallowed, reduced,

compromised, settled, withdrawn, or resolved by any mechanism approved by the Bankruptcy Court.

- Non-Ephedra Claim becoming an Allowed Claim, the Plan Administrator shall remit to the holder of such Allowed Claim Cash equal to the amount such holder would have received as of that date under the Plan if the Allowed portion of the Disputed Claim had been an Allowed Claim as of the Effective Date. To the extent that a Disputed Non-Ephedra Claim is not Allowed or becomes an Allowed Claim in an amount less than the amount of the Disputed Claim set forth in the proof of claim, or as previously estimated by the Bankruptcy Court, the excess of the amount of Cash that would have been distributed to the holder of the Disputed Claim if the Claim had been Allowed in full over the amount of Cash actually distributed on account of such Disputed Claim shall be Available Cash. Holders of Disputed Non-Ephedra Claims shall not be entitled to interest if such Disputed Claim becomes an Allowed Claim except to the extent such holder is entitled to interest under the Plan as a holder of an Allowed Claim.
- 23. <u>Executory Contracts and Unexpired Leases</u>. On the Effective Date, all executory contracts and unexpired leases to which any of the Debtors are parties shall be deemed rejected as of the Effective Date, except for an executory contract or unexpired lease that (i) has been assumed or rejected pursuant to Final Order of the Bankruptcy Court entered prior to the Effective Date or (ii) is the subject of a separate motion to assume or reject filed under section 365 of the Bankruptcy Code by the Debtors prior to the Effective Date.
- 24. Rejection Claims Relating to Executory Contracts and Unexpired Leases

  Rejected Pursuant to the Plan. In the event that the rejection of an executory contract or

  unexpired lease by any of the Debtors pursuant to the Plan results in damages to the other party

or parties to such contract or lease, a Claim for such damages, if not heretofore evidenced by a filed proof of claim, shall be forever barred and shall not be enforceable against the Debtors, or any property to be distributed under the Plan unless a proof of claim is filed with the Bankruptcy Court and served upon the Plan Administrator on or before the date that is thirty (30) days after notice of the Confirmation Date and such bar date has been provided to the holder of such Claim.

- Exemption from Transfer Taxes. Pursuant to section 1146(c) of the Bankruptcy Code, the assignment or surrender of any lease or sublease, or the delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with the Plan, including any deeds, bills of sale, or assignments executed in connection with any disposition of assets contemplated by the Plan (including transfers of assets to and by the Ephedra Personal Injury Trust and transfers and assignments of assets to Ideasphere pursuant to the Purchase Agreement) shall not be subject to any stamp, real estate transfer, mortgage recording, sales, use, or other similar tax.
- 26. <u>Retention of Jurisdiction</u>. The Court may properly, and upon the Effective Date shall, retain jurisdiction over the matters arising under, arising out of, or related to the Debtors' chapter 11 cases and the Plan, as set forth in Section 11 of the Plan and pursuant to sections 105(a) and 1142 of the Bankruptcy Code.
- 27. <u>Dissolution of Creditors Committee and Ephedra Claimants Committee.</u>

  Pursuant to Section 12.1 of the Plan, the Creditors Committee and the Ephedra Claimants

  Committee shall dissolve on the Effective Date, or as soon thereafter as is reasonably practicable to permit the (i) Creditors Committee and the Ephedra Claimants Committee and their respective professionals the right to pursue, review, and object to any applications for compensation and reimbursement of expenses filed in accordance with Section 11.1(h) of the Plan and (ii) Creditors

Committee the right to continue to pursue, or compromise and settle, any litigation commenced by the Creditors Committee on behalf of the Debtors prior to the Effective Date.

- 28. <u>Substantial Consummation</u>. On the Effective Date, the Plan shall be deemed to be substantially consummated under sections 1101 and 1127(b) of the Bankruptcy Code.
- 29. <u>Payment of Statutory Fees</u>. All fees due and payable pursuant to section 1930 of chapter 123 of title 28 of the United States Code have been or will be paid on the Effective Date or as soon thereafter as is practicable.
- 30. <u>Governmental Approvals Not Required</u>. This Confirmation Order shall constitute all approvals and consents required, if any, by the laws, rules, or regulations of any state or any other governmental authority with respect to the implementation or consummation of the Plan and any documents, instruments, or agreements, and any amendments or modifications thereto, and any other acts referred to in or contemplated by the Plan, the Disclosure Statement, and any documents, instruments, or agreements, and any amendments or modifications thereto.
- Bankruptcy Court of compensation for services rendered or reimbursement of expenses incurred through and including the Confirmation Date, including any such applications under sections 503(b)(2), 503(b)(3), 503(b)(4), or 503(b)(5) of the Bankruptcy Code, (i) shall file their respective final applications for allowance of compensation for services rendered and reimbursement of expenses incurred by the date that is ninety (90) days after the Effective Date and (ii) shall be paid in full in such amounts as are allowed by this Court (A) five (5) Business Days after the date upon which the order relating to any such Administrative Expense Claim is entered or (B) upon such other terms as may be mutually agreed upon between the holder of such

an Administrative Expense Claims and the Debtors, or, if on or after the Effective Date, the Plan Administrator.

- 32. Post-Confirmation Date Fees and Expenses of Professionals. The Debtors shall, in the ordinary course of business and without the necessity for any approval by the Bankruptcy Court, pay the reasonable fees and expenses, incurred after the Confirmation Date, of the professional persons employed by the Debtors, the Ephedra Claimants Committee, and the Creditors Committee in connection with the implementation and consummation of the Plan, the claims reconciliation process, and any other matters as to which such professionals may be engaged. The fees and expenses of such professionals shall be paid within ten (10) Business Days after submission of a detailed invoice therefor. If the Plan Administrator disputes the reasonableness of any such invoice, the Plan Administrator shall timely pay the undisputed portion of such invoice, and the Plan Administrator or the affected professional may submit such dispute to the Bankruptcy Court for a determination of the reasonableness of such invoice.
- 33. Modification of Plan. Upon reasonable notice to the Creditors Committee and the Ephedra Claimants Committee, the Plan may be amended, modified, or supplemented by the Debtors in the manner provided for by section 1127 of the Bankruptcy Code or as otherwise permitted by law without additional disclosure pursuant to section 1125 of the Bankruptcy Code, except as the Court may otherwise direct. In addition, after the Confirmation Date, so long as such action does not materially adversely affect the treatment of holders of Claims or Equity Interests under the Plan, the Debtors may institute proceedings in the Court to remedy any defect or omission or reconcile any inconsistencies in the Plan or the Confirmation Order, with respect to such matters as may be necessary to carry out the purposes and effects of the Plan. Prior to the Effective Date, the Debtors may make appropriate technical adjustments and modifications to

the Plan without further order or approval of the Bankruptcy Court, provided that such technical adjustments and modifications do not adversely affect in a material way the treatment of holders of Claims or Equity Interests. Notwithstanding the foregoing, without the consent of the Settling Third Parties, the Plan may not be amended pursuant to this paragraph if after such amendment, the Plan would not include the Plan Release and Injunction in Favor of Settling Third Parties as provided in paragraph 4 of the 2002-2004 Ephedra PI Settlement Agreement.

- 34. <u>Severability</u>. Each term and provision of the Plan, as it may have been altered or interpreted by the Court, is valid and enforceable pursuant to its terms.
- 35. Notice of Entry of Confirmation Order. On or before the date that is ten (10) days after entry of this Confirmation Order, the Debtors shall serve by first class mail a notice of the entry of this Confirmation Order, in substantially the form annexed hereto as Exhibit "C" (the "Confirmation Notice"), to each of the following at their respective addresses last known to the Debtors: (i) the Office of the United States Trustee for the Southern District of New York, (ii) the attorneys for the Creditors' Committee, (iii) the attorneys for the Ephedra Claimants Committee, (iv) all parties entitled to notice pursuant to the Bankruptcy Court's order dated September 12, 2003 establishing certain notice procedures in these chapter 11 cases, (v) the Plaintiffs' Coordinating Counsel listed in the District Court's CMO No. 1, (vi) the Defendants' Coordinating Counsel listed in the District Court's CMO No. 1, (vii) the Securities and Exchange Commission and the District Director of the Internal Revenue Service for the Southern District of New York, (viii) all persons or entities listed in the Debtors' schedules of assets and liabilities, or any amendments thereto, (ix) any other known holders of Claims against or Equity Interests in the Debtors, and (x) parties to executory contracts or unexpired leases rejected pursuant to the Plan. Such service shall constitute good and sufficient notice pursuant to

Bankruptcy Rule 2002(f)(7) and 2002(i)-(l) of the confirmation of the Plan and entry of this Confirmation Order. The Debtors shall also cause the Confirmation Notice to be published as promptly as practicable after the Effective Date once in each of <u>The Wall Street Journal</u> (National Edition) and The New York Times (National Edition).

- 36. <u>Conditions to Effective Dates</u>. The Plan shall not become effective unless and until the conditions set forth in Section 9.2 of the Plan have been satisfied or waived pursuant to Section 9.5 of the Plan.
- 37. Governing Law. Except to the extent that the Bankruptcy Code or other federal law is applicable, the rights, duties, and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with, the laws of the State of New York without giving effect to the principles of conflict of laws thereof.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

Specifically include any particular provision of the Plan in this Confirmation Order will not diminish the effectiveness of such provision, it being the intent of this Court that the Plan is confirmed in its entirety and incorporated herein by this reference. To the extent of any inconsistency between the provisions of the Plan and this Confirmation Order, the terms and conditions contained in this Confirmation Order shall govern. The provisions of this Confirmation Order are integrated with each other and are nonseverable and mutually dependent unless expressly stated by further order of this Court.

Dated: New York, New York July 26, 2005

/s/ JED S. RAKOFF

THE HONORABLE JED S. RAKOFF UNITED STATES DISTRICT COURT SOUTHERN DISTRICT OF NEW YORK

/s/ ROBERT D. DRAIN

THE HONORABLE ROBERT D. DRAIN UNITED STATES BANKRUPTCY COURT SOUTHERN DISTRICT OF NEW YORK

# UNITED STATES DISTRICT COURT DISTRICT OF MASSACHUSETTS

) ) IN RE: NEW ENGLAND COMPOUNDING )	MDL No.: 1:13-md-2419-FDS
PHARMACY CASES  ) Ma ) This Document Relates to All Cases )	aster Docket No. 12-12052 - FDS

#### ORDER GRANTING CHAPTER 11 TRUSTEE'S MOTION TO TRANSFER PERSONAL INJURY TORT AND WRONGFUL DEATH CASES TO THIS COURT PURSUANT TO 28 U.S.C. §§ 157(b)(5) AND 1334

Upon consideration of the Chapter 11 Trustee's motion (the "Motion") which seeks to transfer to this Court pursuant to 28 U.S.C. §§ 157(b)(5) and 1334 personal injury and wrongful death cases pending in various state and federal courts, and the memorandum in support thereof (the "Memorandum"); and the Court having found that notice of the Motion is proper; and it further appearing that the relief requested in the Motion is in the best interests of the Debtor, its estate, creditors, claimants and other parties-in-interest; and after due deliberation and sufficient cause appearing thereof, it is hereby

1. **ORDERED**, that the Motion is granted; and it is further

DM3\2477178.6

<sup>&</sup>lt;sup>1</sup> Capitalized terms used but not defined herein shall have the same meanings ascribed to them in the Motion and Memorandum.

Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 2 of 18

2. **ORDERED**, that, pursuant to 28 U.S.C. § 157(b)(5), the personal injury and

wrongful death cases identified on Exhibit A attached hereto (the "Pending Actions") are

removed and transferred to this Court for all purposes, including trial, from the courts in which

they are currently pending; and it is further

**ORDERED**, that, pursuant to 28 U.S.C. § 157(b)(5), the clerk of the respective 3.

court for each of the Pending Actions (the "Pending Court") shall transfer the case file pertaining

to the respective Pending Action to the Clerk of the District Court for the District of

Massachusetts within \_\_ days of filing a notice of removal and transfer of the Pending Action

along with a copy of this Order; and it is further

3. **ORDERED**, that a notice of removal and transfer, pursuant to 28 U.S.C.

§ 157(b)(5), of each of the Pending Actions, substantially in the form of Exhibit 1 hereto, and a

copy of this Order shall be filed in the respective Pending Court and provided to all parties in

interest in each of the Pending Actions; and it is further

4. **ORDERED**, that this Order shall not modify or in any way affect the automatic

stay that applies to the Pending Actions under 11 U.S.C. § 362 and which shall remain in full

force and effect pending further Order of this Court or the Bankruptcy Court; and it is further

5. **ORDERED**, that this Order is entered without prejudice to any position that the

Trustee may take regarding whether 11 U.S.C. §§ 105 and 362 operate to stay any Pending

Action or to render null and void the post-petition filing of any Pending Action.

Dated: March , 2013

Boston, Massachusetts

Judge F. Dennis Saylor

United States District Court Judge

2

### **EXHIBIT A**

Pending Actions To Be Transferred to United States District Court for District of Massachusetts Pursuant to 28 U.S.C. § 157(b)(5)

## Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 4 of 18 Exhibit A1

### **Federal Cases Not in NECC MDL**

Plaintiff(s)	Venue	Co-Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Linda S. Boggs	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:2012-cv-50132	Russell Updike
	Virginia, Roanoke	Management, P.C., d/b/a Insight Imaging,		Nolan Nicely
		Roanoke		Jennifer K.M. Crawford
				Wilson, Updike & Nicely
				228 North Maple Avenue
				PO Drawer 590
				Covington, VA 24426
Kimberly Brown	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-584	Russell Updike
·	Virginia	Management, P.C., d/b/a Insight Imaging,		Nolan Nicely
		Roanoke		Jennifer K.M. Crawford
				Wilson, Updike & Nicely
				228 North Maple Avenue
				PO Drawer 590
				Covington, VA 24426
George Cary; Lilian Cary	USDC- District of	NECP d/b/a NECC; Barry Cadden; Arl Bio	Adversary No: 13-	Frederic L. Ellis
(Deceased)	Massachusetts	Pharma INC d/b/a Analytical Research	01025	Edward D. Rapaki
		Laboratories; Ameridose, LLC	Civil Docket: 1:13-	Joseph M. Makalusky
			cv-10228	Ellis & Rapaki, LLP
				85 Merrimac Street
				Suite 500
				Boston, MA 02114
				J. Douglas Peters
				Charfoos & Christensen, PC
				Hecker-Smiley Mansion
				5510 Woodward Avenue
				Detroit, MI 40202
Marvin Clark	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-00566	Patrick T. Fennell, Esq.
	Virginia, Roanoke	Management, P.C., d/b/a Insight Imaging,		Law Offices of Crandall & Katt
		Roanoke		366 Elm Avenue, S.W.
				Roanoke, VA 24016

## Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 5 of 18 Exhibit A1

### Federal Cases Not in NECC MDL

Plaintiff(s)	Venue	Co-Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Darrel Cummings	USDC- Southern District of Florida	NECP d/b/a NECC; GEO Group, Inc; Tom Levins, Warden; Ms. N. Finnisse, MSM, HAS; Dr. J. Dauphin, D.O.; Dr. Jules Heller, M.D.; Dr. Robert Lins, M.D.; Officer McIntire, CO1	9:12-cv-81413	PRO SE
Roseanna Fusco	USDC- Middle District of Florida, Ocala Division	NECP d/b/a NECC	5:13-cv-00095	John H. Piccin, Esq. Piccin Law Firm 320 N.W. 3rd Avenue PO Box 159 Ocala, FL 34478  Colson Hicks Eidson Colson Matthews Martinez Gonzalez Kalbec & Kane
Coleta Ray Gipson, as Personal Respresentative to the Estate of Gayle Lynn Gipson	USDC- Eastern District of Michigan, Southern Division	NECP d/b/a NECC	4:13-cv-10818	Alexander & Angelas, P.C. 30200 Telegraph Road Suite 400 Bingham Farms, MI 48025
Forrest Linthicum	USDC- Western District of Virginia	NECP d/b/a NECC; Image Guided Pain Management, P.C., d/b/a Insight Imaging, Roanoke	7:12-cv-00585	Russell Updike Nolan Nicely Jennifer K.M. Crawford Wilson, Updike & Nicely 228 North Maple Avenue PO Drawer 590 Covington, VA 24426
Michelle Powell	USDC- Western District of Virginia	NECP d/b/a NECC; Image Guided Pain Management, P.C., d/b/a Insight Imaging, Roanoke	7:2012-cv-00614	Patrick T. Fennell, Esq. Law Offices of Crandall & Katt 366 Elm Avenue, S.W. Roanoke, VA 24016

# Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 6 of 18 Exhibit A1

#### Federal Cases Not in NECC MDL

Plaintiff(s)	Venue	Co-Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Basil E. Proffitt	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-00615	Willard J. Moody, Jr.
	Virginia, Roanoke	Management, P.C., d/b/a Insight Imaging,		The Moody Law Firm, Inc.
		Roanoke		500 Crawford Street
				Suite 300
				Portsmouth, VA 23705
				The Miller Firm, LLC
				The Sherman Building
				108 Railroad Avenue
				Orange, VA 22960
Mary E. Radford	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-00562	LichtensteinFishwick PLC
	Virginia, Roanoke	Management, P.C., d/b/a Insight Imaging,		Liberty Trust Building
		Roanoke		Suite 400, 101 South Jefferson Street
				Roanoke, VA 24004-0601
Robert Schroder; Margaret	USDC- District of	NECP d/b/a NECC; Barry Cadden; Gregory	Adversary No: 13-	Ellis & Rapaki, LLP
Schroder	Massachusetts	Conigliaro; Lisa Conigliaro; Glenn Chin;	01024	85 Merrimac Street
		Medical Sales Management, Inc.;		Suite 500
		Ameridose, LLC; ARL Bio Pharma, Inc.	Civil Docket: 1:13-	Boston, MA 02114
		d/b/a Analytical Research Laboratories;	cv-10227	
		GDC Properties Management, LLC		Charfoos & Christensen, PC
				Hecker-Smiley Mansion
				5510 Woodward Avenue
				Detroit, MI 40202
Christopher Sherrill	USDC- Western District of	NECP d/b/a NECC; Image Guided Pain	7:12-cv-00565	Patrick T. Fennell, Esq.
	Virginia, Roanoke	Management, P.C., d/b/a Insight Imaging,		Law Offices of Crandall & Katt
		Roanoke		366 Elm Avenue, S.W.
				Roanoke, VA 24016

# Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 7 of 18 Exhibit A1

### Federal Cases Not in NECC MDL

Plaintiff(s)	Venue	Co-Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Denia Taliaferro	USDC- Western District of Virginia	NECP d/b/a NECC; Image Guided Pain Management, P.C., d/b/a Insight Imaging, Roanoke	7:12-cv-00586	Russell Updike Nolan Nicely Jennifer K.M. Crawford Wilson, Updike & Nicely 228 North Maple Avenue PO Drawer 590 Covington, VA 24426
Mary Sharon Walker	USDC- Western District of Virginia	NECP d/b/a NECC; Image Guided Pain Management, P.C., d/b/a Insight Imaging, Roanoke	7:12cv 00564	Willard J. Moody, Jr. The Moody Law Firm, Inc. 500 Crawford Street Suite 300 Portsmouth, VA 23705  The Miller Firm, LLC The Sherman Building 108 Railroad Avenue Orange, VA 22960
Barbara Whittaker	USDC- Western District of Virginia	NECP d/b/a NECC; Image Guided Pain Management, P.C., d/b/a Insight Imaging, Roanoke	7:12cv 00563	Patrick T. Fennell, Esq. Law Offices of Crandall & Katt 366 Elm Avenue, S.W. Roanoke, VA 24016

#### Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 8 of 18 Exhibit A2

### **State Court Cases Pending Removal**

Plaintiff(s)	Venue	Defendant(s)	Docket	Plaintiff(s)' Counsel
			No.	
Virginia Lois Neely	Davidson County, TN	NECP d/b/a NECC	13C66	Douglas E. Jones
				Schulman, LeRoy & Bennett, P.C.
				501 Union Street, 7th Floor
				P.O. Box 190676
				Nashville, TN 37219-0676

# Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 9 of 18 Exhibit A3

Plaintiff(s)	Venue	Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Sara Culp Andrews	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2495	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
David A. Barley, Jr.	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2437	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Virginia Law Berger	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2439	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Viola Mills Bowling	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2453	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Brenda Brewster	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2438	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Wilma Brumfield	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2399	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Kathleen Cooley and Robert	Middlesex County, MA	NECP d/b/a NECC; Barry Cadden	12 4980-B	Harry Herzog
Cooley				Herzog & Carp
				427 Mason Park Boulevard
				Katy, TX 77450
				Law Offices of Thomas R. Murphy, LLC
				133 Washington Street, 2nd Floor
				Salem, MA 01970

### Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 10 of 18 Exhibit A3

Plaintiff(s)	Venue	Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Harry Michael English	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2435	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Carolyn Fidler, Curtis Lynn	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2509	Cranwell, Moore & Emick, P.L.C.
Shell, Brian Glenn Caddell,		Management, P.C., d/b/a Insight Imaging		111 West Virginia Avenue
Richard Wayne Henegar, Lynda				Vinton, VA 24179
Bradfield, Elizabeth P.				
Kirkpatrick, Kathy N. Cole, Dana				
M. King				
Christy Fralin	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2434	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Robin Annette Hannabass	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2436	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Magaret Hanson	Middlesex County, MA	NECP d/b/a NECC	MICV2012-	Frank Prokos, Esq.
			04645-F	Law Office of Frank Prokos, LLC
				1600 Providence Highway
				Walpole, MA 02081
William E. Johnson and Sandra	Davidson County, TN	NECP d/b/a NECC	12C4920	Randall L. Kinnard
J. Johnson				Daniel L. Clayton
				Kinnard, Clayton & Beveridge
				127 Woodmont Boulevard
				Nashville, TN 37205
Donna Montee	Middlesex County, MA	NECP d/b/a NECC	MICV2012-	Kimberly Dougherty, Esq.
			04861	Janet, Jenner & Suggs, LLC
				75 Arlington Street
				Suite 500
				Boston, MA 02116

### Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 11 of 18 Exhibit A3

Plaintiff(s)	Venue	Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Jocelyn Kae Norris	Cumberland County, TN	NECP d/b/a NECC	CV0005650	Scott & Cain
				550 West Main Avenue
				Suite 601
				Knoxville, TN 37902
James Palmer and Michelle	Cumberland County, TN	NECP d/b/a NECC	CV0005649	Scott & Cain
Palmer				550 West Main Avenue
				Suite 601
				Knoxville, TN 37902
Nancy Phillips	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2398	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Dana Scott Pruitt and Lesa	Davidson County, TN	NECP d/b/a NECC	12C4921	Randall L. Kinnard
Shadowhawk, surviving adult				Daniel L. Clayton
children of Elizabeth A. Pruitt				Kinnard, Clayton & Beveridge
(deceased)				127 Woodmont Boulevard
				Nashville, TN 37205
Craig Simas and Joan Simas	Middlesex County, MA	NECP d/b/a NECC; Ocean State Pain	12 4591	Wickstrom Morse, LLP
		Management, Inc		60 Church Street
				Whitinsville, MA 01588
Kathy Sinclair	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2397	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016
Reba M. Skelton and Roy H.	Davidson County, TN	NECP d/b/a NECC	12C4918	Randall L. Kinnard
Skelton				Daniel L. Clayton
				Kinnard, Clayton & Beveridge
				127 Woodmont Boulevard
				Nashville, TN 37205
Lloyd Smith	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2454	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016

# Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 12 of 18 Exhibit A3

Plaintiff(s)	Venue	Defendant(s)	Docket No.	Plaintiff(s)' Counsel
Margaret Snopkowski	Suffolk County, MA	NECP d/b/a NECC	SUCV2012-	Elizabeth A. Kaveny
			03872	Burke Wise Morrissey & Kaveny
				161 North Clark Street
				Suite 3250
				Chicago, IL 60601-3330
Donna West	Roanoke, VA	NECP d/b/a NECC; Image Guided Pain	CL12-2396	Patrick T. Fennell, Esq.
		Management, P.C., d/b/a Insight Imaging		Law Offices of Crandall & Katt
				366 Elm Avenue, S.W.
				Roanoke, VA 24016

### Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 13 of 18 Exhibit A4

#### **Cases Naming Non-Debtor Defendants Only**

Case Name	Venue	Defendants	Plaintiff(s)' Counsel
Edward Adams	USDC- District of Massachusetts	Barry Cadden; Lisa Cadden; Gregory	Frederic L. Ellis
	Adversary No: 13-01026	Conigliaro; Glenn Chin; ARL Bio	Edward D. Rapaki
	Civil Docket: 1:13-cv-10229	Pharma, Inc. d/b/a Analytical	Joseph M. Makalusky
		Research Laboratories; Ameridose,	Ellis & Rapaki, LLP
		LLC; Medical Sales Management, Inc.;	85 Merrimac Street
		GDC Properties Management, Inc.	Suite 500
			Boston, MA 02114
			J. Douglas Peters
			Charfoos & Christensen, PC
			Hecker-Smiley Mansion
			5510 Woodward Avenue
			Detroit, MI 40202
Bradley	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Epperly	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Filson	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Foutz	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013

### Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 14 of 18 Exhibit A4

#### **Cases Naming Non-Debtor Defendants Only**

Case Name	Venue	Defendants	Plaintiff(s)' Counsel
Harris	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Kalinoski	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
May	Davidson County, TN	Saint Thomas Outpatient Neurological	William D. Leader
		Center, LLC; Howell Allen Clinic A	George H. Nolan
		Professional Corporation	Leader, Bulso & Nolan, PLC
			414 Union Street, Suite 1740
			Nashville, TN 37219
McFarlane	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Reed	Davidson County, TN	Saint Thomas Outpatient Neurological	William D. Leader
		Center, LLC; Howell Allen Clinic A	George H. Nolan
		Professional Corporation	Leader, Bulso & Nolan, PLC
			414 Union Street, Suite 1740
			Nashville, TN 37219
			William H. Lassiter, Jr.
			John Overton Belcher
			Lassiter Tidwell & Davis, PLLC
			One Nashville Place
			150 4th Avenue North, Suite 1850
			Nashville, TN 37219

### Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 15 of 18 Exhibit A4

#### **Cases Naming Non-Debtor Defendants Only**

Case Name	Venue	Defendants	Plaintiff(s)' Counsel
Leonard Shaffer and	USDC- District of Massachusetts	Barry Cadden; Lisa Cadden; Gregory	Frederic L. Ellis
Karen Shaffer	Adversary No: 13-01023	Conigliaro; Glenn Chin; ARL Bio	Edward D. Rapaki
	Civil Docket: 1:13-cv-10226	Pharma, Inc. d/b/a Analytical	Joseph M. Makalusky
		Research Laboratories; Ameridose,	Ellis & Rapaki, LLP
		LLC; Medical Sales Management, Inc.;	85 Merrimac Street
		GDC Properties Management, Inc.	Suite 500
			Boston, MA 02114
			J. Douglas Peters
			Charfoos & Christensen, PC
			Hecker-Smiley Mansion
			5510 Woodward Avenue
			Detroit, MI 40202
Smith, James Wirt	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Smith, Randolph	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Whitlow	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013
Wingate	Roanoke, VA	Image Guided Pain Management,	H. David Gibson
		P.C.; John Mathis, M.D; Robert	Gentry Locke Rakes & Moore
		O'Brien, M.D.; Insight Health	10 Franklin Road Southeast
		Corporation, P.C.	P.O. Box 40013
			Roanoke, VA 24022-0013

#### **EXHIBIT 1**

Form of Notice of Removal and Transfer

# UNITED STATES DISTRICT COURT DISTRICT OF MASSACHUSETTS

) Docket No
) NOTICE OF REMOVAL AND TRANSFER ) PURSUANT TO 28 U.S.C. § 157(b)(5)
) (Transferred from: [NAME OF STATE ) COURT AND STATE COURT CASE NO.]
) ) Docket No ) (Related to Master Docket No. 12-12052-FDS) )

In accordance with the Opinion and Order issued on **[DATE]** by the Honorable Dennis F. Saylor, IV, Judge, United States District Court for the District of Massachusetts (the "Order"), notice is hereby given as follows:

- 1. Pursuant to 28 U.S.C. § 157(b)(5), this case is removed and transferred from the [NAME OF STATE COURT], Case No. [STATE COURT CASE NUMBER] (the "State Court Action") to this Court. (A copy of the Order is attached as Exhibit A.)
- 2. The Clerk of Court in the State Court Action will be directed to transfer the entire court case filed pertaining to this action to the Clerk of the District of Massachusetts within twenty-one (21) days of filing in the State Court Action of a Notice of Filing of Notice of Removal and Transfer Pursuant to 28 U.S.C. § 157(b)(5).
- 3. A Notice of Filing of Notice of Removal and Transfer Pursuant to 28 U.S.C. § 157(b)(5) promptly is being filed with the Clerk of Courts in the State Court Action and provided to all parties of record.

Case 1:13-md-02419-FDS Document 38-3 Filed 03/10/13 Page 18 of 18

For the foregoing reasons, this action is hereby removed and transferred from the **[NAME OF STATE COURT]** to this Honorable Court.

Dated: [DATE OF REMOVAL]

[LOCATION]

Respectfully submitted,

[REMOVING COUNSEL]

Attorneys for [REMOVING DEFENDANT]